Differential Expression of Streptococcus

Pneumoniae Genes during Pathogenesis



Kim Suk LeMessurier, B Biotech (Hons)

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy from the University of Adelaide

May 2007

Discipline of Microbiology and Immunology School of Molecular and Biomedical Sciences The University of Adelaide Adelaide, S.A., Australia

Abstract	vii
Declaration	Х
Acknowledgements	xi
List of abbreviations	xiii

Chapter One: Introduction

1.1 The history of the pneumococcus	
1.2 Cell structure	
1.3 Pneumococcal pathogenesis	5
1.3.1 Nasopharyngeal colonisation	5
1.3.2 Risk factors for developing pneumococcal disease	7
1.3.3 Pneumococcal disease	9
1.3.4 Animal models of pathogenesis	12
1.4 Antibiotic therapy and resistance	
1.5 Pneumococcal vaccines	
1.5.1 Polysaccharide vaccines	15
1.5.2 Polysaccharide-protein conjugate vaccine	15
1.5.3 Serotype replacement	16
1.5.4 Protein vaccines	17
1.6 Important pneumococcal virulence factors	18
1.6.1 Choline binding proteins	19
1.6.1.1 Choline binding protein A (CbpA)	19
1.6.1.2 Pneumococcal surface protein A (PspA)	21
1.6.2 Pneumolysin (Ply)	23

i

1.6.3 Ion acquisition proteins	26
1.6.3.1 Pneumococcal surface antigen A (PsaA)	26
1.6.3.2 PiaA	27
1.6.4 Neuraminidase A (NanA)	28
1.6.5 SpxB	29
1.7 Differential gene expression	30
1.7.1 Phase variation	30
1.7.2 In vitro gene regulation	31
1.7.3 In vivo gene regulation	33
1.8 Aims of this work	35

Chapter Two: Materials and Methods	
2.1 General reagents	36
2.2 Bacterial strains and growth media	36
2.2.1 Pneumococcal growth and storage	36
2.2.2 Opacity phase determination	37
2.2.3 Optochin sensitivity	37
2.2.4 Quellung reaction	37
2.2.5 Pneumococcal passaging	38
2.2.6 Growth curve determination	38
2.3 Oligonucleotides	38
2.4 In vivo pathogenesis models	39
2.4.1 Mice	39
2.4.2 Growth of challenge bacteria	39

2.4.3 IN challenge of mice	39
2.4.4 Euthanasia of mice	40
2.4.5 Enumeration of recovered bacteria	41
2.4.6 Assaying mice infected with luminescent pneumococci	41
2.5 RNA isolation and modifications	42
2.5.1 Acid-phenol RNA extraction	42
2.5.2 Enrichment for prokaryotic RNA	43
2.5.3 Linear amplification of RNA	43
2.6 RNA analysis	44
2.6.1 Reverse transcription polymerase chain reaction (RT-PCR)	44
2.6.2 Real-time reverse transcription polymerase chain reaction	
(Real time RT-PCR)	45
2.6.2.1 Real-time RT-PCR using LUX primers	45
2.6.2.2 Real-time RT-PCR analysis	45
2.6.3 Microarray analysis	46
2.6.4 Agarose gel electrophoresis	47
2.7 Bacterial transformation	47
2.7.1 Preparation of competent cells	47
2.7.2 Transformation of E. coli and S. pneumoniae	48
2.8 Protein analysis	49
2.8.1 Preparation of whole cell lysates (WCL)	49
2.8.2 SDS-PAGE	49
2.8.3 Western blot analysis	50
2.8.4 Haemolysis assay	50

Chapter Three: Optimisation of a mouse model of invasive pneumococcal disease and methods for extraction of RNA from *in vivo*-derived pneumococci

3.1 Intr	3.1 Introduction	
3.2 Dev	3.2 Development of a CD-1 mouse model for RNA recovery	
	3.2.1 Establishment of an IN infection model	53
	3.2.2 Isolation of pneumococci from the nasopharynx	54
	3.2.3 Blood perfusion	55
	3.2.4 Detection of bacterial contaminants	56
3.3 Add	itional studies using serotype 4 challenge strains TIGR4 and WCH43	58
	3.3.1 TIGR4 infection kinetics in CD-1 mice	58
	3.3.2 TIGR4 growth in vitro	60
	3.3.3 WCH43 growth in vitro	61
	3.3.4 WCH43 infection kinetics in CD-1 mice	61
3.4 Add	itional studies using serotype 6A challenge strain WCH16	62
	3.4.1 WCH16 kinetics in CD-1 mice	62
3.5 Con	parison between D39 and WCH16 pathogenesis in CD-1 mice	64
	3.5.1 Construction of luminescent D39 and WCH16 strains	65
	3.5.2 Visualisation of D39 and WCH16 pathogenesis in live CD-1 mice	66
	3.5.3 Determination of Ply production by, and haemolysic ability of,	
	D39 and WCH16 cultures	67
	3.5.4 Survival of mice IN challenged with D39 or WCH16	68
3.6 RNA	A processing and quantitation	69
	3.6.1 mRNA half-lives	69
	3.6.2 Optimisation of differential centrifugation	70

	3.6.3 Enrichment of prokaryotic RNA and linear amplification	72
3.7 Discu	ission	74

Chapter Four: Analysis of key virulence gene expression (for D39 and WCH16) *in vivo* by real time RT-PCR

79
80
80
82
82
83
86
87
90
91
93
94

Chapter Five: Global comparison of pneumococcal genes harvested *in vivo* by microarray analysis

5.2 Comparison of gene expression in pneumococci involved in colonisation, lung

infection and IPD by microarray analysis

96

5.2.1 Analysis of D39 gene expression between niches	97
5.2.2 Analysis of D39 microarray results	98
5.2.3 Preliminary results from WCH16 microarray analysis	100
3.3 Discussion	102
5.3.1 Global in vivo gene expression analysis of S. pneumoniae using a single	
mouse model and challenge strain	103
5.3.2 Comparison of D39 microarray data with other published studies	107
5.3.3 Considerations and conclusion	110

Chapter Six: Final discussion

6.1 Summary of this work	
6.1.1 Recovery of pneumococci from an IN infection model and RNA	
extraction and modifications	114
6.1.2 D39 and WCH16 possess different pathogenesis profiles	116
6.1.3 D39 and WCH16 genes that exhibit contrasting in vivo regulation	117
6.1.4 D39 and WCH16 genes that exhibit similar in vivo regulation	119
6.1.5 Bacteria recovered from the nasopharynx and brain exhibit similar	
expression of some key virulence genes	120
6.1.6 Global microarray analysis identifies differentially expressed genes	
during carriage or disease	121
6.1.7 Conclusion	122
6.2 Future directions	123
Bibliography	126
Appendix	157

Abstract

Streptococcus pneumoniae is a nasopharyngeal commensal in most healthy individuals. However, it can translocate from this niche to deeper tissues, causing diseases such as otitis media, meningitis, sepsis and pneumonia, which are responsible for significant morbidity and mortality worldwide. At the commencement of this work, inherent difficulties in harvesting sufficient bacterial numbers from experimental animals restricted the examination of pneumococcal gene expression during pathogenesis, and thus virulence gene transcription patterns were largely unknown outside of an *in vitro* environment. This thesis aimed to investigate such transcriptional patterns *in vivo*, and to hence gain a better understanding of pneumococcal behaviour during colonisation and disease.

This work describes refinement of an intranasal *S. pneumoniae* infection model in CD-1 mice that enables pneumococci to be harvested from multiple niches with low contamination by nasopharyngeal microflora or host tissue, and minimal cross-contamination with circulating pneumococci in the vascular system. The challenge route simulates the acquisition of *S. pneumoniae* in the human population, and progression to IPD occurs naturally. RNA extraction, enrichment and linear amplification procedures were optimised so that RNA could be obtained from *in vivo* site in sufficient quantities and with sufficient integrity to be used in semi-quantitative assays. Linear amplification allowed the examination of gene expression in niches where low bacterial numbers had previously prevented such analyses.

Real-time RT-PCR and microarray analyses were used to examine bacterial RNA samples recovered from the nasopharynx, lungs, blood and brains of CD-1 mice, providing the first comparative transcriptional data for pneumococci during carriage and disease, within the same animal model. Two pneumococcal serotypes were examined; a type 2

(D39) and a type 6A (WCH16) strain. CbpA, Ply, and SpxB were shown to be important for carriage in both strains, with pneumococci up-regulating the expression of the genes encoding these virulence proteins in the nasopharynx. This provides *in vivo* evidence supporting the ascribed roles of these proteins in reducing the level of competing microflora and promoting nasopharyngeal adherence. Similarly, D39 *nanA* and *pspA* transcription levels were up-regulated in the nasopharynx. The level of *pspA* mRNA was also higher in the blood than the lungs, suggesting an increased requirement in the bloodstream, where PspA is involved in reducing complement-mediated opsonisation. Despite the antiphagocytic role of the pneumococcal polysaccharide capsule in the bloodstream, D39 *cpsA* mRNA was present in similar quantities in the nasopharynx, lungs and blood, which may support previous studies indicating post-transcriptional regulation of capsule expression. However, *cpsA* expression was up-regulated in the blood for WCH16. These results may indicate the existence of strain-specific differences in virulence gene regulation.

Microarray analysis of *in vivo*-harvested *S. pneumoniae* D39 found that mRNAs encoding components of phosphotransferase systems, CbpA, a putative neuraminidase, and v-type sodium ATP synthase subunits were significantly higher in bacteria involved in carriage than bacteraemia. Conversely, the expression of genes involved in competence, and *dinF* (present on a competence-induced operon), were up-regulated in the blood compared to the nasopharynx, providing evidence that competence is induced during bacteraemia. Pneumococci also showed increased expression of genes involved in fatty acid metabolism, pgdA, lytB and cbpG in the blood compared to the nasopharynx. This study used a single pneumococcal strain and infection model and, therefore, overcomes inherent issues of serotype/strain- and animal model- specific gene expression that may have complicated interpretation of data in previous studies.

This thesis reports some of the first *in vivo* pneumococcal gene expression data gained using a single animal model and pneumococcal strain. The data reinforce the

putative roles of several virulence factors, and provides novel transcription data for pneumococci during carriage. Results suggest the existence of core genes that are essential for infection in multiple pneumococcal serotypes, whereas other genes appear to have strain-specific roles.

Declaration

This work contains no material which has been accepted for the award of any other degree or diploma in any university or other tertiary institution, and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text.

I give consent to this copy of my thesis, when deposited in the University Library, being available for loan and photocopying.

The author acknowledges that copyright of published works contained in this thesis (as listed below) resides with the copyright holder/s of those works.

Kim Suk LeMessurier

Published work included in this thesis:

LeMessurier, K.S., Ogunniyi, A.D. and Paton J.C. (2006) Differential expression of key pneumococcal virulence genes *in vivo*. *Microbiology* **152**: 305-311.

Acknowledgements

Thanks to Professor James Paton and Dr David Ogunniyi (DaveO) for the supervision and guidance you've provided during my honours and postgraduate studies. It goes without saying that I couldn't have completed this without both of you, but further to this, your input during my time in the Paton lab has left me with an interest for *S. pneumoniae* that goes beyond the methods you've taught me. Thank you.

Thanks to Dr Uwe Stroeher and Dr Judy Morona for being endless wells of information, and being patient enough to share it with me (sometimes more than once, on those occasions I suffered Short Memory Syndrome). Thanks Dr Tony Focareta, for your encouragement while I was writing up, and for providing me with my entire knowledge-base of cheesemaking.

Thanks to Jan Cook for being the Lab Angel, and for listening to me rant about everything from reagents not working, to the (still unsolved!) mystery of the forever overflowing bin, to my stream of irresponsible and sometimes bizarre flatmates. Also, thanks to Katie Spackman (our ex-Lab Angel).

Thanks to Dr. Habib Alloush for generously providing the pAL2 plasmid used in this study. Thanks to the both the LAS and CSU staff for their help, suggestions and services provided during my canditature. The quality of your work made mine a whole lot less painful.

Thanks to Dr Alistair Standish, Dr Rikki Graham, Dr Kerrie May, Damien Chong and Marcin Grabowicz for all those Fridays nights at the pub. And the Thursday nights. And the nights following all those experiments that bellyflopped. Special thanks to Kerrie and Marcin for being wonderfully supportive while I've been writing up. It's meant a great deal to me. Thanks to the other students, Lauren McAllister, Richard Harvey and James (Jnr) Byrne for putting up with me in the write-up area we shared. Because of you guys, I always had someone to talk to and to bounce ideas off. And I always knew the monthly rainfall (cheers, Richard). Thanks to the other current members of the Paton Lab – Dr Hui Wang, Ursula Talbot, Dr Adrienne Paton, Dr Layla Mahdi and Dr Sylvia Herold – and the past members, for making the lab and interesting and entertaining place to be.

Thanks to my parents for giving me the opportunities they have.

Thanks to my friends Eliana and Taryn, for their personal (and alcoholic) support. I may not have a liver anymore, but I do have a thesis.

List of abbreviations

Abbreviations acceptable to the American Society for Microbiology are used without definition in this thesis. Additional and frequently used abbreviations are defined when first used in the text, and are listed below.

$A_{260}, A_{280}, A_{600}$	Absorbance at 260nm, 280 nm, or 600nm respectively
BA	Blood agar
BBB	Blood-brain barrier
BMEC	Brain microvascular endothelial cells
C3b	Complement component 3b
CbpA	Choline binding protein A
CBP	Choline binding protein
CBR	Choline binding region
CFU	Colony forming units
ChoP	Phosphorylcholine
CPS	Capsular polysaccharide
CSF	Cerebrospinal fluid
CSP-1	Competence stimulating peptide 1
CSP-2	Competence stimulating peptide 2
C-terminus	Carboxy terminus
Ct	Cycle threshold
DFI	Differential fluorescence induction
DIG	Digoxigenin
dNTP	Deoxynucleoside triphosphate
EDTA	Ethylene diamine tetraacetic acid
Ery	Erythromycin
g	Gravity units
hr	Hour(s)
Ig	Immunoglobulin
IN	Intranasal
IP	Intraperitoneal
IPD	Invasive pneumococcal disease
Ig	Immunoglobulin
kDa	Kilodalton(s)
LB	Luria Bertani broth

LytA	Autolysin A
М	Molar
min	Minute(s)
mRNA	Messenger ribonucleic acid
NanA	Neuraminidase A
N-terminus	Amino terminus
O/N	Overnight
ORF	Open reading frame
PAF	Platelet activating factor
PBP	Penicillin binding protein
PBS	Phosphate buffered saline
PCR	Polymerase chain reaction
PiaA	Pneumococcal iron acquisition A
pIgR	Polymeric immunoglobulin receptor
Ply	Pneumolysin
PS	Polysaccharide
PsaA	Pneumococcal surface antigen A
PspA	Pneumococcal surface protein A
PTS	Phosphotransferase systems
rRNA	Ribosomal ribonucleic acid
RT-PCR	Reverse-transcription polymerase chain reaction
SD	Standard deviation
SDS	Sodium dodecyl sulphate
sec	Second(s)
SEM	Standard error of the means
STM	Signature-tagged mutagenesis
THY	Todd-Hewitt broth supplemented with yeast extract
TLR	Toll-like receptor
WCL	Whole cell lysate
WT	Wild-type

CHAPTER 1 - Introduction

1.1 The history of the pneumococcus

In 1880, Louis Pasteur and George Sternberg independently isolated an organism present in human saliva that was capable of causing systemic disease in rabbits. This gram-positive, catalase-negative bacterium has since been referred to as *Microbe septicemique du salive* (Pasteur and Chamberland, 1881), *Micrococcus pasteuri* (Sternberg, 1885), *Pneumococcus* (Fraenkel, 1886), and *Diplococcus pneumoniae* (Winslow *et al.*, 1920). Due to the propensity of the pathogen to form chains in liquid culture media and its association with pneumonia, it was termed *Streptococcus pneumoniae* in 1974, the name by which it is known today. Over twelve decades since its initial identification, *S. pneumoniae* (commonly referred to as the pneumococcus) continues to be responsible for a spectrum of diseases ranging from relatively minor ailments such as otitis media and conjunctivitis, to potentially fatal pneumonia, septicaemia, and meningitis. Young children, the elderly and people with underlying medical conditions such as AIDS or sickle-cell disease are particularly susceptible to

1

pneumococcal infection, and at least one in every 500 Australian children will develop pneumococcal meningitis or septicaemia by the age of two years (Mackenzie *et al.*, 2005). In developing countries, the number of pneumococcal-related deaths has changed very little since the first isolation of the pathogen, with at least a million deaths per annum amongst children under 5 years of age attributed to pneumococcal pneumonia (World Health Organisation, 2007).

Since 1880, the pneumococcus has carved an illustrious history in the field of biology (Austrian, 1981b). It was the first bacterium observed to retain aniline-gentian violet stain, which lead to the development of the Gram-stain (Gram, 1884), and the discovery that capsular swelling occurs when pneumococci are exposed to specific antisera pioneered the development of the 'Quellung reaction', which is now commonly used to serotype human isolates (Armstrong, 1931; Austrian, 1981b). Klemperer and Klemperer (1891) showed that serum from rabbits injected with heat-killed pneumococci was able to elicit protection against re-infection with the challenge strain, but not necessarily against other clinical isolates. Furthermore, when non-sensitised rabbits received an infusion of serum from immunised rabbits they were protected against infection with the same strain, introducing the concept of humoral immunity. In the 1920s, the capsular polysaccharide (CPS) was shown to be responsible for pneumococcal serological activity, and was also demonstrated to elicit strain-specific protection in mice, making it the first identified non-protein antigen (Heidelberger and Avery, 1923; Heidelberger, 1927).

However, perhaps the most important contribution of *S. pneumoniae* to molecular science was a discovery published by Avery, Macleod and McCarty in 1944. In 1928, Griffith reported that co-injection of mice with heat-killed smooth pneumococci and live

un-encapsulated (avirulent) pneumococci resulted in the death of some animals. Examination of bacteria recovered from the dead mice revealed that the live unencapsulated strain had acquired CPS serologically identically to the heat-killed cells in the inoculum (Griffith, 1928). The concept of bacterial transformation was confirmed by Neufeld shortly afterwards (Neufeld and Schnitzler, 1928). However, it wasn't until 1944 that the genetic carrier responsible for transformation was identified as DNA (Avery *et al.*, 1944). Further experiments by McCarty and Avery confirmed this finding, demonstrating that the addition of pancreatic DNase resulted in loss of transformation (McCarty and Avery, 1946). The report that DNA is the carrier of genetic information remains one of the most important discoveries in the molecular biosciences.

1.2 Cell structure

The pneumococcus is a gram-positive organism. Its cell envelope is comprised of a plasma membrane, cell wall, and polysaccharide capsule (as shown in Figure 1.1). The plasma membrane consists of a lipid bilayer, and is approximately 75 nm thick (Tomasz, 1981). The outer cell wall contains peptidoglycan, teichoic and lipoteichoic acids. The latter two components are decorated with phosphorylcholine (ChoP), which promotes adherence by recognising the platelet-activating factor (PAF) receptor on host cells. The phosphylcholine moieties are also a site of anchorage on the cell surface for pneumococcal choline-binding proteins via their choline binding domains (Cundell *et al.*, 1995a).

Perhaps the most defining pneumococcal virulence factor is its polysaccharide capsule (CPS), which possesses antiphagocytic properties. (Briles *et al.*, 1998). The capsule, noted by Pasteur in 1890 as being "a sort of aureole which corresponds perhaps to

NOTE: This figure is included in the print copy of the thesis held in the University of Adelaide Library.

Figure 1.1 Diagram of the pneumococcal surface

Major pneumococcal surface components and important virulence factors are indicated. CbpA and PspA dimers, and autolysin (LytA) are shown bound to phosphorylcholine resides on teichoic and lipoteichoic acids. Other important virulence factors depicted are capsular polysaccharide, neuraminidase and the cytotoxin pneumolysin. Adapted from Briles *et al.* (1998).

a material substance" surrounding pneumococci, was the first pneumococcal virulence factor to be described in the literature (Austrian, 1981a). Subsequently, antibodies to the capsule were found to be protective in animal models, which made it a target for vaccine development (Section 1.5). The pneumococcal CPS varies from approximately 200-400 nm in thickness (Sorensen et al., 1988). Within a particular pneumococcal strain and serotype, a thinner CPS has been correlated with a decrease in virulence, with strains completely lacking a capsule (or 'rough' strains) being avirulent (Austrian, 1981a; MacLeod and Kraus, 1950). The capsule composition is variable. Over 90 different CPS types have been identified, which have negligible immunological cross-reactivity (Garcia et al., 2000). This complicates the development of CPS-based vaccines, as such vaccines are specific to the serotypes included in the formulation. The genes involved in CPS production are located in a locus and typically encode a flippase, polymerase, and various transferases that catalyse the formation of linkages, the addition of sugars and modification of the repeat unit (Paton and Morona, 2000). The first four genes of the locus, referred to as cpsA-D, are highly conserved between all serotypes except types 3 and 37 and have been implicated in CPS regulation (Llull et al., 2001; Morona et al., 1999).

The amount of CPS produced by the pneumococcus varies in different environmental conditions, which possibly affects the degree of exposure of surface proteins believed to be important in pneumococcal pathogenesis (Kim and Weiser, 1998). In some niches, particularly the bloodstream, many such factors associated with the cell wall are partially or completely masked by the polysaccharide capsule. This is thought to prevent the recognition by phagocytes of complement components deposited on the cell surface, thereby reducing pneumococcal clearance from the bloodstream. It also may hamper antibody-mediated clearance of bacteraemic pneumococci. Conversely, a lower level of CPS may increase the exposure of surface proteins such as choline binding protein A (CbpA) and pneumococcal surface protein A (PspA). These proteins are believed to partially extend beyond the polysaccharide capsule, and perform various functions in pathogenesis, such as preventing complement-mediated opsonisation and promoting epithelial adherence and invasion (Rosenow *et al.*, 1997; Tu *et al.*, 1999; Zhang *et al.*, 2000). CbpA and PspA are attached to the pneumococcal surface through interactions with ChoP moieties on teichoic and lipotechoic acids embedded in the peptidoglycan backbone of the cell wall and cell membrane, respectively (see Figure 1.1). These virulence factors are described in Section 1.6.1.

1.3 Pneumococcal pathogenesis

1.3.1 Nasopharyngeal colonisation

Asymptomatic colonisation of the upper respiratory tract by *S. pneumoniae* almost invariably precedes localised and systemic disease, and is also a major factor in horizontal transmission of pneumococcal disease (Malfroot *et al.*, 2004). Prior to the common use of antibiotics, up to 40% of adults were colonised in the nasopharynx by at least one pneumococcal strain. Although these figures are estimated to be lower today, the pneumococcus is still carried asymptomatically by many healthy individuals (Musher, 1992). In most adults, serotype-specific antibodies are produced following acquisition of *S. pneumoniae* and the pneumococcus is cleared after a period of weeks, although in some cases carriage may last for up to one year (Catterall, 1999; Musher, 1992). Carriage is more prevalent in the young, perhaps as a result of a poorly developed immune system and a high degree of contact between individuals, with up to 95% of healthy children colonised

by at least one pneumococcal serotype in some studies (Crook *et al.*, 2004). Nearly all infants are colonised at least once by *S. pneumoniae* by the age of two. Additionally, young infants exhibit a longer duration of colonisation, increasing the opportunity for transmission of the strain to non-colonised individuals (Gray *et al.*, 1980).

Adherence to the mucosal epithelium is crucial for pneumococcal persistence in the nasopharynx (Andersson et al., 1983). Although non-specific factors such as surface charge, pH and hydrophobic interactions are likely to influence bacterial adherence, attachment of S. pneumoniae to host cells in the nasopharynx is thought to be largely mediated by specific interactions between the bacterium and the host (Andersson *et al.*, 1983; Cundell et al., 1995a; Rosenow et al., 1997). However, the nature of the host receptors and bacterial adhesin(s) involved is yet to be fully characterised. Pneumococcal adherence to nasopharyngeal cells has been suggested to occur through interactions with the N-acetyl-D-glucosamine β_{1-3} galactose (GlcNAc β_{1-3} Gal) disaccharide, part of a glycolipid receptor displayed on pharyngeal epithelial cells (Andersson et al., 1983). ChoP-PAF receptor interactions have been demonstrated to mediate adherence to, and invasion of, lung and tracheal epithelia (Ishizuka et al., 2001). However, the addition of a PAF receptor antagonist has no effect on the binding of pneumococci to Detroit-562 nasopharyngeal cells *in vitro*, suggesting that it does not play a similar role in adherence in the nasopharynx (Zhang et al., 2000). Instead, the pneumococcal surface protein CbpA (Section 1.6.1.1) has been implicated in adherence to the nasopharyngeal epithelium and the subsequent invasion of these cells, for some S. pneumoniae strains. A CbpA⁻ strain is attenuated almost 100-fold in its ability to establish nasopharyngeal carriage in infant rats, and exhibits over a 90% reduction in its ability to invade nasopharyngeal cells (Rosenow et al., 1997; Zhang et al., 2000). Knockout mice defective in pIgR trafficking showed a delayed onset of bacteraemia after IN challenge (Zhang *et al.*, 2000). Together, these studies suggest roles for CbpA in both carriage and the progression of disease. This is discussed further in Section 1.6.1.1. Pneumococcal adherence and virulence factor A (PavA) may also contribute to adherence in this niche, as it has been shown to bind to immobilized fibronectin (Holmes *et al.*, 2001). In addition, there are several pneumococcal proteins (discussed in Section 1.6) that may contribute to colonisation, either by directly promoting adherence, or indirectly, for example by preventing pneumococcal killing by apolactoferrin or reducing competition by other nasopharyngeal bacteria (Shakhnovich *et al.*, 2002; Shaper *et al.*, 2004; Spellerberg *et al.*, 1996; Tu *et al.*, 1999).

Exactly what provokes the change from a benign commensal to an aggressively invasive pathogen is unclear, although invasive disease often occurs shortly after the acquisition of a new serotype, and is rarely associated with prolonged carriage (Boulnois, 1992; Gray *et al.*, 1980). Developing vaccines capable of interfering with asymptomatic carriage may be desirable, because such carriers are the principal reservoir for infection (Briles *et al.*, 2000). Thus, eliminating carriage in the individual will confer protection against invasive disease in both the vaccinated and non-vaccinated individuals (herd immunity). However, vaccines directed against carriage face strong selective pressure in the nasopharynx, and may also leave binding sites for pneumococci available for other pathogens.

1.3.2 Risk factors for developing pneumococcal disease

Although carriage of *S. pneumoniae* is common within the population, most occurrences of serious pneumococcal disease occur in children < 2 years of age and adults

> 65 years of age (Parsons and Dockrell, 2002). Infants have an immature immune system which is less able to respond to polysaccharide antigens, leaving them vulnerable to encapsulated bacteria. Additionally, pneumococcal carriage and acquisition of new strains is most frequent in young children, which may contribute to the higher incidence of disease (Crook et al., 2004; Hussain et al., 2005). Some underlying medical conditions predispose pneumococcal infections, such congenital immunodeficiency, human to as immunodeficiency virus (HIV) infection, leukaemia and lymphoma. Sufferers of sicklecell disease are particularly vulnerable, as they exhibit defective reticulo-endothelial clearance of encapsulated bacteria from their bloodstream. Patients taking immunosuppressive medication, such as transplant patients, are also at increased risk of pneumococcal infection. Individuals with respiratory conditions including viral respiratory infections, chronic bronchitis, and chronic obstructive pulmonary disease, or behavioural conditions such as alcoholism and smoking, also face increased chances of developing invasive pneumococcal disease (Centers for Disease Control and Prevention, 1997). Smokers are 4.1 times more likely to develop pneumococcal diseases than individuals who are not exposed to cigarette smoke, and passive smokers have a 2.5-fold increased risk (Nuorti et al., 2000). This may be a result of impaired ciliary and mucociliary defence, epithelial damage, increased bacterial colonisation and adherence, and reduced Other high risk factors include low socio-economic conditions, immunoglobulins. accommodation in a nursing home and attendance at a day-care facility (Kalin et al., 2000; Certain indigenous populations also have increased risk of Levine *et al.*, 1999). pneumococcal disease. Alaskan natives are 8- to 10-fold more likely to develop meningitis and bactaeremic pneumonia then other population groups in the USA (Davidson et al., 1994). In Australia, the incidence of invasive pneumococcal disease in Australian aboriginal children is the highest in the world (Torzillo et al., 1995).

1.3.3 Pneumococcal disease

Pneumococcal disease occurs when colonising organisms in the nasopharynx translocate from this niche to the middle ear, alveolar space or bloodstream causing otitis media, pneumonia and bacteraemia, respectively (McCullers and Tuomanen, 2001). Occasionally, bacteria also traverse the blood-brain barrier and cause meningitis. Otitis media, a common manifestation of pneumococcal infection in young children, results from the progression of colonising pneumococci in the nasopharynx up the Eustachian tube and into the middle ear cavity, where they establish infection. Once in this niche, bacterial replication results in an influx of neutrophils, and cell wall components appear to induce the inflammation characteristic of otitis media (Carlsen *et al.*, 1992). Recurrent bouts of otitis media can result in hearing impairment. Although not fatal, otitis media is a considerable economic and health burden in developed countries. *S. pneumoniae* is the most common bacterial cause of otitis media in the United States of America, where it is responsible for an estimated seven million middle ear infections per annum (Dowell *et al.*, 1999).

The spontaneous aspiration of bacteria from the upper respiratory tract into the alveolar space can result in pneumonia. Pneumococci are thought to attach to pneumocytes via interactions with host glycoconjugates containing GlcNAc β 1-3Gal and GlcNAc β 1-4Gal sequences (Andersson *et al.*, 1981; Cundell and Tuomanen, 1994). A second interaction between ChoP on bacterial teichoic acid and the host cell platelet activating factor receptor has been suggested to mediate both adherence and invasion of the lung and lower tracheal epithelia (Cundell *et al.*, 1995a; Ishizuka *et al.*, 2001). The PAF receptor is known to be rapidly internalised following binding of its natural ligand,

and the pneumococcus hijacks this pathway to transmigrate through the cell (Le Gouill et al., 1997). Moreover, histological studies using pneumococci expressing green fluorescent protein (GFP) have found bacteria inside lung epithelial cells, but not in the tight junctions of these cells, confirming that invasion can occur via trans-cellular migration (Kadioglu et al., 2001). Studies have indicated that CbpA-mediated interactions activate vascular endothelium, resulting in increased capillary permeability and the subsequent leakage of proteinaceous fluid, red blood cells and macrophages into the alveoli. Proteinaceous fluid provides a medium that promotes pneumococcal growth and dissemination. The activation of endothelial cells also increases the amount of surface-expressed PAF receptor, which may promote invasion of these cells (Cundell et al., 1995a). In addition, several pneumococcal components are able to provoke the up-regulation in expression of various cytokines and other pro-inflammatory mediators by resident alveolar macrophages, resulting in the recruitment of polymorphonuclear lymphocytes (Bergeron *et al.*, 1998). It is likely that the activation of vascular endothelium (resulting in increased PAF receptor and capillary permeability) facilitates the entry of pneumococci into the bloodstream, causing potentially fatal bacteraemia. Pneumococci can also enter the bloodstream from the nasopharynx via intracellular translocation facilitated by interactions with the human polymeric immunoglobulin receptor human pIg receptor (pIgR), as discussed in Section 1.6.1.1 (Cundell et al., 1995a; Kadioglu et al., 2001).

Once in the blood, CPS is critical for preventing clearance by host phagocytic cells (Section 1.2). Circulating bacteria can adhere to, and subsequently invade, brain microvascular endothelial cells (BMEC), a process requiring CbpA (Ring *et al.*, 1998). Transmigration across BMEC may also occur via the PAF receptor (Cundell *et al.*, 1995a; Koedel *et al.*, 2002). Thus, invasion of BMEC allows pneumococci to cross the blood-

brain barrier (BBB) and enter the sub-arachnoid space. Prior to bacteria breaching the BBB, the sub-arachnoid space is almost completely devoid of general host defense mechanisms such as polymorphonuclear leucocytes, complement components, and immunoglobulins (Koedel *et al.*, 2002; Simberkoff *et al.*, 1980; Smith and Bannister, 1973). Leukocytes are recruited to the sub-arachnoid space following pneumococcal invasion. However, the cerebrospinal fluid (CSF) lacks sufficient complement concentrations to achieve opsonic activity by leukocytes (Simberkoff *et al.*, 1980), which allows pneumococci to proliferate in this niche without being cleared by the immune system. Pneumococcal components and products such as peptidoglycan, lipoteichoic acid, pneumolysin, hydrogen peroxide and bacterial DNA cause brain cell death and immune activation (Koedel *et al.*, 2002). This results in neuronal damage and massive inflammation, symptomatic of meningitis.

Bacteraemia and meningitis are both referred to as 'invasive pneumococcal disease' (IPD). IPD is the leading cause of death in children younger than two years of age in developing countries, and is associated with an overall case fatality rate of approximately 25%, even with antibiotic treatment (Pallares *et al.*, 1995). Although mortality rates are generally lower in developed countries, the financial cost is high, with the economic burden of pneumococcal infection in the USA in excess of 2 billion dollars per annum (Cunha *et al.*, 1997). Before the introduction of subsidised Prevnar vaccinations for infants in Australia, approximately one in 500 Australian children developed IPD before two years of age (Mackenzie *et al.*, 2005). Furthermore, *S. pneumoniae* meningitis has a case-fatality rate of approximately 20%, which is the highest for community acquired meningitis in the USA (Schuchat *et al.*, 1997).

1.3.4 Animal models of pathogenesis

S. pneumoniae is considered a strict human pathogen, and humans are the principle reservoir for infection. However, a serotype 3 clone has been associated with lower airway disease in horses (Burrell *et al.*, 1986; Neufeld and Schnitzler, 1928; Timoney, 2004). Like its human counterparts, equine *S. pneumoniae* is able to exist asymptomatically in the upper respiratory tracts of healthy animals (Timoney, 2004). Bacteraemia occasionally occurs in foals, although it is rarely associated with adult horses (Meyer *et al.*, 1992). Interestingly, equine isolates have deletions in *ply* and *lytA* genes (encoding pneumolysin and autolysin, respectively), and exhibit a lack of autolytic and haemolytic activity (Whatmore *et al.*, 1999). Regardless, *S. pneumoniae* is still isolated from horses with respiratory infections and bacteraemia, and is able to induce disease in experimental models (Blunden *et al.*, 1994; Timoney, 2004). Clinically, pneumococci are usually isolated from horses already infected with other pathogens, suggesting that *S. pneumoniae* may be an opportunistic pathogen in horses (Chapman *et al.*, 2000).

Although *S. pneumoniae* is only a natural pathogen for humans and horses, several other animals are capable of replicating the pathogenesis observed in humans to varying degrees, thus allowing it to be examined *in vivo*. Chinchillas are the favoured animal in which to study otitis media, while meningitis is often examined using a rabbit model (Giebink, 1999; O'Donoghue *et al.*, 1974). However, due to the handling, expense and availability considerations involved in using these larger animals, the mouse model of infection is most frequently used for *in vivo* experiments. Depending on the strain of mouse and *S. pneumoniae*, experimental mice can develop a range of infections, including nasopharyngeal colonisation, pneumonia, bacteraemia and meningitis (Orihuela *et al.*, 2003; Wu *et al.*, 1997). Several different routes of infection are commonly used in

pneumococcal disease models. Direct challenge models, such as intraperitoneal (IP), intracisternal (IC) or transbullar (TB), involve the injection of bacteria directly into the peritoneal, intracisternal, or middle ear cavities, respectively (Barry *et al.*, 1996; Giebink, 1999; Krontz and Strausbaugh, 1980). Such challenge models allow a specific facet of disease to be examined, as a controlled number of bacteria are introduced into the niche. Because the pneumococcus is primarily a respiratory pathogen with transmission occurring by the aerosol route, an intranasal (IN) challenge model more accurately represents natural acquisition of the pathogen than direct challenge models. Nasopharyngeal colonisation is established following IN infection. From this niche, diseases such as pneumonia, bacteraemia and meningitis can occur naturally. Anaesthetising the animals prior to IN infection relaxes the tracheal cilia, promoting the aspiration of pneumococci in the challenge inocula into the lungs and resulting in a more uniform level of infection.

1.4 Antibiotic therapy and resistance

Since the late 1940s, antibiotic therapy has been employed extensively to treat bacterial diseases, including those resulting from pneumococcal infection. Exposure of bacteria to antibiotics inevitably selects for acquisition of resistance. This problem is particularly significant considering the ability of the pneumococcus, a naturally transformable organism, to acquire resistance by horizontal gene transfer, and the rapid global dissemination of resistant clones (Witte, 1999). In addition, the propensity of the pneumococcus to colonise the nasopharynx of carriers for many months increases the chance of antibiotic exposure. Strains of *S. pneumoniae* with reduced sensitivity to penicillin have been reported since the mid 1960s. Pneumococcal penicillin resistance results from a decreased affinity of the antibiotic for altered penicillin binding proteins

14

(PBPs), which are transcarboxypeptidases involved in cell wall synthesis (Zighelboim and Tomasz, 1980). Penicillin interacts with the PBPs, due to the analogous nature its β -lactam ring and the D-Alanyl-D-Alanine terminus of the peptidoglycan precursors. This prevents transpeptidation from occurring, disrupting cell wall synthesis and leading to cell lysis (Tipper and Strominger, 1965). Currently, over a third of pneumococcal strains exhibit resistance to penicillin (Doern et al., 2001). The incidence is much higher in countries where antibiotic use has been poorly regulated, and is most prevalent for the 'paediatric' serotypes such as 6B, 9V, 9A, 14 and 19F (Schrag *et al.*, 2000). Many penicillin-resistant strains also exhibit resistance to other classes of antibiotics including tetracyclines, chloramphenicol, macrolides and co-trimoxazole. In the United States of America, approximately 11% of S. pneumoniae clinical isolates are resistant to three or more different classes of antibiotics (Sahm et al., 2001). Such multi-drug resistance complicates the treatment of pneumococcal disease, often requiring the use of more expensive antimicrobial agents, and also contributes to outbreaks in hospitals and other care facilities, where antibiotic use is high and individuals are more prone to disease (Appelbaum, 1995; Butler and Cetron, 1999). Although much effort is being directed towards developing new antimicrobials, it is clear that the life-spans of individual antibiotics are limited in terms of their clinical efficacy, and that alternative methods of treatment and prevention must be sought if pneumococcal disease is to be kept under control.

1.5 Pneumococcal vaccines

Antibiotic resistance does not appear to be a problem that will be alleviated simply by the development of new antibiotics. The regular use of antibiotics for children, the elderly, and in the nosocomial setting, both augments the development of resistance and increases the prevalence of resistant clones amongst the groups at highest risk of IPD. The limited life-spans and complications of antibiotics have made vaccination against *S. pneumoniae* an increasingly attractive alternative.

1.5.1 Polysaccharide vaccines

A vaccine comprising 14 purified CPS from the most prevalent serotypes was introduced in 1977. In 1983, coverage was increased to 23 CPS serotypes, namely 1, 2, 3, 4, 5, 6B, 7F, 8, 9N, 9V, 10A, 11A, 12F, 14, 15B, 17F, 18C, 19A, 19F, 20, 22F, 23F and 33F (Mulholland, 1999). The 23-valent vaccine, Pneumovax 23, covers the serotypes responsible for approximately 85% of disease in the USA and Europe (Robbins *et al.*, 1983). In adults, the polysaccharide vaccine is able to elicit a humoral immune response in the majority of vaccinated individuals, thereby resulting in the production of antibodies that enhance opsonisation and phagocytosis of pneumococci (Musher *et al.*, 1990). However, as polysaccharides are T-independent antigens, such CPS -based vaccines are poorly immunogenic in the groups most at risk of developing pneumococcal disease: elderly people, young children and immunocompromised individuals, and also fail to induce immunological memory (Briles *et al.*, 1998).

1.5.2 Polysaccharide-protein conjugate vaccine

When conjugated to a protein carrier, CPS induces a T-dependent immune response and hence has higher success in inducing a lasting response and immunological IgA memory in children under two years of age (Choo *et al.*, 2000; Robbins and Schneerson, 1990). Protein conjugate vaccines also protect against pneumococcal carriage in the nasopharynx, thus resulting in protection to the community as well as the individual (Mbelle *et al.*, 1999; Russell *et al.*, 2000). A vaccine comprising seven pneumococcal capsular types conjugated to diphtheria toxiod (Prevnar) was licensed in the USA in 2000. These seven serotypes (4, 6B, 9V, 14, 18C, 19F, and 23F) accounted for over 80% of IPD in children under five years of age in the USA before the introduction of Prevnar (Centers for Disease Control and Prevention, 2005). In the subsequent three years, IPD decreased by 94% in children under two years in the USA (Centers for Disease Control and Prevention, 2005). The incidence of disease caused by the serotypes included in the conjugate vaccine was also reduced by 55% in adults 50 years or older, a high risk group that was not a vaccination target (Lexau *et al.*, 2005). In addition, the incidence of IPD in infants too young to be vaccinated also decreased significantly, demonstrating significant protection of non-vaccinated individuals by herd immunity (Poehling *et al.*, 2006).

Unfortunately, the high cost of producing the polysaccharide conjugate restricts the number of serotypes that can be included in a single vaccine, which results in limited cover and hence serotype replacement. Currently, a course of immunisations with Prevnar costs about ten times more than immunisation with Pneumovax 23. This cost prohibits the use of Prevnar in developing countries where IPD is a major cause of infant mortality. In addition, conjugate vaccines do not prevent recurrent otitis media (van Heerbeek *et al.*, 2006).

1.5.3 Serotype replacement

Given the negligible immunological cross-reactivity between serotypes of anti-CPS antibodies, polysaccharide vaccines face a high risk of redundancy should strains undergo CPS type switching by recombination at the capsule locus between included and nonincluded serotypes (Bogaert *et al.*, 2004b). Additionally, the niche previously occupied by vaccine serotypes may be colonised by other serotypes either newly acquired or previously present in low numbers, resulting in a serotype shift. Epidemiological studies have confirmed that such serotype replacement is a real issue in practice, demonstrating that although vaccination with capsule-based formulations results in a reduction in carriage of the included serotypes, there is an increase in carriage of non-vaccine serotypes in immunised individuals (Mbelle et al., 1999). A recent study in the USA indicated that, since the introduction of Prevnar, total incidence of pneumococcal bacteraemia has decreased by 57%. Bacteraemia caused by vaccine serotypes decreased from 85% to 34%. (Steenhoff *et al.*, 2006). Initially, it was suggested that some vaccine-included serotypes would elicit a degree of cross-reactive protection against pneumococci within a serogroup. For example, the 6A and 6B CPS differ only by a single glycosidic bond in the structural repeat unit; 6A possesses a rhammopyranosyl- $(1 \rightarrow 3)$ -D-ribitol bond and type 6B has a rhamnopyranosyl- $(1 \rightarrow 4)$ -D-ribitol bond (Robbins *et al.*, 1979). Similarly, serotypes 19A and 19F differ only by the nature of the linkage between CPS repeat units (Katzenellenbogen and Jennings, 1983). Therefore, immunisation with one capsular type was expected to confer cross-protection against the other due to the antigenic similarity. However, data collected by Steenhoff et al. (2006) indicated that the most notable increase in serotype prevalence during the course of their study was for the "vaccine-related" serotypes 19A and 6A.

1.5.4 Protein vaccines

Until recently, the focus of vaccine development has been on the pneumococcal capsule. Pneumococcal strains lacking a capsule were found to be avirulent, and there was a profound gap in knowledge regarding other surface antigens. However, as there are numerous pneumococcal serotypes associated with invasive disease, the development of capsule-based vaccines has been difficult (as discussed above). Recent advances in the

understanding of the structure and function of certain virulence proteins has raised the possibility of using protein antigens in vaccine formulations. A protein-based vaccine has several advantages over current CPS-based vaccines. Firstly, proteins are T-dependent antigens, whereas carbohydrates are not. Therefore, a protein vaccine would have increased immunogenicity in infants and in the elderly. Additionally, immunological memory would be stimulated, thus providing longer-term protection. Secondly, a protein vaccine would be cheaper to produce than conjugate vaccines, making it feasible for use in developing countries. Thirdly, some pneumococcal virulence proteins appear to be conserved between most serotypes, preventing the occurrence of serotype shifting following a vaccination regime with a protein-based formulation, and increasing vaccine coverage.

1.6 Important pneumococcal virulence factors

S. pneumoniae is primarily a colonising organism, existing in the nasopharynx of individuals where it establishes a balance between bacterial aggression and the host's defence system, allowing it to persist in this niche asymptomatically. The development of IPD is incidental and occurs in only a small minority of infected individuals. As the spread of pneumococci occurs most readily via aerosol transfer between a colonised and uncolonised individual, causing systemic disease that might kill the host would be counterproductive to persistence in a population. In this regard, invasive disease is not a favourable outcome and hence the pneumococcus experiences little selective pressure to maintain factors that enhance systemic virulence alone. As a consequence, it has become increasingly apparent that many pneumococcal virulence factors also play an important role in establishing and maintaining stable nasopharyngeal carriage.

1.6.1 Choline Binding Proteins (CBP)

The pneumococcus possesses a number of surface proteins termed 'choline binding proteins' (CBPs) that attach to cell wall teichoic and lipoteichoic acids (Garcia et al., 1998; Rosenow et al., 1997). CBPs share a common Choline Binding Region (CBR) comprising between two and 10 highly conserved 20 amino acid repeats which bind non-covalently to ChoP residues (Giffard and Jacques, 1994). For most CBPs, the CBR is at the C-terminus. The major pneumococcal autolysin, LytA, and PspA (Section 1.6.1.2), were the first two pneumococcal CBPs to be extensively characterised (Garcia et al., 1986; Talkington et al., 1991; Yother and Briles, 1992). Several more CBPs, exhibiting diverse functions, were identified using an immobilised choline affinity matrix, and of these CbpA was the most abundant (Rosenow et al., 1997) (Section 1.6.1.1). Subsequently, other pneumococcal CBPs have been identified, including cell wall hydrolases (LytB and LytC), the putative amidase CbpD and putative serine protease CbpG (Garcia et al., 1999a; Garcia et al., 1999b; Gosink et al., 2000; Kausmally et al., 2005). To date, at least 10 pneumococcal CBPs have been described (Garcia et al., 1986; Garcia et al., 1999a; Garcia et al., 1999b; Gosink et al., 2000; Kausmally et al., 2005; Rosenow et al., 1997; Yother and Briles, 1992).

1.6.1.1 Choline binding protein A (CbpA)

CbpA, also referred to as PspC (pneumococcal surface protein C) and SpsA (*Streptococcus pneumoniae* secretory IgA binding protein), is a surface-exposed CBP. Some pneumococcal strains possess Hic (H-binding inhibitor of complement) instead of CbpA. While Hic exhibits high sequence homology with the N-terminus of CbpA and is thought to share many of its functions, it lacks a choline binding domain. Instead, it

possesses a LPXTG motif and is anchored to the pneumococcal surface in a sortasedependent manner (Janulczyk *et al.*, 2000). The role of CbpA as an adhesin was suggested by Rosenow *et al.* (1997), who demonstrated a decreased ability of a CbpA-negative *S. pneumoniae* mutant (CbpA⁻) to bind cytokine-activated human lung epithelial cells and endothelial cells when compared to the parent strain, and an almost 100-fold reduction in nasopharyngeal carriage in an infant rat model (Rosenow *et al.*, 1997). Likewise, *S. pneumoniae* mutants lacking CbpA are deficient in nasopharyngeal carriage in a mouse model, and are unable to infect and multiply in the lungs (Balachandran *et al.*, 2002). CbpA has also been implicated in pneumococcal adherence to the brain microvascular endothelium and subsequent invasion of the subarachnoid space (Ring *et al.*, 1998).

Zhang *et al.* (2000) reported *in vitro* interactions between CbpA and human pIgR. The group reported that a deficiency in CbpA reduced the ability of mutant pneumococci to invade nasopharyngeal cells by over 90% when compared to the parent strain. Furthermore, knockout mice defective in pIgR trafficking showed a delayed onset of bacteraemia after intranasal challenge. These observations led to the proposal that pneumococci gain access to the interior of the cell by subversion of the pIgR. This is consistent with the primary function of pIgR in apical-basolateral transcytosis of its endogenous ligands IgA and IgM (Mostov, 1994), and with previous reports indicating a strong association between CbpA and secretory IgA (Hammerschmidt *et al.*, 1997). However, studies performed by a separate group indicate that the CbpA-pIgR mediated pneumococcal invasion observed by Zhang *et al.* (2000) was restricted to both the strain of *S. pneumoniae* and the cell type used (Brock *et al.*, 2002). Detroit-562 cells were the only cell line in which CbpA-pIgR-mediated invasion was observed, despite several human pIgR expressing lines being examined. Additionally, of the nine clinical isolates and
laboratory strains of *S. pneumoniae* tested, only R6x, the non-encapsulated strain used by Zhang *et al.* (2000), showed significant invasion of Detroit-562 cells.

In addition to its active role as an adhesin and putative invasive determinant, CbpA has also been shown to bind factor H (Dave *et al.*, 2001). Factor H is a serum protein that inhibits the binding of factor B to C3b, thus protecting host cells from being attacked by its own complement pathway (Dave *et al.*, 2001; Jarva *et al.*, 2003). It is possible that, by binding pneumococcal CbpA, factor H is then able to bind C3b deposited on the cell surface, arresting the complement pathway at this critical stage.

Due to its multiple roles in both carriage and pathogenesis, CbpA presents as a suitable vaccine candidate. Additionally, CbpA extends beyond the capsular layer. Therefore, carriage could be abrogated by either opsonic or neutralising antibodies. Indeed, immunization studies have shown CbpA to be highly protective in mice following IP challenge (Brooks-Walter *et al.*, 1999; Ogunniyi *et al.*, 2001).

1.6.1.2 Pneumococcal surface protein A (PspA)

PspA is a highly immunogenic, surface exposed protein with dual functionality. The protein is tethered to the cell wall via interactions between ChoP moieties on teichoic and lipoteichoic acids and its C-terminal CBR, which shares 90-95% sequence homology with the CBR of CbpA (Brooks-Walter *et al.*, 1999). The highly charged N-terminal region extends beyond the capsule (Yother and Briles, 1992). PspA is found in all clinically important *S. pneumoniae* serotypes, although its molecular size varies from 67-99 kDa (Crain *et al.*, 1990; Waltman *et al.*, 1990). Despite inter-strain variation in amino acid composition of the N-terminus, key protective epitopes exhibit high cross-reactivity

(Briles *et al.*, 1988; Nabors *et al.*, 2000). The α -helical N-terminal portion of PspA is particularly immunogenic, and immunisation with a 43 kDa fragment containing this region has been demonstrated to provide protection against intravenous and intratracheal challenge of mice (Talkington *et al.*, 1991). In human clinical studies, pre-existing antibodies (IgG and IgA) to PspA, in particular the N-terminal region, have been associated with protection against the acquisition of carriage. Indeed, the pre-existence of antibodies to PspA appeared to be the only pneumococcal factor that affected susceptibility in the subjects examined (McCool *et al.*, 2002). This presents PspA as a promising vaccine candidate.

The region 2 of the α -helical domain of PspA has been shown to bind lactoferrin (Hakansson *et al.*, 2001). Lactoferrin is the only identified ligand of PspA, and hence the protein was initially believed to be involved in sequestering iron at respiratory mucosal sites, where the availability of free iron is limited (Hammerschmidt *et al.*, 1999). Presence of lactoferrin alone, however, is not sufficient to support pneumococcal growth in iron-deficient media (Tai *et al.*, 1993). In addition, pneumococci lacking a full-length PspA, but producing a truncated form incapable of binding lactoferrin, have been shown to successfully colonise the human nasopharynx (McCool *et al.*, 2002). Shaper *et al.* (2004) observed that incubation of pneumococci with recombinant PspA could protect pneumococci against the bactericidal effects of apolactoferrin (the iron-depleted form of lactoferrin). This was supported by the ability of antibodies to PspA to increase apolactoferrin-mediated death of pneumococci, indicating that PspA may promote persistence in the nasopharyngeal by preventing killing of pneumococci by apolactoferrin (Shaper *et al.*, 2004).

A second role for PspA is suggested by several observations. Firstly, after intravenous infection, PspA⁻ pneumococci are rapidly cleared from the blood, whereas levels of an otherwise isogenic PspA⁺ strain increased over the same time (Tu *et al.*, 1999) (McDaniel et al., 1987). Secondly, mice infected with PspA⁻ pneumococci have been observed to have significantly decreased C3 concentrations in the serum within 30 minutes after intravenous injection, whereas no significant consumption of C3 was measured in the serum of mice infected with a PspA-positive strain (Tu et al., 1999). In addition, both a PspA⁻ mutant and pneumococci incubated with anti-PspA exhibit increased binding of C3 in vitro (Ren et al., 2003; Ren et al., 2004). These observations implicate PspA in preventing complement-mediated opsonisation. In the alternative complement pathway, C3b that has been deposited on the complement-activating surface is bound by factor B. The resulting complex is then cleaved by factor D, forming C3bBb (Volanakis, 1990). Tu et al. (1999) showed that, while a PspA⁻ mutant was rapidly cleared from the blood of WT mice, it was able to cause bacteraemia in both $C3^{-}$ and factor B⁻ knock-out mice. Therefore, they proposed that PspA functions by blocking formation of, or accelerating the dissociation of, C3bBb, although the mechanism by which this occurs is yet to be defined (Tu et al., 1999).

1.6.2 Pneumolysin (Ply)

Virtually all clinical isolates of *S. pneumoniae* produce pneumolysin (Ply), a 53 kDa cytotoxin that plays an important role in the pathogenesis of pneumococcal disease (Paton *et al.*, 1993). The toxin is able to interact with cholesterol in eukaryotic membranes, triggering its insertion into the lipid bilayer. Oligomerisation of Ply results in formation of transmembrane pores allowing influx of water, ions and some macromolecules, thereby causing cell lysis (Boulnois *et al.*, 1991). Ply is also thought to

induce nitric oxide production by macrophages, resulting in further cell damage (Braun *et al.*, 1999). Unlike related thiol-activated toxins produced by other bacteria, Ply lacks an N-terminal signal sequence required for secretion. In most cases, toxin release coincides with activation of autolysin (LytA), a cell-wall-degrading amidase present in the cell envelope of the pneumococcus (Berry *et al.*, 1989a). Autolysin activation occurs when cell wall biosynthesis ceases during such conditions as stationary growth phase or penicillin treatment. For some strains, Ply has been detected extracellularly during early log phase before significant cell lysis has occurred, suggesting that a LytA-independent release mechanism may also exist in some instances (Balachandran *et al.*, 2001). In most strains, however, Ply release parallels autolysis (Berry *et al.*, 1989a).

It is likely that uncontrolled inflammatory responses contribute to the severity of pneumococcal disease and the high associated mortality rate. This response is due, in part, to cell wall degradation products (Tuomanen *et al.*, 1987). The contribution of Ply to inflammation also appears to be significant, as the toxin is capable of inducing inflammation by several means. Firstly, the purified toxin is able to elicit TNF α , IL-1 β , and IL-6 production by blood monocytes (Houldsworth *et al.*, 1994). The toxin has also been shown to stimulate the production of pro-inflammatory cytokines IL-8 and TNF α by neutrophils (Cockeran *et al.*, 2002). High Ply levels are associated with both increases in transcription and release of IL-8, the latter being possibly due to Ply-mediated alterations of the plasma membrane integrity (Cockeran *et al.*, 2002). Secondly, pneumolysin can activate the classical complement pathway (Paton *et al.*, 1984). This has been proposed to occur following the binding of C1, the first component of complement, to a Ply domain exhibiting similarity to the Fc portion of IgG (Mitchell *et al.*, 1991; Rossjohn *et al.*, 1998).

In vitro and in vivo models have suggested that Ply has roles in multiple stages of pneumococcal pathogenesis (Paton, 1996). Ply facilitates progression of disease by inhibiting ciliary beating in human respiratory epithelium, thus augmenting the retention of bacteria in the lungs where pneumococci can cause pneumonia (Boulnois et al., 1991), and is capable of disrupting tight junctions between epithelial cells, which may provide an alternate pathway by which the pneumococcus infiltrates the bloodstream (Steinfort et al., 1989). Ply-negative pneumococci have been shown to exhibit reduced proliferation in the lungs, and are compromised in their ability to injure the alveolar-capillary barrier compared to the wild-type (WT) (Rubins et al., 1995). Mice injected IP with a pneumolysin negative mutant survive significantly longer than mice challenged with the WT (Berry et al., 1989b; Berry et al., 1992). Additionally, a significant increase in survival time was observed for mice immunised with purified pneumolysin protein prior to challenge with a virulent strain of S. pneumoniae (Paton et al., 1983). Passive immunisation with naturally acquired human antibodies specific for Ply is also protective against IP challenge in mice, implicating the toxin strongly in pneumococcal virulence (Musher et al., 2001). Interestingly, although higher levels of Ply are often associated with invasive disease, the haemolytic level of the Ply appears to be independent of the ability of the strain to cause disease. In some instances, the haemolytic level of Ply has been lower in invasive than in non-invasive isolates (Kirkham et al., 2006; Lock et al., 1996). This may indicate that pneumolysin has an additional function in sepsis.

The role of pneumolysin in pneumococcal carriage is less defined. Ply has been shown to be important for activation of toll-like receptor 4 (TLR-4)-dependent inflammatory responses. This is thought to be an important factor in maintaining a constant level of colonising bacteria in the nasopharynx whilst preventing the development of invasive disease (Malley *et al.*, 2003). Kadioglu *et al.* (2002) found that a Ply mutation eliminated carriage in the mouse model after only four hours, implicating the toxin strongly in nasopharyngeal colonisation. However, studies by different groups indicate that Ply deficiency does not adversely affect colonisation either *in vitro* or *in vivo* (Ogunniyi *et al.*, 2007; Rayner *et al.*, 1995; Rubins *et al.*, 1998). Recently, the use of real-time bioluminescent imaging has indicated that Ply-negative mutants are attenuated in nasopharyngeal colonisation, translocation to the lungs, and survival in the bloodstream, supporting a role for Ply in carriage and demonstrating the importance of the toxin in multiple stages of pathogenesis (Orihuela *et al.*, 2004a).

1.6.3 Ion acquisition proteins

1.6.3.1 Pneumococcal surface antigen A (PsaA)

PsaA, the third gene in the *psa* operon, encodes a conserved 37 kDa lipoprotein belonging to the lipoprotein receptor-associated antigen family (Johnston *et al.*, 2004). Initially PsaA, and indeed many other members of this family, were considered putative adhesins for their respective pathogens. PsaA⁻ pneumococci displayed deficient adherence to type II pneumocytes (Berry and Paton, 1996) and to Detroit-562 cells (Romero-Steiner *et al.*, 2003). Additionally, pneumococci pre-incubated with anti-PsaA antisera (absorbed with a PsaA- mutant) showed a decreased ability to bind to Detroit-562 cells (Romero-Steiner *et al.*, 2003). This suggested that PsaA contributes directly or indirectly to pneumococcal adherence. However, the operon was later found to encode components of an ATP-binding cassette Mn^{2+} transporter (Dintilhac *et al.*, 1997). Manganese acts as a cofactor for bacterial proteins involved in glycolysis, gluconeogenesis, sugar and amino acid metabolism, peptide cleavage, nucleic acid degradation, signal transduction, and

oxidative stress defence, and therefore contributes significantly to the survival and growth of many bacterial pathogens during infection (Jakubovics and Jenkinson, 2001). Pneumococci deficient in PsaA are unable to establish colonisation, cause lung disease, bacteraemia, or otitis media in *in vivo* models, which can be attributed to the very low levels of Mn^{2+} present in these niches (Marra *et al.*, 2002b; McAllister *et al.*, 2004). In addition, *psaA* mutants possess a defective oxidative stress response, which contributes to the attenuated virulence exhibited (Tseng *et al.*, 2002). Moreover, intranasal immunisation of mice with PsaA has been shown to protect against pneumococcal carriage in the nasopharynx (Briles *et al.*, 2000; Johnson *et al.*, 2002), which makes it a potential mucosal vaccine candidate.

1.6.3.2 PiaA

PiaA is the 42 kDa lipoprotein component of one of several pneumococcal iron acquisition systems. It has recently been identified as a vaccine candidate due to its immunogencity, ubiquitous presence in all typical pneumococcal strains and conservation between strains, based on RFLP analysis (Brown *et al.*, 2001b; Whalan *et al.*, 2006). Competitive studies have demonstrated that a PiaA⁻ strain possesses attenuated virulence compared to the WT following co-infection in both a pulmonary and systemic model of infection, indicating that iron uptake via the PiaA ABC transporter is important for full virulence and lung infection (Brown *et al.*, 2001a). In addition, IP immunisation with PiaA protects against systemic challenge and IN immunisation has been shown to be protective against pneumonia (Brown *et al.*, 2001b; Jomaa *et al.*, 2006). The antibodies elicited appear to protect by promoting the clearance of pneumococci by opsonophagocytosis rather than by preventing iron uptake (Jomaa *et al.*, 2005).

1.6.4 Neuraminidase A (NanA)

Many bacterial respiratory pathogens possess at least one neuraminidase, an exoglycosidase that cleaves terminal sialic acid moieties from glycolipids, mucins, glycoproteins and oligosaccharides. S. pneumoniae possesses two defined neuraminidases (NanA and NanB) and a third putative neuraminidase (NanC), the most important of which is considered to be NanA. NanA is 108 kDa in size, and is attached to the cell wall by a LPXTG anchor at the C-terminus (Camara et al., 1994). Pneumococci have been shown to form both transparent and opaque colony phenotypes, based on their appearance when viewed on transparent media. The transparent opacity phenotype is commonly associated with nasopharyngeal colonisation and invasion of the BMEC, while bacteria recovered from the blood of mice with sepsis form opaque colonies (discussed in Section 1.7.1). Transparent 6A and 6B pneumococci have been shown to possess a higher level of functional *nanA* mRNA than opaque variants by microarray analysis, suggesting a role for neuraminidase A in the nasopharynx where this phenotype predominates (King et al., 2004) (Weiser et al., 2001). NanA is capable of both desialating the cell surfaces of Neisseria meningitidis and Haemophilus influenzae and modifying host cell glycoconjugate receptors such as lactoferrin and IgA2 (King et al., 2004; Shakhnovich et al., 2002). This is thought to provide a competitive advantage for the pneumococcus by reducing carriage of other bacterial species that reside in the nasopharynx, whilst promoting adhesion to the epithelia. In vivo, a nanA mutant has been shown to exhibit attenuated persistence in a chinchilla colonisation model. The mutant was completely cleared from the nasopharynx 14 days post-infection, whereas the parent stain prevailed in this niche for 28 days before being cleared. The nanA mutant also showed a decreased ability to persist in the middle ear of chinchillas compared to the WT, although the effect of the NanA deficiency was not apparent until eight days following transbullar inoculation (Tong *et al.*, 2000). NanA does not appear to play a role in sepsis, as a *nanA* knockout mutant is not attenuated in a murine IP challenge model (Berry and Paton, 2000).

1.6.5 SpxB

SpxB encodes a pyruvate oxidase that catalyses the reduction of pyruvate, free phosphate and oxygen to form carbon dioxide, acetylphosphate and hydrogen peroxide. The latter contributes to inflammation and cell damage during infection, and induces TLR-2 and TLR-4 independent apoptosis in brain endothelial cells (Bermpohl et al., 2005). Hydrogen peroxide is also capable of inhibiting the growth of competing microflora, thereby promoting pneumococcal carriage (Spellerberg et al., 1996). This has been demonstrated in vitro, with Haemophilus influenzae, Neisseria meningitidis, and Moraxella *catarrhalis* showing inhibited growth when co-cultured with pneumococci (Pericone *et al.*, 2000). In addition to catalysing H_2O_2 production, SpxB contributes towards pneumococcal resistance to hydrogen peroxide, with mutants lacking a functional SpxB exhibiting 10^2 to 10^3 -fold greater killing in the presence of high levels of hydrogen peroxide than the WT (Pericone et al., 2003). This allows pneumococci to grow in the presence of higher hydrogen peroxide concentrations than many other nasopharyngeal bacteria. Pyruvate oxidase has been shown to be required for pneumococcal growth *in vivo* (Spellerberg *et al.*, 1996), but not *in vitro*. SpxB⁻ D39 possesses a reduced ability to colonise the nasopharynx of IN infected mice, and is also defective in establishing lung and blood infection (Orihuela et al., 2004a).

1.7 Differential gene expression

With the exception of a few serotype 3 strains that are able to cause disease in horses, *S. pneumoniae* is a strict human pathogen (Timoney, 2004). Nevertheless, the pneumococcus is exposed to the varying environments of the nasopharynx, lung, bloodstream, brain and inner ear during human infection. Numerous gene expression profiles are likely to exist in order to enable the pneumococcus to survive and function while occupying different niches in the human host. It is likely these are regulated by a number of different environment cues, for example extracellular signals such as oxygen tension and pH, or host molecules such as immune factors and receptors.

1.7.1 Phase variation

Pneumococci undergo spontaneous opacity phase variation *in vitro* at a frequency of 10^{-3} to 10^{-6} per generation, giving rise to colonies that appear transparent or opaque when viewed under oblique, transmitted light after growth on transparent agar media (Weiser *et al.*, 1994) (see Figure 1.2). An additional 'intermediate' opacity phase was defined by Weiser *et al.* in 1994. For some strains, pneumococcal opacity phase variation has been shown to be independent of pH, growth media, temperature and supplemented CO_2 *in vitro*, with bacteria grown in these conditions showing similar opacity phase switching to a control (McKesser, 2003). *In vivo*, bacteria recovered from the blood of infected mice are predominantly of opaque variants and possess elevated levels of PspA protein when analysed by Western Immunoblot Analysis (Rosenow *et al.*, 1997; Weiser *et al.*, 1994). Furthermore, immunoelectron microscopy has shown opaque colonies to possess more capsular polysaccharide than transparent colonies, which is important in preventing phagocytosis by monocytes in the bloodstream and masking other NOTE: This figure is included in the print copy of the thesis held in the University of Adelaide Library.

Figure 1.2 Pneumococcal opacity phenotype

Single colonies of *S. pneumoniae* strain 62 (type 18C) following growth on THY/catalase agar when viewed under oblique light. A) Shows the transparent colony morphology, and B) shows an opaque colony. Weiser (1994).

pneumococcal surface structures (Kim et al., 1999). Pneumococci cultured from the blood of patients with IPD are also of the opaque phenotype (Weiser *et al.*, 2001). Conversely, the transparent phenotype is believed to express a higher amount of cell wall teichoic acid (Kim and Weiser, 1998). Elevated protein levels of CbpA and SpxB have also been reported for transparent pneumococci compared to the opaque phenotype, which may promote carriage and facilitate adherence to the nasopharyngeal epithelium (Kim and Weiser, 1998; Overweg et al., 2000). In addition, a reduced content of CPS, when compared to its opaque counterpart, may promote the exposure of pneumococcal surface proteins involved in these processes. This hypothesis is supported by enhanced binding of the transparent phenotype (compared to opaque pneumococci) to nasopharyngeal epithelial, lung epithelial and brain microvascular endothelial cell lines, locations where adherence is essential for the pathogenicity of the pneumococcus (Cundell et al., 1995b; Ring et al., 1998; Weiser et al., 1994). Mice challenged IN with transparent pneumococci develop nasopharyngeal colonisation, whereas opaque pneumococci appear to colonise the nasopharynx poorly (Weiser and Kapoor, 1999). Moreover, pneumococci recovered from human carriers are predominantly the transparent phenotype (Weiser et al., 2001). Although opacity phase has been found to be associated with different niches in vivo, the exact mechanism of phase variation is not known.

1.7.2 In vitro gene regulation

Environmental conditions also affect pneumococcal gene expression at a transcriptional level. Using differential fluorescence induction (DFI) technology, artificially induced differences in environmental conditions (specifically low iron, high osmolarity, carbon dioxide, blood agar, and temperature shift) have been shown to influence the activity of a number of pneumococcal promoters *in vitro* (Marra *et al.*,

2002a). DFI involves the creation a promoter trap library of DNA fragments cloned upstream of a promoterless *gfp* gene (encoding green fluorescent protein). The level of promoter activity relates directly to the fluorescence under different environment conditions, which is then used to sort the cells by fluorescence-activated-cell sorter (FACS). Interestingly, this method was not reported to identify changes in promoter activity for key pneumococcal virulence factors such as pneumolysin, the choline binding proteins and genes of the capsule locus. This may have been indicative of constitutive or growth-stage-specific expression. Alternatively, DFI may not have been sensitive enough to pick up the changes in promoter induction, or perhaps regulation of these virulence factors does not occur at a transcriptional level. It is also possible that the environment conditions tested are not responsible for changes in transcription for those genes, which is an unavoidable limitation of such *in vitro* experimentation.

Pneumococcal opacity phase has previously been demonstrated to correlate with either epithelial and endothelial cell contact, or disease. King et al. (2004) examined differences in gene expression between 6A and 6B opacity variants using microarray analysis, using opaque and transparent clinical isolates cultured *in vitro*. They observed that the expression of several genes, including *blpN*, *blpY*, and an operon encoding a sugar ABC transporter, appeared up-regulated in transparent 6A and 6B strains compared to their opaque counterparts. The expression of AliA and a fructose-6-phosphate aminotransferase was higher in the opaque phenotype. *nanA* mRNA was found to be 3.61 times higher in the 6A transparent bacteria, and 2.15 times higher in 6B transparent bacteria, than the respective opaque strains. However, like the findings of Marra *et al.* (2002a), no differences in gene expression were observed between opaque and transparent cultures for many of the key virulence genes previously identified in the literature, despite the

association of the distinct phenotypes with invasive disease or carriage *in vivo* (Cundell *et al.*, 1995b; King *et al.*, 2004; Ring *et al.*, 1998; Weiser *et al.*, 1994; Weiser and Kapoor, 1999). This may suggest that the regulation of virulence proteins is not at the transcriptional level, or perhaps that other factors present in a human host, but not in the *in vitro* conditions used in these two studies, are responsible for triggering the regulatory networks.

1.7.3 In vivo gene regulation

Although neither Marra et al. (2002a) nor King et al. (2004) observed differential expression for many key virulence genes in pneumococci that were subjected to different *in vitro* conditions, or between the opacity phenotypes *in vitro*, a study by Ogunniyi *et al.* (2002) suggested that these genes are indeed regulated *in vivo*. This was indicated by changes in mRNA levels in bacteria harvested from the bloodstream of mice following IP infection, when compared to mRNA isolated from bacteria in the original challenge culture. mRNA levels of *pspA*, *plv*, *psaA* and *cps2A* were shown to be higher in the bloodstream of infected mice at both 12 and 24 hr post-challenge. At both time points, the relative levels of *cbpA* transcript were similar to the *in vitro* control, perhaps not surprisingly considering the primary role of CbpA as a nasopharyngeal adhesin. In contrast, when Marra et al., (2002a) performed in vivo screenings of a bank of strains expressing GFP under the control of specific pneumococcal promoters, those controlling the expression of major virulence genes such as *lytA*, *ply* and the genes in the *cps* locus were not found to be induced. Strains were examined in three in vivo models; a mouse lung infection model, a gerbil otitis media model, and an IP chamber implant model, which involved the surgical insertion of a chamber containing bacteria into the peritoneal cavity of CD-1 mice. Hence, at the commencement of this thesis work, the transcriptional

In 2004, Orihuela et al. published a microarray analysis study comparing pneumococcal gene expression in bacteria harvested from various in vivo infection models and cell lines, to a baseline control. The study identified genes that were up- or downregulated in the bloodstream of BALB/cJ mice following IN challenge with a D39 derivative, or in the brains of New Zealand white rabbits 4 hr after intracisternal infection with a TIGR4 derivative, compared to growth in C+Y media. To simulate nasopharyngeal colonisation an un-encapsulated derivative of TIGR4 was co-cultured with Detroit-562 epithelial cells, and the RNA extracted from the harvested bacteria was compared to that extracted from pneumococci grown in tissue culture medium. In contrast to the *in vitro* studies reviewed in Section 1.7.2, several previously identified virulence factors were found to exhibit niche-dependent expression, including several CBPs; *cbpA* and *cbpJ* expression were up-regulated 2.3- and 3.6-fold, respectively, following incubation with Detroit-562 nasopharyngeal cells, whereas *cbpG* and *cbpF* were induced in the blood of IN infected mice. *lytA* expression was down-regulated in the CSF. In addition, pneumococci repressed *ply* expression in the CSF and when co-cultured with Detroit-562 nasopharyngeal cells. However, the methods used in this study have several limitations, most importantly the need to use different animal models, different infection routes, and different pneumococcal strains to examine disease in each niche. These concerns were also raised by the authors, and are discussed further in Chapter 5.

1.8 Aims of this work

The pneumococcus is capable of causing a number of diseases in the human host which are responsible for significant morbidity, mortality and economic burden worldwide. At the time of commencing this thesis work, the relative expression *in vivo* of specific pneumococcal factors considered to be important in disease pathogenesis was poorly understood. This was largely due to the inability to harvest sufficient RNA from animal hosts to perform quantitative mRNA assays such as real-time RT-PCR. As a consequence, much of the information regarding the role of key pneumococcal virulence factors was gleaned from mutagenesis studies and *in vitro* assays. There was little known about the regulation of the genes encoding these virulence factors *in vivo*.

The aim of this work was to examine differential virulence gene expression in *S. pneumoniae* in an *in vivo* system. Specifically,

1. To develop a suitable *in vivo* mouse model for the harvesting of pneumococci for RNA analysis, and to develop and optimise downstream procedures allowing recovery of RNA of sufficient quality and quantity to perform RNA assays.

2. To examine the expression of key pneumococcal virulence factors in different *in vivo* niches using real-time reverse-transcription PCR analysis.

3. To examine the pneumococcal gene expression in different *in vivo* niches using global cDNA microarray analysis.

CHAPTER 2 - Materials and methods

2.1 General reagents

Most chemicals used were AnalaR grade and were purchased from Ajax Chemicals (NSW, Australia), unless otherwise stated. Antibiotics were purchased from Roche Diagnostics, except gentamicin, which was purchased from Sigma-Aldrich. Sodium dodecyl sulphate was purchased from Sigma-Aldrich (St. Louis, MD., USA). Trypsin was obtained from Trace Biosciences (Sydney, Australia).

2.2 Bacterial strains and growth media

2.2.1 Pneumococcal growth and storage

The pneumococcal strains used in this study are shown in Table 2.1. *S. pneumoniae* were routinely grown in Todd Hewitt broth with yeast (THY) (36.4 g/l Todd Hewitt [Oxoid, England], 10 g/l yeast extract), or on blood agar (BA) plates (39 g/l Columbia base agar [Oxoid]), 5% [v/v] defibrinated horse blood).

For storage, pneumococci were grown in serum broth (10% [v/v] donor horse serum in nutrient broth [10 g/l peptone (Oxoid), 10 g/l Lab Lemco powder (Oxoid) and 5

Strain	Capsular Serotype	Source	Reference
D39	2	Laboratory strain	Avery, 1916
TIGR4	4	T. Mitchell, University of Glasgow	Bricker & Camilli, 1999
WCH43	4	WCH, Adelaide	Clinical isolate
WCH16	6A	WCH, Adelaide	Clinical isolate
D39(LUX)	2	Paton Laboratory, Adelaide	This study
WCH16(LUX)	6A	Paton Laboratory, Adelaide	This study

Table 2.1: Pneumococcal strains used in this study

g/l NaCl]) then stored at -80° C. Alternatively, pneumococci were grown in THY broth, centrifuged upon reaching mid-log phase, and resuspended in THY broth with 30% glycerol prior to storage at -80° C.

2.2.2 Opacity phase determination

For the assessment of opacity phase phenotype, pneumococci were plated on THY with 1.5% agar containing 315 U/ml catalase (Roche Diagnostics, Germany), and incubated at 37°C for approximately 36 hr. Colony morphologies were determined using oblique, transmitted light as described by Weiser *et al.* (1994).

2.2.3 Optochin sensitivity

To confirm bacteria were *S. pneumoniae*, strains were tested for optochin sensitivity by plating on blood agar in the presence of an optochin disc (Oxoid), at 37° C in an atmosphere of 95% air and 5% CO₂. Pneumococci are inherently sensitive to optochin, whereas other Streptococcal spp. are resistant.

2.2.4 Quellung reaction

Production of a specific capsular serotype was confirmed by the Quellung reaction using type-specific pneumococcal antisera (obtained from Statens Seruminstitut, Copenhagen, Denmark). Strains, taken from overnight blood agar cultures, were grown in serum broth (in the presence of the appropriate antibiotic(s)) for 3 hr at 37°C. 5 μ l of serum broth culture was then mixed with 4 μ l of antiserum and examined microscopically for specific capsular swelling.

2.2.5 Pneumococcal passaging

Pneumococci were grown in serum broth until the culture reached an OD_{600} of approximately 0.8. An aliquot of culture was then centrifuged at 4000 × *g* for 10 min at 4°C. The pellet was resuspended in serum broth and administered IP to CD-1 mice. 24 hr post-challenge, approximately 50 µl of blood was recovered from the infected mice following sacrifice, plated on BA plates and incubated overnight at 37°C in an atmosphere of 95% air and 5% CO₂. THY stocks were prepared from the passaged bacteria, as described in Section 2.2.1.

2.2.6 Growth curve determination

S. pneumoniae strains were cultured in THY following overnight growth on BA. Initial absorbance at 600nm (A_{600}) was measured at the commencement of growth, and strains were subsequently incubated at 37°C for 6-8 hr during which A_{600} readings were taken at predetermined time points. At each of these time points, a 50 µl aliquot of culture was serially diluted in 450 µl sterile phosphate buffered saline (PBS) then plated on BA and incubated overnight at 37°C, to determine the number of CFU (colony forming units).

2.3 Oligonucleotides

The oligonucleotides used in this work were purchased from Sigma-Aldrich and are listed in Table 2.2. LUX fluorogenic primers, labelled with JOE, were obtained from Invitrogen (California, USA) and are listed in Table 2.2.

Name	Sequence $5 \rightarrow 3$	Strain (Accession No.)
16S F (JOE)	CAACATCACCGCATAAGAGTGGATG(JOE) TG	R6 (gi: 15459504)
16S R	CAACGCAGGTCCATCTGGTA	R6 (gi: 15459504)
16S R2	AACGCAGGTCCATCTGGTAGTG	R6 (gi: 15459504)
cbpA F (JOE)	GATCGTTTGGTTGGGTAGTTACGACGA (JOE)C	D39 (gi: 6469846)
<i>cbpA</i> R	GCAGAAGCTAAGAAGAAGGTTGAA	D39 (gi: 6469846)
<i>cbpA</i> 6B F	TGCTAAGTTGAAGGAAGCTGTTG	BG9163 (gi: 6469854)
cbpA 6B R (JOE)	CGAGGTAGCTCTCCAGAAACTCCT(JOE)G	BG9163 (gi: 6469854)
pspA F (JOE)	CTAGCGAACTGAGGAGAAAGCCGCGC (JOE)AG	R6 (gi:15902044)
<i>pspA</i> R	GGCTAGATACGCTTGTTGAACTGC	R6 (gi:15902044)
<i>pspA</i> 6B F	TTCCGTGCTCCTCTTCAATCTG	DBL6A (gi: 6752380)
pspA 6B R (JOE)	CGGTGTATTGTCCAGCTTGCTCAC(JOE)G	DBL6A (gi: 6752380)
ply F (JOE)	CAACGAAGGATAGAGGCGACTGTCG(JOE) TG	R6 (gi: 15459416)
ply R	GAAAGAAAGAAGCGGAGCTTG	R6 (gi: 15459416)
psaA F	GAACCACTTCCTGAAGACGTTAAG	R6 (gi:15902044)
psaA R (JOE)	GAAACACAAFCATTGCCACCTGT(JOE)TC	R6 (gi:15902044)
cps2A F (JOE)	CAACTCGCGGGCATTTATGGAG(JOE)TG	R6 (gi: 15457864)
cps2A R	GAAGTGAAGTTCAATCGCACATAG	R6 (gi: 15457864)
piaA F (JOE)	CAGCACAAGCAAATTACGGTGTAAGTGC (JOE)G	R6 (gi: 15458539)
piaA R	TCGTCAAATAGGTTAGCTTTACCA	R6 (gi: 15458539)
nanA F (JOE)	GACGTTGAGCAGTGGTATTTGGAACG (JOE)C	R36A (gi: 587550)
nanA R	CCGAAAGTTGAGTTTCATTTGC	R36A (gi: 587550)
<i>spxB</i> F (JOE)	CACCGGCGACTCACTTGATTAACGG(JOE) G	D39 (gi: 32744844)
<i>spxB</i> R	TTAACTGGACGTGATCCAAGG	D39 (gi: 32744844)

Table 2.2: Oligonucleotides used in this study

The oligonucleotides used in this study are listed in this table. For LUX fluorogenic primers, the JOE label is indicated by (JOE).

2.4 In vivo pathogenesis model

2.4.1 Mice

Animal experimentation was conducted in accordance to the Australian Code of Practice for the Care and Use of Animals for Scientific Purposes (7th edition). CD-1 outbred female mice, obtained from the Institute of Medical and Veterinary Science (Adelaide, SA, Australia) were used in all experiments. Mice were aged between 5 and 6 weeks at the commencement of experiments, and weighed between 20 g and 24 g with an average weight of 22 g. The animals were known to be free of common mouse pathogens when obtained, and were maintained under contamination barrier conditions during experimentation.

2.4.2 Growth of challenge bacteria

THY broth supplemented with 5% yeast extract was inoculated with *S. pneumoniae* taken from overnight blood agar plates, and grown at 37°C to $A_{600} = 0.25$ (equivalent to mid logarithmic phase). For challenge of mice, 5 ml of culture was centrifuged at 2200 × *g* for 15 min, and supernatant was carefully decanted. The bacterial pellet was resuspended in 2 ml of sterile PBS, and kept on ice prior to intranasal challenge. The challenge dose administered was determined retrospectively by plating appropriate dilutions of the inoculum on blood agar plates immediately after infection of mice.

2.4.3 IN challenge of mice

Previous studies by Wu *et al.* (1997) indicate that there is no detectable transmission of pneumococci from one mouse to another when animals are co-housed. Therefore, mice were separated into cages of 10-15 animals and maintained in a specific

pathogen-free environment for the duration of the experiment. Prior to challenge, mice were anaesthetised by intraperitoneal injection with Nembutal (pentobarbitone sodium, Rhone-Merieux) at a dose of 66 μ g per 1 g of body-weight. 25 μ l of the desired pneumococcal strain was pipetted into the nares and involuntarily inhaled. Mice were monitored closely until they had regained consciousness, which took approximately one hour. The number of bacteria in the challenge inoculum was determined retrospectively by growth on BA at 37°C in 95% air, 5% CO₂ O/N.

2.4.4 Euthanasia of mice

The following procedure was adapted from the method originally described by Wu et al. (1997). Animals were euthanased by asphyxiation with carbon dioxide at 24, 48 and 72 hr following challenge. The trachea of each mouse was cut below the larynx, and the loosely adherent bacteria in the nasopharynx were collected by washing with 1 ml trypsin buffer (0.5% trypsin, 0.02% ethylenediaminetetracetic acid [EDTA], in PBS, filter sterilised) by insertion of a 26-gauge needle sheathed in tubing into the exposed tracheal end of the upper respiratory tract. Trypsin buffer was allowed to drip into the nasopharynx slowly and was collected from the nose, with each wash taking approximately 40 sec. The left atrium of the heart was cut, and a small sample of blood removed for determination of bacteraemia level. Mice were perfused with a small amount of cold PBS in order to collect 2 ml blood for RNA extraction. Mice were subsequently perfused with sterile PBS until the eluent was clear, to remove blood-borne bacteria from the lungs. Lungs were removed and homogenized on ice in 2 ml sterile PBS using a tissue homogeniser (Cat X120, Germany), which was rinsed in PBS, decontaminated in 70% (v/v) ethanol, and rinsed again in PBS between samples. To separate pneumococci from host cells, lung homogenates and blood samples were centrifuged twice at $855 \times g$ for 6 min at 4 °C, an adaptation of a method described previously (Ogunniyi *et al.*, 2002). For the 6A strain, whole brains were removed from perfused mice through an incision in the skull, and placed in 2 ml sterile PBS. The organ was then homogenised and centrifuged at $850 \times g$ for 6 min at 4°C, as for lung and blood samples. Nasopharyngeal washes and lung, blood and brain supernatants were subsequently centrifuged at $15,500 \times g$ for 2 min at 4°C, the supernatant decanted, and the bacterial pellet stored at -80° C until further processing.

2.4.5 Enumeration of recovered bacteria

Prior to pelleting harvested bacteria, 40 μ l was removed from the nasal washes and the supernatant of the lung, blood and brain samples centrifuged at 850 × g. The aliquot was serially diluted in sterile PBS and plated onto blood agar in duplicate in order to enumerate pneumococci present in the sample, and to determine the presence, if any, of contaminating microflora. Blood plates were incubated at 37°C in 95% air, 5% CO₂ O/N. Colonies selected randomly from a subset of samples were plated and tested for Optochin sensitivity, as described in Section 2.2.3.

2.4.6 Assaying mice infected with luminescent pneumococci

Mice were anaethetised by IP injection with Nembutal (pentobarbitone sodium, Rhone-Merieux) at a dose of 66 μ g per 1 g of body-weight. After anaesthesia was established, mice were placed in an imaging chamber without restraint and bioluminescence was visualised using an IVIS 100 Imaging System (Xenogen Corporation, Alameda, CA.) with a 500-nm filter to target the luciferase (γ_{max} , 490 nm). Animals were scanned both ventrally and dorsally for between 2 min and 7 min during which time the imaging chamber was maintained at 30°C. The bioluminescent images were superimposed over grayscale reference photographs taken prior to bioluminescence acquisition. Total photon emission from selected and defined areas within the images of each mouse was quantified using the LivingImage software package (Xenogen Corporation).

2.5 RNA isolation and modifications

2.5.1 Acid-phenol RNA extraction

RNA was isolated from bacterial pellets with acid-phenol:chloroform:isoamyl alcohol (125:24:1; pH 4.5, Ambion, Austin, TX., USA) essentially as described previously (Ogunniyi et al., 2002). Cells were pelleted by centrifugation at $15,500 \times g$. The supernatant was removed, and the cells resuspended in acid-400 μl phenol:chloroform:isoamyl alcohol, pre-warmed to 65°C. Samples were incubated at 65°C for 5 min, after which 400 µl pre-warmed NAES solution (50 mM sodium acetate, 10 mM EDTA, 1% [w/v] SDS, pH 5.1 treated with diethyl pyrocarbonate [DEPC]) was added to the lysate and mixed by inversion before being incubated at 65°C for a further 5 min. The lysate was then placed on ice for 2 min and centrifuged at $15,500 \times g$ for 2 min at 4°C. The aqueous phase was removed and placed in a new RNase-free tube, and the phenol extraction was repeated. Following the phenol extractions, 30 µl of 0.05% (w/v) DEPCtreated 3 M sodium acetate and 750 µl 100% ethanol was added to the aqueous phase, and mixed by inversion. The extract was then precipitated at -80° C overnight in the presence of 40 ng/µl glycogen (Sigma). Following precipitation, nucleic acids were pelleted by centrifugation at $15,500 \times g$ for 30 min at 4°C. The pellet was washed with 1 ml 70% ethanol (0.7 vol 100% ethanol and 0.3 vol DEPC-treated water) and centrifuged for 5 min at $15,500 \times g$. The supernatant was aspirated by pipetting, and residual ethanol was removed by air-drying for 15 min. The nucleic acids were suspended in 18 μ l RNase-free, DNase-free water (Roche). Subsequently, the preparation was treated with 10 units of RNase-free DNase (Roche) at 37°C for 30 min in the presence of 1 U/ μ l recombinant RNasin ribonuclease inhibitor (Promega Life Sciences, WI., USA), after which RQ1 DNase stop buffer (Promega) was added to inactivate the DNase. The purity of the RNA preparation was confirmed by one-step reverse transcription polymerase chain reaction (RT-PCR, Section 2.6.1) with or without reverse transcriptase, using 16S rRNA-specific primers (Table 2.2), and the products were visualised after electrophoresis on a 2% TBE-agarose gel (see Section 2.6.4).

2.5.2 Enrichment for prokaryotic RNA

RNA samples from a specific niche were pooled from 4 to 5 mice, based on the number of CFU recovered and also on the absence or minimal presence of contaminating bacteria and RNA was purified further using a Qiagen RNeasy[™] minikit. RNA obtained from lung homogenates was enriched for prokaryotic RNA using the MICROB*Enrich*[™] kit (Ambion), in accordance to the manufacturer's instructions. The amount of RNA recovered following purification/enrichment was determined by A_{260/280} measurements.

2.5.3 Linear amplification of RNA

RNA samples were purified by using Qiagen RNeasy[™] clean-up columns, according to the manufacturer's instructions. RNA was then amplified using a SenseAmp Plus[™] RNA Amplification Kit (Genisphere, Hatfield, PA., USA), in accordance to the manufacturer's instructions for Amplification of Prokaryotic Samples, or Amplification of Degraded Samples (as required). A second round of amplification was performed following primary amplification, in accordance to Genisphere's instructions for Second

Round Amplification. 12.5 μ l of tailed RNA was transcribed using either vials 13, 14 and 15 included in the SenseAmp PlusTM kit, or using a MEGAscript High Yield Transcription KitTM (Ambion). For tailed RNA transcribed using MEGAscript, 2 μ l each of ATP, CTP, GTP and UTP solutions, 2 μ l reaction buffer, and 2 μ l enzyme mix were added to the template RNA, as suggested by Genisphere, and reactions were incubated at 37°C overnight. The remaining 12.5 μ l tailed RNA was stored at –70°C for later transcription. The yield of RNA following linear amplification was determined by A_{260/280} measurements. Amplified RNA was stored at –70°C until use.

2.6 RNA analysis

2.6.1 Reverse transcription polymerase chain reaction (RT-PCR)

Oligonucleotide primers used for RT-PCR analysis were obtained from Sigma-Aldrich, and listed in Table 2.2. To test for contamination of RNA samples with DNA, RT-PCR was carried out using the Access RT-PCR System (Promega) in accordance with the manufacturer's instructions. A standard 12 μ l RT-PCR reaction mix contained DEPCtreated water, 0.2 mM of each dNTP, 1 μ M of each primer, 1 mM MgSO₄, AMV/*Tfl* reaction buffer, 0.1 U/ μ l AMV Reverse Transcriptase, 0.1 U/ μ l *Tfl* DNA Polymerase, 0.25 μ l RQ1 DNase stop solution and 0.3 μ l RNA extract. A control reaction mix was performed in tandem, for which RNase-free DNA-free water was substituted for Reverse Transcriptase.

RT-PCR amplification was performed using a Hybaid Touchdown Thermocycler (Hybaid, Middlesex, England). The standard amplification conditions consisted of one

reverse transcription cycle at 48°C for 45 min. The subsequent PCR step consisted of 30 cycles, each consisting of 30 sec of denaturation at 95°C followed by 45 sec of annealing at the optimal temperature and 1 min of elongation at 68°C.

2.6.2 Real-time reverse transcription polymerase chain reaction (Real-time RT-PCR)

2.6.2.1 Real-time RT-PCR using LUX primers

Real-time RT-PCR was performed using LUX fluorogenic primers obtained from Invitrogen, listed in Table 2.2. Real-time RT-PCR was carried out using SuperScriptTM III One-Step RT-PCR System with Platinum *Taq* DNA Polymerase (Invitrogen) in accordance with the manufacturer's instructions. A standard 20 µl reaction contained 0.3 µl Superscript III RT/Platinum *Taq* Mix, 10 µl reaction mix, 1 µM of each primer, and 4 µl RNA extract. Amplification was performed using a Corbett Research Model RG-2000 Real-time Thermal Cycler. The standard amplification conditions consisted of one reverse transcription cycle at 48°C for 45 min. The subsequent PCR step comprised of 40 cycles, each consisting of 15 sec of denaturation at 95°C, followed by 30 sec of annealing at the optimal temperature and 30 sec of elongation at 72°C. The reaction was acquired at both 60°C and 72°C.

2.6.2.2 Real-time RT-PCR analysis

Relative mRNA concentration was calculated using the comparative cycle threshold $(2^{\Delta\Delta Ct})$ method (Livak and Schmittgen, 2001), in which the amount of target mRNA in one niche is compared to the amount of mRNA in another, relative to an internal control (16S rRNA). Results are expressed as relative changes in target gene mRNA levels

in one niche compared another. Standard deviations (SD) were initially determined as the $\sqrt{((\text{SD sample})^2 + (\text{SD 16S})^2)}$, and this was then applied to the formulas: $\text{SD} + 2^{\Delta\Delta\text{Ct}}$.

2.6.3 Microarray analysis

Total bacterial RNA was labelled and hybridised using the Genisphere Array 900 MPX Kit (Genisphere), according to the manufacturer's instructions. 3 µg of each RNA preparation was used per microarray slide. cDNA synthesis steps specified in the Genisphere Array 900 MPX method were performed using SuperScript IIITM (Invitrogen), according to Genisphere's instructions. Microarray slides (Bacterial Microarray Group, St Georges's Hospital, University of London, UK) consisted of PCR products for each of the defined open reading frames of both S. pneumoniae TIGR4 and R6, spotted in duplicate. Prior to use, arrays were incubated in filtered pre-hybridisation solution (1% [w/v] BSA, 17.5% [v/v] SCC, 0.1% [v/v] SDS in sterile water) for 30 min at 37°C. Arrays were then washed twice in sterile water to remove all pre-hybridisation solution, washed once in isopropanol, and quickly dried using nitrogen. Subsequent hybridisation of the labelled RNA to the microarray slide was carried out at 65°C. The array was read using an Axon 4000B Dual-wavelength Scanner (Molecular Devices Sunnyvale, CA., USA), and images were acquired using GenePix Pro 4.0 (Axon Instruments, Inc.). Array readings were normalised by adjusting the PMT 532 nm and 635 nm intensities until 16S rRNA ratios were approximately 1: 1. Gene lists were prepared based upon filtration algorithms and signal-to-noise ratios by GenePix Pro 4.0, with a 4- fold cut-off for each spot and with P value < 0.01. The P value was calculated from Student t-tests between the intensities of individual spots. This analysis is described further in Section 5.2.2.

2.6.4 Agarose gel electrophoresis

DNA was electrophoresed through horizontal agarose gels (0.8-2% [w/v] agarose dissolved in TBE buffer [44.5 mM Tris, 44.5 mM boric acid, 1.25 mM EDTA, pH 8.4]) immersed in TBE buffer containing ethidium bromide (100 μ g in 2.5 l) at 140-180 V. Prior to loading DNA samples, a one tenth volume of tracker dye was added (15% [w/v] Ficoll, 0.1% [w/v] bromophenol blue, 100 ng/ml RNase A). DNA bands were visualised by short wave UV transillumination and images were captured by a Tracktel video imaging system attached to a Mitsubishi thermal printer. Approximate sizes of visualised fragments were calculated by comparison of their mobility with that of DNA size markers derived from HpaII-digested pUC19 DNA (fragment sizes: 501, 404, 331, 242, 190, 147, 111, 110, 67, 34, 26 base pairs).

2.7 Bacterial transformation

2.7.1 Preparation of competent cells

E. coli cells were grown at 37°C to an absorbance at 600nm (A₆₀₀) of 0.4, then centrifuged for 10 min at 2880 × g at 4°C. The pellet was then resuspended in cold 0.1 M MgCl₂, and placed on ice for 10 min before pelleting at 2880 × g for 10 min at 4°C. The pellet was then resuspended in a one fifth volume of cold 0.1 M CaCl₂, and placed on ice for 90 min. Competent cells were stored at -80°C in 100 µl aliquots containing 20% glycerol.

Pneumococcal cells were grown in c-CAT medium (10 g/l Bacto Casamino acids

[Difco, MD, USA], 5 g/l Bacto Tryptone, 5 g/l NaCl, 10 g/l Bacto yeast extract, 4% [v/v] 0.4 M K₂HPO₄, 0.002% [w/v] glucose, 150 mg/l glutamine) to an A₆₀₀ of 0.25-0.3, then diluted to an A₆₀₀ of 0.01 in 10 ml CTM medium (0.2% [w/v] BSA, 1% [v/v] 0.1 M CaCl₂) and grown to an A₆₀₀ of 0.1. The cells were then pelleted at 10,000 × g for 10 min, and resuspended in 812 µl CTM-pH 7.8 and 188 µl 80% glycerol, and stored at -80° C in 100 µl aliquots.

2.7.2 Transformation of E. coli and S. pneumoniae

Transformation of *E. coli* involved adding ~1 µg DNA directly to competent DH5 α , and leaving on ice for 15 min. Cells were then heat-shocked at 42°C for 2 min, and placed on ice for a further 15 min. 1 ml LB was then added and after shaking for 45 min at 37°C, the culture was pelleted, resuspended in 200 µl LB, and plated on LB agar plates with the appropriate antibiotic selection.

Transformation of S. pneumoniae involved 500 µl of CTM-pH 7.8 and 2.5 µl of 10 competence stimulating peptide-1 (CSP-1) µg/ml (amino acid sequence: MRLSKFFRDFILQRKK [Chirontech (Victoria, Australia)]) or competence stimulating peptide-2 (CSP-2) (amino acid sequence EMRISRIILDFLFLRKK [Mimotopes (Victoria, Australia) (Havarstein et al., 1995)]) were added to an aliquot of competent cells. The cells were then incubated at 37°C for 10 min before addition of approximately 1 ng of DNA, and then incubated at 32°C for 30 mins, followed by 37°C for 2-4 hr. After incubation, cells were plated onto BA plates containing 0.2 µg/ml Ery, and incubated for approximately 16 hr at 37°C in 95% air/ 5% CO₂.

2.8 Protein analysis

2.8.1 Preparation of whole cell lysates (WCLs)

Pneumococci were cultured in THYuntil they reached early to mid-log growth, as determined by A_{600} readings ($A_{600} = 0.25$). Cells were centrifuged at $2500 \times g$ for 15 min. The pellet was resuspended in one tenth of the original culture volume of PBS, thereby concentrating the culture 10-fold. Pneumococcal cells were then lysed by incubating the cultures at 37°C for 10 min in the presence of 0.1% sodium deoxycholate (DOC) solution. The lysate was centrifuged at 15,500 × g for 2 min and the supernatant was removed and stored at -20°C until use.

2.8.2 SDS-PAGE

Pellets of 10 ml bacterial cultures were resuspended in 100 μ l of LUG (5% [v/v] β mercaptoethanol, 62.5 mM Tris, 2% [w/v] SDS, 10% [v/v] glycerol, 0.05% [w/v] bromophenol blue). Samples were boiled for 5 min prior to analysis, and stored at -20°C until use.

Sodium dodecyl sulphate polyacrylamide gel electrophoresis (SDS-PAGE) was performed using the method described by Laemmli, (1970). Proteins were separated by electrophoresis at 170 V through a stacking gel (6% acrylamide) and a separating gel (12 to 20% acrylamide). Proteins were stained by gentle agitation for 1 h at 65°C in 0.06% (w/v) Coomasie Brilliant Blue R250, dissolved in 10% (v/v) glacial acetic acid and 25% (v/v) methanol. Destaining was performed by washing with several changes of 10% (v/v) acetic acid and 10% (v/v) isopropanol.

2.8.3 Western blot analysis

Proteins were separated by SDS-PAGE (as described in Section 2.8.2) and electroblotted onto nitrocellulose (Pall Life Sciences, MI., USA) at 300 mA for 1 hr as described by Towbin *et al.* (1979). After transfer, the membrane was blocked in 5% (w/v) skim milk powder (Diploma) in TTBS (20 mM Tris-HCl, 154 mM NaCl, 0.5% (v/v) Tween-20, pH 7.4) for 20 min, with gentle agitation. The blocking solution was removed and the membrane probed with the appropriate polyclonal antiserum at a dilution of 1/3000 in TTBS overnight at room temperature with gentle agitation. The filters were washed with TTBS 3 times for 10 min, before the addition of blotting grade goat anti-mouse IgG-Alkaline Phosphate conjugate (BioRad Laboratories, Hercules, CA., USA) at a dilution of 1/15,000 in TTBS, and incubated for 1 h at room temperature with gentle agitation. The membrane was washed 4 times for 5 min each with TTBS, and then equilibrated in 15 ml DIG 3 (100 mM Tris-HCl, 100 mM NaCl, 50 mM MgCl₂, pH 9.5) for 2 min before addition of 45 µl DIG 4 (75 mg/ml nitroblue tetrazolium salt in 70% dimethylformamide), and 35 µl DIG 5 (50 mg/ml 5-bromo-4-chloro-3-indolylphosphate toluidinium salt in dimethylformamide). After the desired colour reaction had taken place, the reaction was stopped by the addition of TE (10 mM Tris, 1 mM EDTA, pH 8.0). The membrane was then rinsed in water and dried.

2.8.4 Haemolysis assay

3% human red blood cell (hRBC) suspension was prepared by centrifuging 2 ml hRBC for 1 min in a microfuge, and washing the pellet twice with sterile PBS. Following washing, the solution was centrifuged at $15,500 \times g$ for 2 min, and the RBC pellet was resuspended in 34 ml PBS and 50 µl β-mercaptoethanol.

To determine pneumolysin activity of pneumococcal cultures, haemolysis assays were performed as follows. 100 µl PBS was pipetted into each well of a 96-well roundbottomed microtiter plate (Sarstedt, South Australia), and serial 2-fold dilutions of pneumococcal lysates were performed in duplicate. 100 µl was removed and discarded from the last well of each row. Lysate was not added to four rows, which were used as controls. In one row, an extra 100 µl PBS was added to determine the blank absorbance value. In a further two rows, $100 \ \mu$ l 3% hRBC solution was added to determine the level of lysis in the absence of pneumolysin, and 100 µl 3% hRBC solution with purified Ply added was used as a positive control. 100 µl 3% hRBC solution was pipetted into the wells containing lysate, and the plate was then incubated at 37°C for 30 min. Following incubation, the microtiter plate was centrifuged at $850 \times g$ for 10 min. 100 µl of supernatant from each well was transferred to a new flat-bottomed microtiter plate, and the absorbance at 540 nm was read using a Spectramax M2 spectrometer (Molecular Devices). The results were graphed as average absorbance against dilution. The haemolytic activity was determined as the reciprocal of the dilution at which 50% of the RBCs had lysed.

CHAPTER 3 – Optimisation of a mouse model of invasive pneumococcal disease and methods for extraction of RNA from *in vivo*-derived pneumococci

3.1 Introduction

To date, there have been limited *in vivo* studies reported in the literature that examine the expression of pneumococcal virulence genes during pathogenesis. This can be attributed to restrictions of *in vivo* models of infection, including limited bacterial recovery from specific niches such as the nasopharynx, difficulty in obtaining samples without substantial contamination by host cells and resident microflora, and sensitivity of quantitative mRNA assays available. As a result, knowledge of how key pneumococcal virulence factors are expressed and regulated *in vivo* has been mainly extrapolated from mutagenesis studies and *in vitro* assays. The work in this chapter aimed to develop an appropriate and working pneumococcal IN infection model and downstream RNA extraction/amplification protocols that would yield RNA in sufficient amount and purity for quantitative assays to be performed.

3.2 Development of a CD-1 mouse model for RNA recovery

3.2.1 Establishment of an IN infection model

In order to examine gene expression in varying in vivo niches, it was essential to use a mouse model of infection from which pneumococci could be harvested from the nasopharynx, lung, blood and brain in numbers sufficient for RNA analysis. An IN infection model was considered the most appropriate, as this simulates the natural route of infection for the pneumococcus. The CD-1 mouse strain (also referred to as the Swiss mouse strain) was chosen for this work for several reasons. Firstly, it is an outbred strain and, therefore, better represents a population of individuals. The use of an inbred population leaves the possibility that results are due to a specific characteristic or phenotype of the mouse strain. Secondly, pneumococcal pathogenesis in CD-1 mice closely resembles the pathogenic progression in its human host. Pneumococci are able to be carried in the nasopharynx and then progress to IPD (Wu et al., 1997), whereas for some other mouse strains, such as BALB/c, the establishment of nasopharyngeal colonisation rarely leads to fulminant infection. The serotype 2 strain D39 was used to develop this CD-1 IN infection model. D39 was first isolated from the blood of a patient in the USA by Avery in 1916 and is a prototype laboratory pneumococcal strain that has been intensively studied. D39 is highly virulent for mice: it is capable of both colonising the nasopharynx and causing invasive disease in CD-1 mice, and continues to be commonly used in models of pneumococcal infection. In addition, the availability of its complete genomic sequence made it particularly attractive for this study (Lanie et al., 2007).
An IN model was adapted from that described by Wu et al. (1997). 5- to 6- week old female CD-1 mice were intranasally infected with D39 under Nembutal anaesthesia, as described in Sections 2.4.2 and 2.4.3. Mice were euthanased by CO₂ asphyxiation at 24 and 72 hr post-infection. The nasopharynx of each mouse was washed with 1 ml 0.5% trypsin solution in PBS and then placed on ice until further use. Blood was perfused from each mouse as described in Section 3.2.3. Approximately 2 ml of blood and PBS was recovered from the chest cavity during early perfusion, and held on ice. Following complete perfusion, the lungs were excised and homogenised on ice in 2 ml sterile PBS using a tissue homogeniser. To separate pneumococci from host cells, lung homogenates and blood samples were centrifuged twice at $1000 \times g$, a method adapted from that of Ogunniyi et al. (2002). The optimisation of this procedure is described in Section 3.6.2. Nasopharyngeal washes, and lung and blood supernatants were subsequently centrifuged at $15,000 \times g$ for 2 min at 4°C, and the bacterial pellet was stored at -80°C until further processing. Prior to pelleting harvested bacteria, an aliquot of the sample was removed, diluted in PBS, and plated on BA in order to enumerate pneumococci and determine the presence of contaminating microflora. This method for euthanasing mice and processing recovered samples is described in more detail in Section 2.4.4. The method was initially tested using 12 female CD-1 mice, aged 5 weeks. Mice were challenged IN with 1.6×10^7 D39 (determined retrospectively by plating on BA for enumeration), and a number of mice were euthanased at 24 and 72 hr post-challenge, as described in this section. Bacterial counts harvested from the nasopharynx, lungs and blood are shown in Figure 3.1.

3.2.2 Isolation of pneumococci from the nasopharynx

A washing procedure to remove loosely adherent bacteria from the nasopharyngeal cavities of mice was described by Wu *et al.* (1997), involving washing the severed



Figure 3.1: Disease progression in CD-1 mice following IN infection with D39

In this pilot experiment, D39 was cultured in THY until mid-exponential phase and 6 CD-1 mice were each challenged IN with 1.6×10^7 *S. pneumoniae* D39. Mice were euthanased at 24 and 72 hr post-challenge, and nasopharyngeal washings (N), lung homogenates (L) and blood samples (B) were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. The time point is indicated in subscript. Three mice died between 24 and 72 hr. The median for each time/niche is indicated by a bar, and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. <40 CFU.

nasopharynx with Ringer's Solution. Bacteria that are associated strongly with, or have invaded, the epithelium can be recovered by excising and homogenising the nasopharyngeal tissue. However, homogenised tissue has been found to contain a significant proportion of other nasopharyngeal bacteria (LeMessurier, 2002). The presence of other bacteria was highly undesirable, as 16S rRNA, which was to be used as an internal control for quantitative RNA assays, is highly conserved amongst bacteria. For this reason, pneumococci residing in the nasopharynx were recovered by washing rather than by removing the nasopharyngeal tissue, as nasopharyngeal washes contain a much lower proportion of contaminating microflora. In an attempt to increase the recovery of more adherent bacteria by washing, colonising pneumococci were loosened and washed out of the nasopharynx of infected mice using a PBS buffer containing trypsin. Trypsin is commonly used at 0.25% to detach epithelial cell lines from the substratum *in vitro* whilst leaving pneumococci intact (Talbot et al., 1996). To test the effect of trypsin on pneumococcal viability, D39 cultures in mid-exponential phase were incubated with PBS (control) or a PBS buffer containing either 0.25% or 0.5% trypsin (0.02% EDTA) for 1 hr and then diluted and plated on BA for enumeration. After overnight incubation at 37°C, similar CFU counts were obtained for the pneumococci incubated with both concentrations of trypsin, and the control culture (Table 3.1). Therefore, pneumococci were considered to retain viability in 0.5% trypsin, which was the concentration used for nasopharyngeal

washes in this work.

3.2.3 Blood perfusion

The blood of mice that have developed sepsis commonly contains 10^{7} - 10^{9} pneumococci per ml. Circulating pneumococci in the capillaries has often been a factor overlooked in the literature when examining tissue infection and it is possible that small

Culture supplement	CFU/ml
PBS	$5.0 imes 10^6$
0.25% trypsin	5.1×10^{6}
0.5% trypsin	$4.8 imes 10^6$

Table 3.1: Survival of D39 in trypsin

S. pneumoniae D39 were incubated with PBS (control) or a PBS buffer containing either 0.25% or 0.5% trypsin (0.02% EDTA) for 1 hr and then diluted and plated on BA. CFU/ml were calculated after incubation at 37°C ON.

volumes of blood containing pneumococci may artificially elevate numbers of pneumococci recovered from the lungs and the brain. To reduce the contamination of lung and brain samples by bacteria present in the blood, mice were perfused with cold, sterile PBS prior to the removal of the organs. This was achieved by cutting the right atrium of the heart directly following euthanasia, then running PBS from an elevated container and into the heart via tubing and a 26 gauge needle inserted in the left ventricle. Mice were perfused until the PBS flowing from the right atrium was clear of blood, which generally took 2 min. Animals exhibiting lung damage from pneumococcal infection did not perfuse as efficiently as mice with healthy lungs, with the flow of PBS during perfusion often being slower. These mice required approximately 3 min of perfusion until the PBS was clear of blood. Perfusing the animals in this manner was found to effectively remove most, if not all, blood-derived pneumococci from the lung and brain tissues. This was demonstrated by the observation that D39-infected mice possessing a high pneumococcal burden in the blood did not necessarily possess pneumococci in the lungs or brain, as determined by CFU counts for the respective niches. Perfusion with ice-cold PBS also rapidly chilled the tissues.

3.2.4 Detection of bacterial contaminants

A variety of commensal bacteria exist in the respiratory tracts of mice. Additionally, bacteria native to other sites in the body, such as the gastrointestinal tract, can also be found in the murine nasopharynx following experimental infection, due to the poor hygiene exhibited by ill animals. As 16S rRNA was to be used as an internal control for quantitative RNA analysis, the high homology between the 16S rRNA sequences of pneumococci and many other bacterial species, in particular other streptococci (Kawamuru *et al.*, 1995), made the contamination of pneumococcal samples recovered from the mouse with other bacteria undesirable.

Some nasopharyngeal bacteria can be easily distinguished from S. pneumoniae on the basis of morphology on BA. However, in addition to phenotypically distinct bacteria, alpha-haemolytic colonies that appeared almost identical to S. pneumoniae were recovered from the lungs and nasopharynx of some CD-1 mice. This contaminant was distinguished from S. pneumoniae by growth on optochin plates (to which it was resistant), and was subsequently identified as *Enterococcus faecalis*, a related bacterium commonly present as a commensal in the mammalian intestinal tract and oral cavity (Kayaoglu and Orstavik, 2004). Control mice challenged IN with sterile PBS were found to possess small numbers of E. faecalis in their respiratory tracts after 72 hr. When mice were challenged with pneumococci, animals possessed even greater numbers of this contaminant in the lungs at this time point (data not shown). Commonly, pneumococci recovered from the upper respiratory tract are grown on BA in the presence of low levels of gentamicin (to which S. pneumoniae is inherently resistant), allowing the growth of pneumococci while suppressing the growth of other respiratory microflora (Converse and Dillon, 1977). However, enterococci recovered from the CD-1 mice were also resistant to low levels of gentamicin, making this unsuitable as a basis for recognition of the contaminant.

Contaminating flora encountered in laboratory animals may often be eliminated by a course of antibiotics prior to challenge. However, as young mice (5-6 wk) were required for this model of infection, and antibiotics take several weeks to be completely eliminated from a treated mouse, this approach did not seem feasible. Therefore, CD-1 mice were sourced from a different supplier for the *in vivo* experiments described in this thesis. Bacterial samples harvested from mice were routinely plated on plain BA to visually screen for contaminants such as *Staphlycoccus* spp. Any suspicious colonies, and also a random selection of colonies, were then streaked on BA in the presence of an optochin disc, as described in Section 2.2.3. Resistant colonies were deemed to be contaminating bacteria, and the corresponding samples were discarded. Thus, all RNA assays were carried out on samples harvested from mice that did not exhibit contamination. In addition, colonies were randomly selected from samples recovered from all sites examined and grown in serum broth for 3 hr. Cultures were then tested by the Quellung Reaction (Section 2.2.4) to confirm serotype-specific CPS production by all organisms present in the culture.

3.3 Additional studies using serotype 4 challenge strains TIGR4 and WCH43

3.3.1 TIGR4 infection kinetics in CD-1 mice

Using the model established in Section 3.2, several other strains were examined for their pathogenicity in CD-1 mice. Initially, TIGR4 was selected. TIGR4 was isolated from a 30-year-old male patient in Norway in the late 1980s, and was a particularly attractive strain for this work as its complete genomic sequence is available (Tettelin *et al.*, 2001). The strain has been shown to be highly virulent in a MF1 murine pneumonia model, with a median survival time of 28 hr post-challenge (Ibrahim *et al.*, 2004; Ibrahim *et al.*, 2005). In BALB/cJ mice, TIGR4 establishes only low-level lung and blood infection by 72 hr following IN infection (Orihuela *et al.*, 2003). However, results from Oriheula *et al.* (2003) suggested that TIGR4-infected BALB/cJ mice may develop meningitis over a longer time-course. Both MF1 and CD-1 mice are outbred strains. In addition, we have previously found BALB/c mice to be resistant to disease following IN challenge with the serotype 2 pneumococcal strain D39 compared to CD-1 mice (unpublished data), so it was thought likely that TIGR4 would be able to both colonise and progress to invasive disease in CD-1 mice.

In preliminary IN pathogenicity trials, TIGR4 was unable to cause bacteraemia or lung infection, and could only establish low-level colonisation of the nasopharynx by 72 hr following challenge (data not shown). Subsequently, TIGR4 was grown in serum broth and passaged three times through CD-1 mice as described in Section 2.2.5. To assess the kinetics of TIGR4 infection in CD-1 mice, four female CD-1 mice were challenged IN with 3×10^7 bacteria (Section 2.4.3) in a pilot experiment. Surviving mice were euthanased 72 hr post-challenge and bacteria present in each niche were enumerated, as described in Sections 2.4.4 and 2.4.5. In this trial experiment all mice exhibited colonisation of the nasopharynx at 72 hr, albeit at a level approximately 10-fold lower than that commonly observed for the serotype 2 strain D39. All surviving mice were also found to have lung infection and a high level of bacteraemia (Figure 3.2). Due to these preliminary data indicating the ability of TIGR4 to cause both invasive disease and colonise the nasopharynx, it was initially selected for examination of differences in gene expression between host niches. However, the levels of infection observed in the trial experiment were not reproducible. In subsequent experiments, when TIGR4 was harvested from CD-1 mice at 24 and 72 hr post IN-challenge (challenge dose of 4.2×10^7 bacteria per mouse), bacteria were recovered from nasopharyngeal washes of over half the mice, but were absent from both the lungs and the blood (Figure 3.3A). This experiment was repeated using a challenge dose of 1.4×10^7 bacteria per mouse (Figure 3.3B). Although mice euthanased at 24 hr exhibited pneumococci in the lungs and bloodstream, few



Figure 3.2: Pilot experiment: Disease progression in CD-1 mice following IN infection with TIGR4

4 CD-1 mice were challenged IN with 3×10^7 *S. pneumoniae* TIGR4 per mouse, and surviving animals were euthanased 72 hr post-challenge. One mouse died prior to 72 hr from bacteraemia. Nasopharyngeal washings, lung homogenates and blood samples were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. The median for each time/niche is indicated by a bar, and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. <40 CFU.

Figure 3.3: Disease progression in CD-1 mice following IN infection with TIGR4

TIGR4, a serotype 4 clinical isolate, was cultured in THY until mid-exponential phase and CD-1 mice were challenged IN with the culture. Mice were euthanased at 24 and 72 hr post-challenge, and nasopharyngeal washings (N), lung homogenates (L) and blood samples (B) were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. The time-point is indicated in subscript. **A**) and **B**) represent data obtained from two independent experiments (challenge dose 4.2×10^7 and 1.4×10^7 CFU respectively) with the median for each time/niche indicated by a bar and the limit of detection (40 CFU) shown by a dotted line. ND denotes no growth, i.e. <40 CFU.



animals euthanased at 72 hr had bacteria in either of these niches.

3.3.2 TIGR4 growth in vitro

Despite the tertiary *in vivo* passaging, TIGR4 still grew poorly in THY broth, with A_{600} readings indicating that growth often ceased after only a few hours of incubation. When growth curves in THY broth were established (Section 2.2.6) and compared to the CFU/ml at each time point measured, a decrease in A_{600} readings coupled with a decrease in CFU/ml was observed after 4.5 hr, suggesting that autolysis may be occurring (Figures 3.4A and 3.4B).

A similar decrease in CFU/ml was observed for cultures grown in THY broth supplemented with 0.5% choline, which is routinely added to pneumococcal cultures during prolonged incubation periods to prevent cell lysis from occurring (Briles et al., However, the A_{600} reading increased over the time course, indicating that 1996). pneumococcal growth and division were occurring with minimal lysis. The decrease observed in CFU/ml may have been due to cell chain formation or clumping. When TIGR4 was grown in THY supplemented with 10% horse serum, both A₆₀₀ and CFU/ml increased during the entire time course (Figures 3.4A and 3.4B). As suggested by the magnitude of the error bars representing SEM between two replicate experiments, the variation in TIGR4 growth was significant between experiments. This experiment was conducted additional times (data not shown) and similar fluctuations in growth curves was observed. As a consequence of poor growth *in vitro* resulting in concomitant uncertainties regarding dosage calculation and unreliable infection of CD-1 mice, it was decided that TIGR4 was not a suitable challenge strain for *in vivo* comparison of gene expression using this model.



Figure 3.4: TIGR4 growth in THY supplemented with serum or choline

TIGR4, taken from overnight BA plates, was cultured at 37°C in THY media supplemented with either 10% horse serum or 0.5% choline, as indicated. Bacteria were also grown in THY with no supplement. A) A_{600} was recorded at predetermined time points over a 6- hr incubation period. B) At these time points, culture was diluted and plated for enumeration. The error bars indicate SD between two replicate experiments.

3.3.3 WCH43 growth in vitro

As TIGR4 did not seem able to establish stable and reproducible infection in CD-1 mice, a second serotype 4 *S. pneumoniae* strain (WCH43, a clinical isolate obtained from the Women's and Children's Hospital, South Australia) was examined. WCH43 was cultured in THY broth to determine a growth curve (described in Section 2.2.6). Unlike TIGR4, WCH43 exhibited reliable and reproducible growth in THY broth between two separate experiments (Figure 3.5).

3.3.4 WCH43 infection kinetics in CD-1 mice

A pilot experiment was performed to determine the ability of this strain to colonise the nasopharynx and invade other niches. Six female CD-1 mice, aged five weeks, were challenged IN with 4×10^7 bacteria (determined retrospectively by plating on BA) as described in Section 2.4.3. The level of bacteraemia for each challenged mouse was monitored at 24, 48 and 72 hr following challenge by performing tail bleeds to obtain a sample of blood (which involved making a small incision with a sterilised scalpel along the tail vein and removing an aliquot of blood using heparinised tubing), which was then plated on BA (Figure 3.6A). The surviving mice were euthanased 72 hr after challenge, and nasopharyngeal washes, lung and blood samples were obtained. The blood sample was obtained prior to the commencement of perfusion. The numbers of bacteria in each sample were determined by plating on BA (Figure 3.6B). Approximately 10^5 pneumococci were recovered in the nasopharyngeal wash for all infected mice, indicating that WCH43 is able to colonise CD-1 mice for at least three days. However, the strain caused lung disease



Figure 3.5: WCH43 growth in THY

WCH43, taken from overnight BA plates, was cultured at 37° C in THY media and A_{600} readings were recorded over a 7 hr incubation. The error bars represent SD between two replicate experiments.

Figure 3.6: Pilot experiment: Disease progression in CD-1 mice following IN infection with WCH43

WCH43 (a serotype 4 clinical isolate) was cultured in THY until mid-exponential phase and 6 CD-1 mice were each challenged IN with 4×10^7 bacteria. **A**) depicts the level of bacteraemia observed for each mouse, determined by tail bleeds at 24, 48 and 72 hr post-challenge. One mouse died between 48 and 72 hr time points. **B**) Surviving mice were euthanased at 72 hr and nasopharyngeal washings, lung homogenates and blood samples were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. For both **A**) and **B**), the median is indicated by a bar and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. <40 CFUs.



and bacteraemia in only two out of the five surviving mice (one mouse died between 48 and 72 hr time points). Interestingly, although the level of bacteraemia reached $10^6 - 10^7$ CFU/ml for mice euthanased at 72 hr, the animals did not display the typical prodromal symptoms of sepsis observed in mice challenged IN with D39 or TIGR4 (hunched appearance, poor hygiene, glazed eyes and lethargy). This strain was not studied further due to the low proportion of mice in which nasopharyngeal colonisation progressed to invasive disease.

3.4 Additional studies using serotype 6A challenge strain WCH16

3.4.1 WCH16 infection kinetics in CD-1 mice

Due to the unreliable progression of the type 4 strains TIGR4 and WCH43 from nasopharyngeal colonisation to invasive disease, a serotype 6A strain was examined. Serogroup 6 comprises of serotypes 6A and 6B, which differ from each other by a the nature of the glycosidic linkage between the rhamnose and ribitol moieties in the polysaccharide repeat unit (Robbins *et al.*, 1979). For these experiments, the type 6A clinical isolate WCH16 (obtained from the Women's and Children's Hospital, South Australia) was used. This strain had been characterised previously in our laboratory. It had been shown to both exhibit reliable growth *in vitro*, and is virulent for mice when challenged IP (A.D. Ogunniyi, personal communication).

Six female CD-1 mice, aged five weeks, were challenged IN with 2×10^7 bacteria (determined retrospectively by plating on BA) as described in Section 2.4.3. The level of bacteraemia for each challenged mouse was monitored at 24, 48 and 72 hr following

challenge by performing tail bleeds to obtain a small sample of blood, which was then diluted and plated on BA. The numbers of bacteria present in the blood of each mouse at these time points are shown in Figure 3.7A. Surviving mice at 72 hr post-challenge were euthanased by CO_2 asphyxiation, and nasopharyngeal washes, lung, blood, and brain samples were obtained. The blood sample was obtained prior to the commencement of perfusion. The numbers of bacteria in each sample were determined by plating on BA (Sections 2.4.4 and 2.4.5). A moderate level of bacteraemia (approximately 10⁵ CFU/ml of blood) was observed for four out of the six mice at 24 hr post challenge, and this level was maintained over the three-day trial. One mouse died presumably due to pneumococcal infection, as indicated by the heavy bacterial load in the bloodstream, between the 48- and 72- hr time points. Additionally, one mouse died during tail bleeding at 24 hr and was found to have 4.4×10^5 CFU/ml of blood in its bloodstream. A similar level of nasopharyngeal colonisation was observed for mice infected with WCH16 as for D39, with all animals euthanased at 72 hr having approximately 10^5 pneumococci per nasopharyngeal wash. No bacteria were recovered from the lungs of the mice euthanased at this time point, although all surviving animals were bacteraemic. The level of infection in these three sites is shown in Figure 3.7B. All mice challenged IN with WCH16 developed bacteraemia, albeit at a lower level than for D39. Although mice in the pilot experiment did not possess pneumococci in the lungs at 72 hr post-challenge, bacteria were present in the brain tissue, suggesting the onset of meningitis. The differences in pathogenicity between WCH16 and D39 made it an attractive strain to use in further experiments examining gene expression in these different niches.

Figure 3.7: Pilot experiment: Disease progression in CD-1 mice following IN infection with WCH16

WCH16 (a serotype 6A clinical isolate) was cultured in THY until mid-exponential phase, and 6 CD-1 mice were each challenged IN with 2×10^7 bacteria. **A**) depicts the level of bacteraemia observed for each mouse, determined by tail bleeds at 24, 48 and 72 hr post-challenge. One mouse died during tail bleeding at 24 hr. Additionally, one bacteraemic mouse died between the 48- and 72- hr time points. **B**) Surviving mice were euthanased at 72 hr and nasopharyngeal washings, lung and brain homogenates and blood samples were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. For both **A**) and **B**), the median is indicated by a bar, and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. <40 CFU.





3.5 Comparison between D39 and WCH16 pathogenesis in CD-1 mice

While euthanasing mice and enumerating bacteria present in harvested samples at selected time-points is a quantitative way of examining pathogenesis, it is not possible to follow the disease course within a single mouse. This method does not account for the possibility that each euthanased mouse may have cleared existing disease, developed further disease, or died. To account for mouse-to-mouse variability in susceptibility to infection, the number of mice euthanased at any one time point has to be sufficiently large to be considered representative of the experimental population, which may not necessarily be possible due to technical and ethical considerations. An alternative method of monitoring pathogenesis is to use challenge bacteria expressing a reporter such as GFP or bacterial luciferase (Lux) that can emit a detectable signal in live mice, such that an infection can be monitored in individual mice over a time course (Lee and Camilli, 2000). That is, the course of disease can be followed in individual mice. In this experiment, the *lux* cassette was chosen as a reporter for several reasons. Firstly, the Lux system requires live cells for signal generation, because it utilises reduced flavin mononucleotide as an Therefore, any detectable signal can be almost energy source (Hill et al., 1993). completely attributed to viable bacterial cells. Secondly, GFP imaging suffers highbackground fluorescence, whereas background luminescence is relatively low. Thirdly, GFP is relatively stable, with a half-life of 26 hr (Corish and Tyler-Smith, 1999). Although a derivate has been engineered that possesses a half-life of 6 hr, it is still significantly more stable than Luc and Lux luciferase proteins (Qazi et al., 2001; Thompson et al., 1991). GFP fluorescence also transmits poorly through tissue, making it less suitable for use in a live animal model than luciferase (Lee and Camilli, 2000). To further investigate the pathogenesis of D39 and WCH16, bioluminescent derivates were used to monitor the disease progression of each strain in real time.

3.5.1 Construction of luminescent D39 and WCH16 strains

Pneumococcal strains D39LUX and WCH16LUX were constructed by transforming D39 and WCH16 with pAL2 (Beard et al., 2002), a plasmid possessing a LUX cassette from *Photorhabdus* (Xenorhabdus) luminescens (kindly provided by Dr. Habib Alloush, University of Western England, UK), as described in Section 2.7. The plasmid is shown in Figure 3.8 and contains the entire *luxCDABE* operon under the control of the constitutive ami promoter (Alloing et al., 1990), so is able to luminesce without the addition of exogenous substrate. Furthermore, the enzymes encoded by the luxCDABE operon of *P. luminescens* are functional up to 45° C, which is higher than other bacterial luciferases used in the past (Meighen, 1991). This makes P. luminescens luciferase particularly attractive for in vivo work as bioluminescence can occur over the spectrum of temperatures a pathogen may encounter during infection. E. coli DH5a cells were transformed with pAL2 and cultured in LB broth in the presence of 50 μ g/ml Ery O/N. S. pneumoniae D39 or WCH16 were then transformed with pAL2 extracted from the DH5a cells, in the presence of either 2.5 μ l of 10 μ g/ml CSP-1, or 2.5 μ l of 10 μ g/ml CSP-1 and 5 µl of 10 µg/ml CSP-2, for the respective strains, as described in Section 2.7.2 (Table 2.1). Transformants were identified following O/N growth on BA supplemented with 0.2 μ g/ml Ery, and colonies were then re-streaked on BA containing 50 µg/ml Ery and incubated at 37°C with 5% CO₂ overnight. The plates were examined for bioluminescence using IVIS 100 Imaging System (Xenogen), and luminescent colonies were then cultured in THY in the presence of 50 μ g/ml Ery and glycerol stocks were prepared and stored at -80°C. Glycerol stocks were streaked on BA and examined for luminescence again prior to mouse challenge (Figure 3.9).

NOTE: This figure is included in the print copy of the thesis held in the University of Adelaide Library.

Figure 3.8: pAL2 plasmid map

Arrows indicate the locations and orientations of open reading frames. Gram-positive ribosomebinding sites, located upstream of *luxA*, *luxC*, and *luxE* in the modified *lux* cassette, are represented by shaded regions at the 5' ends of open reading frames. Em-R indicates erythromycin resistance gene (Beard *et al.*, 2002).



Figure 3.9: Bioluminescent pneumococcal strains prior to mouse challenge

Photon emissions from D39LUX and WCH16LUX grown O/N on BA containing 50 µg/ml Ery were visualised using a Xenogen IVIS 100 Imaging System. Clockwise from the top left: D39LUX recovered from the blood of an IN infected CD-1 mouse with bacteraemia 3 days post-challenge; D39LUX colonies following transformation with pAL2; WCH16LUX colonies following transformation with pAL2.

3.5.2 Visualisation of D39 and WCH16 pathogenesis in live CD-1 mice

Mice were IN infected with either D39LUX or WCH16LUX cultures, as described in Section 2.4.3. At 1, 2 and 3 days post-challenge, animals were lightly anaesthetised with Nembutal and examined for luminescence using a 2 to 7 min exposure time on a Xenogen IVIS 100 Imaging System, as described in Section 2.4.6. Based on the presence of detectable photon emission levels using the IVIS 100 Imaging System, animals challenged with D39LUX were observed to establish colonisation in the nose and develop infection of the thoracic and abdominal cavities 2-3 days following challenge, possibly indicating the infiltration of pneumococci into the lungs, blood and liver. Mice exhibiting bacteraemia died within the subsequent 24 hr period. Images typical of D39LUX-infected mice 3 days post-challenge are shown in Figure 3.10.

Conversely, many WCH16LUX-infected mice appeared to develop infection of the upper thoracic cavity 2 days post-challenge. A signal from this region correlated with the presence of bacteria in the bloodstream when an aliquot of blood recovered from the mouse tail vein was plated on BA for enumeration. Generally, bacteraemia persisted at similar levels without causing mice to appear physically ill, or dropped to levels below the limit of detection of the IVIS 100 Imaging System, by 3 days post-challenge. Surviving mice at this time point were maintained for a further 4 days, to examine the outcome of disease. By day 7 post-challenge, 2 mice that exhibited bacteraemia at day 3 had died. Blood was recovered from dead mice and plated for CFU counts and confirmed that mice had died from pneumococcal infection, as they both exhibited high numbers of pneumococci in the bloodstream. This indicates that bacteraemia progresses to sepsis and death, or is resolved, in WCH16-infected mice. In addition, a bioluminescent signal was



Figure 3.10: Bioluminescent images of mice IN infected with D39

3 days following IN infection with D39LUX, CD-1 mice were anaesthetised with Nembutal and monitored for bioluminescence using a Xenogen IVIS 100 Imaging System. Photographs of animals were overlayed with photon emission images. Photon intensity is relative to bacterial numbers. All mice shown here are typical of D39 IN-infected mice observed in these experiments. From the left: A mouse with lung infection; a mouse with infection of the lower abdominal cavity; nasopharyngeal carriage; a mouse exhibiting sepsis, close to death.

detected in the anterior head region of most mice by day 3 following dorsal imaging, but was not apparent in ventral images. It is possible pneumococci were present in the Olfactory lobes or the sinuses. The exact site of infection could be determined in future by sacrificing mice emitting a detectable signal from the area and enumerating bacteria present in the various tissues. Images of WCH16LUX-infected mice 3 days post-challenge are shown in Figure 3.11.

3.5.3 Determination of Ply production by, and haemolytic activity of, D39 and WCH16 cultures

Mice challenged IP with WCH16 bacteria often maintain a constant, sub-septic, level of bacteraemia for several days, whereas mice similarly infected with D39 quickly develop sepsis and die (Sections 3.2.1, 3.4.1 and 3.5.2). The absence of Ply has been previously shown to result in development of chronic bacteraemia instead of overwhelming sepsis (Benton et al., 1995), so it is possible that differences in either specific haemolytic activity or overall Ply expression may contribute to the disparity in virulence between D39 and WCH16. To investigate this further, Ply production and haemolytic activity of each strain were examined. D39 and WCH16 were cultured in THY media to mid-exponential phase ($A_{600} = 0.25$) and WCLs were prepared as described in Section 2.8.1. Relative protein concentrations in the WCLs were estimated using Coomassie staining (Section 2.8.2). To determine the relative haemolytic ability of the two strains, haemolytic assays were performed for D39 and WCH16 WCLs (as described in Section 2.8.4), standardised according to the amount of protein in each WCL. Results from two biological replicate experiments were combined and are presented in Figure 3.12A. The relative amounts of Ply in the same D39 and WCH16 WCLs were determined by Western blot analysis, as described in Section 2.8.3, using antisera specific for the Ply



Figure 3.11: Bioluminescent images of mice IN infected with WCH16

3 days following IN infection with WCH16LUX, CD-1 mice were anaesthetised with Nembutal and monitored for bioluminescence over a time course using a Xenogen IVIS 100 Imaging System. Photographs of animals were overlayed with photon emission images. Photon intensity is relative to bacterial numbers. **A**) one of the two mice that developed a high level of sepsis and lung infection and subsequently died **B**) a mouse with bacteria in the thoracic cavity **C**) a mouse with colonisation of the nasopharynx **D**) a mouse with bacteria in the upper skull, possibly indicating the development of meningitis.



Figure 3.12: Haemolytic activity of, and Ply production by, D39 and WCH16 cultures

A) WCLs prepared from D39 and WCH16 mid-exponential cultures were serially diluted in PBS, and haemolytic activity was determined using human RBCs, as described in Section 2.8.4. Data are mean \pm SD from two biological replicate experiments. The dilution factor at which half of the RBCs were lysed for each strain is indicated by grey lines. **B**) Serial 2-fold dilutions of the same D39 and WCH16 lysates, were separated by SDS-PAGE, electroblotted onto nitrocellulose and reacted with mouse polyclonal antiserum specific for D39 Pdb (pneumolysin toxoid).

toxoid, Pdb (Figure 13.12B). Although D39 cultures appeared to produce slightly more Ply than WCH16 and, accordingly, exhibited a slightly higher haemolytic ability, there was not a significant difference between the two strains (<2-fold). Based on these results, it is unlikely that strain-specific differences in overall production or specific activity of Ply are responsible for the different kinetics of bacteraemia exhibited by D39- and WCH16-infected mice.

3.5.4 Survival of mice IN challenged with D39 or WCH16

In order to examine the relative virulence of D39 and WCH16 in CD-1 mice, INchallenge survival studies were performed. Groups of 15 5-6 week old female CD-1 mice were IN challenged under Nembutal anaesthesia with either $2-3 \times 10^7$ D39 or WCH16 bacteria (Section 2.4.3), and deaths were recorded at 6 hr intervals over a period of 21 days. These results are shown in Figure 3.13. 1 mouse from the WCH16-challenged group did not recover from anaesthesia, and according there are only 14 deaths recorded for this group. For D39, the median survival time was 56 hr, and 2 animals were still alive at the end of the experiment indicating 87% mortality. The medium survival time for mice challenged with WCH16 was nearly twice that observed for D39 (107 hr), with 64% mortality. The survival times of D39- and WCH16- infected mice were compared using the Mann-Whitney test and were found to be significantly different (P < 0.02). Analysis using the Fisher Exact test found no significant difference between overall survival (2/15 and 5/14 mice, respectively). However, further survival experiments using different challenge doses may have detected a statistically significant difference in overall survival rate.



Figure 3.13: Survival time of mice IN-challenged with D39 or WCH16

Groups of 15 CD-1 mice were challenged IN with $2-3 \times 10^7$ D39 or WCH16 bacteria under anaesthesia. 1 WCH16 mouse died during anaesthesia. The median survival time for each group is indicated by a horizontal bar.

3.6 RNA processing and quantitation

3.6.1 mRNA half-lives

To estimate the speed and efficiency needed to harvest and process pneumococcal samples recovered from mice with minimal mRNA degradation, several of the genes to be examined by real-time RT-PCR analysis were selected (ply, psaA, cps2A and nanA) and their mRNA half-lives relative to that of 16S rRNA were determined as follows. Pneumococci were cultured in THY until mid-exponential phase was reached ($A_{600} = 0.3$). Rifampicin (100 µg/ml), an antibiotic that binds to the core and holoenzyme of RNA polymerase (Yarbrough et al., 1976), was added to the culture to prevent further transcription from occurring during the course of the experiment. The amount of rifampicin required to effectively halt transcription was determined empirically by using real-time RT-PCR to determine levels of 16S rRNA extracted over a period of 30 min from several cultures, each possessing different concentrations of rifampicin. A 1 ml aliquot was immediately removed and centrifuged at $15,500 \times g$, for 2 min at 4°C to pellet the bacteria. Nucleic acids were extracted from the pellet following the protocol described in Section 2.5.1. Further aliquots were taken at 2, 4, 8, 12, 16 and 20 min following the addition of rifampicin, and nucleic acids were extracted. All samples were ethanolprecipitated at -80°C for 12 hr. Following DNAse treatment, real-time RT-PCR was used to quantitate mRNA relative to 16S rRNA, and mRNA half-lives were determined by calculating the inverse slope of the line of best fit (Table 3.2). When maintained at room temperature over 20 min, the half-lives of the transcripts examined were relatively short, ranging from 3.6 min (*ply*) to 6.1 min (*nanA*). Experiments conducted in parallel, where cultures were incubated on ice over 20 min, indicated that maintaining pneumococcal cultures at a lower temperature following the addition of rifampicin increased the half-lives

mRNA	Half-life (min) at RT	R ²	Half-life (min) on ice	R ²
ply	3.6	0.973	13.5	0.874
psaA	4.2	0.981	12.9	0.878
cps2A	4.6	0.911	-	-
nanA	6.1	0.995	-	-

Table 3.2: Half-lives of ply, psaA, cps2A and nanA mRNAs

100 μ g/ml rifampicin was added to *S. pneumoniae* D39 THY cultures in mid-exponential growth phase. Cultures were either kept at RT or incubated on ice. Aliquots were taken at regular intervals, and RNA extracts were prepared using the acid phenol-cholorform method described in Chapter 2. Real-time RT-PCR was used to quantitate mRNA relative to 16S rRNA, and mRNA half-lives were determined by calculating the inverse slope of the line of best fit. R² is the coefficient of determination for the line of best fit, and indicates how well the time points fit the line.

of these transcripts by approximately 3 to 4-fold. This experiment reinforced the necessity for speed and low temperature during both the harvesting of bacteria from mice, and the processing of samples.

3.6.2 Optimisation of differential centrifugation

The process by which pneumococci are harvested from soft tissue such as the lungs and brains of mice yields a sample consisting of both bacterial cells and host tissue. Despite the use of the differential centrifugation method described by Ogunniyi et al. (2002) to separate host and pneumococcal cells, total RNA extracted from in vivo harvested samples was still found to contain a high percentage of eukaryotic RNA, as determined by comparing total RNA, as estimated by A_{280/260} readings, and real-time RT-PCR using 16S-specific oligonucleotides and a standard curve generated from an *in vitro*derived template of known concentration. Since the presence of eukaryotic RNA influences the efficiency of prokaryotic template amplification, it was desirable to lower the proportion of contaminating host cells prior to RNA extraction, since all ribonucleic acids are linearly amplified during downstream processes. To determine the effect of different centrifugation speeds on pneumococcal recovery from the lungs and brain of infected mice, D39 cultured in THY broth to mid-exponential phase was separated into five 1 ml aliquots. The brain and lungs of two CD-1 mice were obtained as described previously (Section 2.4.4), and homogenised separately in 2 ml sterile PBS. Each of the homogenates was combined with one pneumococcal culture aliquot. 2 ml sterile PBS was added to the remaining aliquot, as a control to determine maximum yield. Aliquots containing brain or lung homogenates were centrifuged at either $1000 \times g$ or $1500 \times g$ for 4 min at 4°C. Following centrifugation, the supernatants were removed and placed in a new 10 ml tube. Centrifugation was repeated to maximise host cell removal. All cultures

(including the PBS control culture) were subsequently centrifuged at $15,500 \times g$ to pellet pneumococci, and pellets were resuspended in PBS. An aliquot was removed and diluted in PBS, then plated on BA to determine the bacterial yield following centrifugation at different speeds (Table 3.3). It was apparent that a higher initial centrifugation speed resulted in a significantly lower yield of pneumococci. This was particularly evident for pneumococci recovered from lung homogenates, with the bacterial yield decreasing by almost 50% when centrifuged at $1500 \times g$ compared with samples centrifuged at $1000 \times g$.

To examine if a higher centrifugation speed reduced the percentage of eukaryotic cells in the supernatant, samples centrifuged at $1000 \times g$ or $1500 \times g$ were subsequently centrifuged at $15,500 \times g$ and RNA was extracted from tissue/bacterial pellets using the acid-phenol method described previously, then DNase treated for 1 hr at 37° C. Brain and lung samples were enriched for prokaryotic mRNA using MicrobEnrich (Ambion), and then linearly amplified (SenseAmp, Genisphere). It has been estimated that 16S rRNA comprises approximately 27% of total RNA in a bacterial cell (Neidhardt *et al.*, 1996). Based on this assumption, real-time RT-PCR using 16S rRNA oligonucleotide primers was used to estimate the quantity of pneumococcal RNA recovered, which was then compared to the total RNA concentration estimated by A_{260} measurements (Table 3.4). For both lung and brain samples, the higher initial centrifugation speed resulted in a lower bacterial yield and a lower purity of prokaryotic RNA. That is, when centrifuged at $1500 \times g$, pneumococci were more readily pelleted than at $1000 \times g$ for all subsequent studies.
	CFU recovered/ml homogenate	% recovery of pneumococci	
No tissue	2.43×10^{5}	-	
Lung			
$1000 \times g$	1.21×10^{5}	50	
$1500 \times g$	$6.38 imes 10^4$	26	
Brain			
$1000 \times g$	8.63×10^4	36	
$1500 \times g$	5.98×10^4	25	

Table 3.3: Pneumococcal recovery from host tissue following differential centrifugation

S. pneumoniae THY cultures were combined with lung or brain tissue homogenates and then centrifuged at either $1000 \times g$ or $1500 \times g$ for 4 min at 4°C. Supernatants were removed and centrifuged again at the same speed. All cultures (including a culture prepared with PBS instead of homogenate) were centrifuged at $15,500 \times g$ to pellet pneumococci, and pellets were resuspended in PBS. An aliquot was removed and grown on BA for enumeration. The yields from cultures containing homogenised tissue were compared to the control culture and presented as a % recovery of pneumococci.

	Total [RNA] ng/ul ^a	Estimated [bacterial RNA] ng/ul ^b	% bacterial RNA ^c
Lung			
$1000 \times g$	1249	466	37
$1500 \times g$	1015	225	22
Brain			
$1000 \times g$	551	242	48
$1500 \times g$	501	145	26

Table 3.4: Efficiency of centrifugation at different speeds in separating pneumococci from host cells

S. pneumoniae D39 THY cultures were combined with lung or brain tissue homogenates and differentially centrifuged twice at either $1000 \times g$ or $1500 \times g$, as described in the text. All cultures were then centrifuged at $15,500 \times g$ to pellet bacteria. RNA was extracted from pellets then enriched and amplified, as described in the text, and RNA was estimated by either absorbance readings or real-time RT-PCR. ^a Indicates total RNA determined by A_{260/280} readings. ^b Indicates estimated [bacterial RNA] based on real-time RT-PCR quantitation of 16S rRNA compared to a standard curve performed with known concentrations of D39 total RNA, and adjusted on the assumption that 16S rRNA comprises 27% of total RNA in a prokaryotic cell (Neidhardt *et al.*, 1996). ^c Indicates the estimated percentage of total RNA in tissue homogenates following differential centrifugation that was prokaryotic RNA.

3.6.3 Enrichment of prokaryotic RNA and linear amplification

Eukaryotic mRNA, 18S rRNA and 28S rRNA were removed from in vivo RNA samples using MicrobEnrich (Ambion) according to the manufacturer's instructions. By removing eukaryotic RNAs, the relative proportion of prokaryotic RNA in a sample increases. Therefore, the efficiency of subsequent linear amplification reactions is also increased. Initially, RNA samples were purified by using RNAeasy clean-up columns (Qiagen). Samples were then enriched twice using a pull-down method which employed magnetic beads coated with capture oligonucleotides for polyadenylated mRNA, 18S rRNA and 28S rRNA to remove many other RNAs present in the sample; A_{260/280} readings were obtained before and after each enrichment and RNA quantity was calculated. To test the viability of enrichment for bacterial RNA using the MicrobEnrich kit, lungs recovered from D39-infected mice were homogenised (Section 2.4.2 to 2.4.4) and RNA was extracted (as described in Section 2.5.1). 16 µg of the lung-derived RNA preparation was enriched. 25 µg of a control RNA sample (which was a mixture of 25 µg rat RNA and 2 µl E. coli RNA at 1 mg/ml, included in the MicrobEnrich kit) was enriched in tandem. Following the enrichment, the RNA quantity in the lung-derived sample decreased by 40% (returning a yield of 14.9 μ g), which was a notable reduction in eukaryotic RNA using MicrobEnrich. 5.9 µg of control RNA was recovered following enrichment, indicating a 76% reduction in amount of RNA. Considering that bacterial RNA species comprised 7% of the original control RNA, and that MicrobEnrich does not remove small RNAs such as 5S rRNA and tRNA, the control reaction confirmed the kit was functioning efficiently. A second enrichment of both lung-derived RNA and control RNA did not result in a further reduction in RNA quantity (data not shown).

Previously, in vivo gene expression studies for bacteria such as S. pneumoniae had

73

been restricted by inherent difficulties encountered with harvesting sufficient numbers of bacteria from experimental animals and the concomitant low yields of bacterial mRNA. This is particularly problematic in niches where the bacteria exist at low levels, such as during asymptomatic carriage in the nasopharynx. This problem was circumvented by using a novel advanced RNA linear amplification kit, SenseAMP (Genisphere), which uses random primers to synthesise cDNA from prokaryotic RNA. A poly-T tail is then added to the 3' end of cDNA using terminal deoxynucleotidyl transferase. Linear amplification is driven by a T7 phage promoter, which is incorporated at the end of the synthetic poly-T tail (Figure 3.14). A second round of amplification was performed for all samples in order to obtain sufficient quantities of *in vivo*-derived RNA for analysis by real-time RT-PCR using LUX primers. To determine the integrity of two rounds of linear amplification, an RNA sample was prepared from S. pneumoniae D39 grown in THY broth to midexponential phase. The RNA sample was amplified twice using SenseAMP, according to the manufacturer's instructions. The original sample and the twice-amplified sample were analysed for several mRNAs (*pspA*, *ply*, *cps2A* and *piaA*) using real-time RT-PCR. The fold difference between the amount of mRNA present for each gene before amplification and after amplification was determined relative to the internal control 16S rRNA. Statistical analysis (Student's two-tailed *t*-test) of quadruplicate reactions showed that there was no significant difference between the relative proportion of a given mRNA species after the two rounds of amplification (P < 0.05) (Table 3.5). This analysis indicated that amplifying an RNA sample twice reasonably maintains the proportions of mRNAs and, therefore, is appropriate to use prior to mRNA analysis.

Figure 3.14 Linear amplification using SenseAmp (Genisphere)

This diagram depicts the steps involved in linear amplification using SenseAmp. 1) RNA is primed using random oligonucleotides and cDNA is produced. 2) First strand cDNA is tailed with dTTP using Terminal Deoxynucleotidyl Transferase. 3) The T7 template is annealed to the 3' tail of the cDNA (via. a polyA sequence). Klenow enzyme fills in the 3' end of the first strand cDNA to produce a double-stranded T7 promoter. The T7 template contains a blocker to prevent second strand synthesis. 4) Copies of the original RNA molecules are generated. Adapted from Genisphere SenseAmp protocol (2006).

NOTE: This figure is included in the print copy of the thesis held in the University of Adelaide Library.

Gene	Preservation of mRNA proportion after amplifications ^a	<i>P</i> -value ^b
pspA	1.3	NS
ply	1.1	NS
cps2A	1.3	NS
piaA	1.2	NS

Table 3.5: Preservation of relative amounts of mRNA between linear amplifications using SenseAmp

RNA was extracted from *S. pneumoniae* D39 grown in THY and amplified twice, as described in the text. The fold differences in the amount of mRNA present for *pspA*, *ply*, *cps2A* and *piaA* in the original RNA extract and after the two rounds of amplification were determined relative to the internal control 16S rRNA. ^a is the fold- difference in [mRNA] between the original and amplified RNA samples and indicates the magnitude of the bias that two amplifications using SenseAmp (Genisphere) conferred to an RNA sample, where 1.0 is no bias. ^b indicates the *P*-value, which was calculated using a two-tailed Student's *t*-test, where NS denotes a value > 0.05. The *P*-value indicates whether the relative amounts of mRNA present after amplification.

3.7 Discussion

The aim of the work in this chapter was to firstly establish an IN infection model in CD-1 mice for at least two *S. pneumoniae* strains, and secondly to optimise methods for isolation and amplification of bacterial mRNA in sufficient quantities and purity to enable comparison of relative expression of virulence genes in distinct niches.

Although *S. pneumoniae* is a strict human pathogen, several animals are capable of replicating certain aspects of its pathogenesis in the natural host. The chinchilla model is commonly used to study otitis media and, in some instances, nasopharyngeal carriage. Meningitis is often examined using a rabbit model. However, due to the handling, expense and availability considerations, the mouse model of infection is favoured when investigating pneumococcal colonisation, lung infection and sepsis. The pneumococcus is primarily a respiratory pathogen, with the development of invasive disease being largely incidental. Transmission occurs by the aerosol route. Therefore, an intranasal challenge route is considered to more accurately represent natural acquisition of the pneumococcus, than other commonly used challenge routes such as IP.

Wu *et al.* (1997) described a murine IN infection model, applicable to a number of different mouse strains. In the present study, this model was adapted to CD-1 mice, such that pneumococci can be harvested from several niches in numbers sufficient for RNA analysis. The time between euthanasia of each mouse and the snap-freezing of harvested bacterial pellets was approximately 15 to 20 min, which is 4 to 5 times the half-life at room temperature of the mRNA examined in this chapter. Maintaining bacterial cultures on ice over a time-course increased the half-lives of the transcripts to 13 to 14 min. This emphasised the need for a model where bacterial samples can be harvested, chilled and

processed quickly in order to maintain the integrity of the RNA, and speed was taken into consideration when optimising downstream processes such as differential centrifugation. The use of blood perfusion with ice-cold PBS prior to removal of the lungs and brain rapidly chilled these organs, and also ensures that pneumococci present in the lungs or brain samples possess minimal contamination with bacteria present in the blood. The efficiency of this process was demonstrated by the observation that mice with high levels of bacteraemia did not always contain detectable numbers of pneumococci in the lungs and brain after perfusion.

Stringent screening methods were employed for determining the presence of contaminating bacteria (such as oral streptococci) in the samples harvested from each niche, which is particularly important considering that visual recognition of contaminating bacteria is not always possible. A small proportion of CD-1 mice were found to possess E. faecalis in their lungs following infection with S. pneumoniae. Mice challenged IN with pneumococci were found to possess much higher numbers of enterococci in lungs than control mice challenged with PBS, indicating that pneumococcal infection augments the spread of *E. faecalis*. Although *E. faecalis* is not generally a respiratory pathogen, it may be opportunistic. Following infection with pneumococci, epithelial damage and decreased ciliary beating may reduce the clearance of enterococci that have been aspirated into the lungs incidentally. Another contributing factor could be the poor hygiene exhibited by ill mice. In addition to visual identification difficulties and contaminating 16S rRNA, it is possible that the presence of E. faecalis would influence the course of pneumococcal infection. As a consequence, samples harvested from mice possessing contaminants of any nature were excluded from further analysis. The isolation of contaminating bacteria that appears visually similar to S. pneumoniae when grown on BA and is resistant to gentamicin, reinforces the requirement for stringent testing of bacteria isolated from laboratory animals during *in vivo* experimentation.

Strains D39 (a serotype 2 strain) and WCH16 (a serotype 6A strain) were chosen to examine mRNA expression changes. Both D39 and WCH16 were shown to effectively colonise the nasopharynx for up to 72 hr, and cause invasive disease in the CD-1 model described. D39 caused lung disease and sepsis, whereas WCH16 progressed to the blood and brain without establishing significant infection in the lungs. Additionally, WCH16 was able to persist at a moderate level in the bloodstream over 72 hr, and many mice infected with WCH16 also developed meningitis. To further investigate the pathogenesis of the two strains, CD-1 mice were IN-infected with either a D39 or WCH16 strain containing a plasmid constitutively expressing bacterial luciferase (Table 2.1). Using a Lux reporter, the same mouse can be imaged at several time-points during infection, conferring a significant advantage over models requiring mice to be euthanased at each time point in order to enumerate bacterial loads. This allows pathogenesis to be studied non-invasively and in real time, albeit at the expense of sensitivity (Francis et al., 2000). Considering the benefits and limitations of both methods, using bioluminescence data and organ harvesting studies in tandem is likely to give a more comprehensive picture of pathogenesis than independently. Monitoring the mice over a 7-day period demonstrated that challenge of CD-1 mice with D39LUX established systemic infection followed by death within 24 hr of these niches being infected. In contrast, WCH16 generally caused what appeared to be infection of the brain and blood in the absence of lung infection, perhaps indicating translocation across the nasopharyngeal epithelia. This correlated with the results obtained by dissecting mice and enumerating bacteria present in these niches in Sections 3.2 and 3.4. Mice IN challenged with WCH16 had a mean survival time approximately 50 hr longer than D39. The greater survival time of WCH16-infected mice may be largely attributable to WCH16 being less proficient in causing sepsis following establishment of bacteraemia than D39. This disparity between serotypes was also observed in real-time using a *lux* reporter, with WCH16LUX-infected mice exhibiting bacteraemia surviving several days and, in some cases, clearing the blood infection.

In contrast to D39 and WCH16, the serotype 4 strains examined (TIGR4 and WCH43) showed poor progression to bacteraemia by 72 hr, despite previous reports of TIGR4 virulence and high levels of sepsis in outbred MF1 mice (Ibrahim *et al.*, 2004; Ibrahim *et al.*, 2005). TIGR4 also showed inconsistent growth *in vitro*. Because of this, these strains were not examined further.

The separation of pneumococci from lung and brain homogenates has been refined in this chapter, allowing for greater efficiency during downstream processing. A higher initial centrifugation speed resulted in a lower yield of bacteria from the tissue suspension, and also a lower ratio of prokaryotic: eukaryotic RNA. These results indicated that an increased initial centrifugation speed did not result in the removal of a higher percentage of eukaryotic cells from the supernatant, but did cause a greater number of pneumococci to be pelleted and hence lost. Accordingly, initial centrifugation conditions were standardised at $1000 \times g$ for 6 min. Following bacterial differential centrifugation, pneumococcal RNA recovered from bacteria from the lungs and brain was enriched and amplified. All RNA is tailed and amplified using SenseAmp. Therefore, it was desirable to increase the efficiency of linear amplification for pneumococcal RNA by decreasing the proportion of contaminating eukaryotic RNA. MicrobEnrich, a kit using a pull-down method with magnetic beads coated with capture oligonucleotides for polyadenylated mRNA, 18S rRNA and 28S rRNA, was employed to reduce eukaryotic RNAs present in the sample. While a single enrichment was found to reduce A_{280/600} readings compared to the original sample, subsequent enrichments did not, indicating that one enrichment was sufficient to remove the majority of eukaryotic RNA species in the pre-amplification sample. Subsequent linear amplification using SenseAmp was shown to preserve the relative 16S rRNA: mRNA ratio such that any fold differences observed between niches for a particular mRNA species, or between species within a niche, could be considered to be a reasonable approximation of real levels and not an artefact of biased amplification. Using the above procedures, sufficient mRNA could be obtained even from *in vivo* niches where pneumococci persist in low numbers to perform quantitative RNA assays.

In this chapter, a CD-1 IN challenge model and downstream RNA extraction and processing techniques are described, establishing conditions whereby *S. pneumoniae* RNA can be extracted in sufficient amounts to examine relative virulence gene expression *in vivo*, as will be described in the following chapters. Moreover, this model allows the transcriptional analysis of bacteria recovered from various *in vivo* niches in a single animal model, including pneumococci involved in nasopharyngeal colonisation, which has not been directly examined in the literature due to technical limitations.

CHAPTER 4 – Analysis of key virulence gene expression (for D39 and WCH16) *in vivo* by real-time RT-PCR

4.1 Introduction

S. pneumoniae is able to proliferate in a number of niches within its host. Most commonly, the pneumococcus exists asymptomatically in the nasopharynx. From this niche, it can translocate up the Eustachian tube into the middle ear and cause otitis media or be aspirated into the lungs and cause pneumonia. Bacteria can enter the bloodstream either from the lungs, or directly from the nasopharynx. Pneumococci can then cross the blood-brain barrier and proliferate in the brain causing meningitis (McCullers and Tuomanen, 2001). It is likely that many genes are regulated to enable the pneumococcus to survive in these varied environmental niches, but such regulation is poorly understood. Since transcription and translation are often coupled in prokaryotes (Campbell, 1996), using quantitative gene expression techniques such as real-time RT-PCR to assay mRNA levels may give an indication of relative protein levels and, therefore, reflect functional regulation. Previously, technical limitations have restricted quantitative experiments

examining pneumococcal gene expression *in vivo*, particularly during nasopharyngeal colonisation. In this chapter, the mRNA levels for specific pneumococcal virulence factors were assayed from resident bacteria in the nasopharynx, lungs, blood and brain using real-time RT-PCR. Two strains were examined; the serotype 2 strain D39 and the serotype 6A strain WCH16, both of which had been shown to reliably colonise and cause disease in a CD-1 IN infection model (Chapter 3).

4.2 Differential expression of D39 and WCH16 virulence genes *in vivo*

4.2.1 Recovery of bacteria from CD-1 mice

Using the model established in Chapter 3, D39 and WCH16 bacteria were recovered from various *in vivo* niches. CD-1 mice were anaesthetised and challenged with D39 or WCH16 as described in Sections 2.4.2 and 2.4.3. This experiment was performed twice for both D39 and WCH16. For the D39 experiments (D39.1 and D39.2) the challenge dose for each mouse was 1.15×10^7 and 2.8×10^7 , respectively. In D39.1, 25 mice were challenged, and 10 mice were sacrificed at the 24 hr time point. In D39.2, 18 mice were challenged and seven were sacrificed at 24 hr. The remaining mice were euthanased at 72 hr post-challenge. The experiments performed in Chapter 3 indicated nasopharyngeal colonisation was established for both D39 and WCH16 by the latter time point and bacteria were also present in other niches, indicating disease. One mouse in D39.1, and four mice in D39.2, died prior to the 72 hr time point. The surviving animals appeared ill, and displayed typical symptoms of sepsis such as hunched appearance, lethargy, glazed eyes and poor hygiene. Bacteria were recovered from the nasopharynx,

lungs and blood of euthanased mice and prepared for RNA extraction (Section 2.4.4). Samples were also plated for CFU enumeration (Section 2.4.5). Typically, 10^6 to 10^8 bacteria/ml of blood were recovered from mice that were sacrificed at 72 hr. Almost all sacrificed mice were colonised with approximately 10^5 bacteria per nasopharynx and possessed pneumococci in their lungs at both time points, indicating the establishment of both carriage and lung disease. The CFU recovered from each of the niches for D39.1 and D39.2 are shown in Figures 4.1A and 4.1B, respectively.

For the WCH16 experiments (WCH16.1 and WCH16.2), 25 CD-1 mice were anaesthetised and challenged with 1.7×10^7 and 3×10^7 bacteria, respectively. At 24 hr and 72 hr post-challenge, bacteria were recovered from the nasopharynx, the lungs, the blood and the brain of 10 and 15 euthanased mice, respectively, and prepared for RNA extraction (Section 2.4.4). In addition to the three niches that were examined for D39infected mice, the brains were also excised from mice challenge with WCH16, as this strain had been observed in Chapter 3 and by luminescence work in Chapter 4 to reach this niche rapidly following IN infection. In contrast to D39-infected animals, mice euthanased at 72 hr appeared relatively healthy, generally exhibiting none of the symptoms commonly associated with sepsis. Accordingly, only moderate numbers of bacteria were recovered from the bloodstream at this time point, and no mice died during the course of either experiment. All mice except one were colonised at a level of approximately 10⁵ bacteria per nasopharynx in both experiments, which is consistent with that seen in the pilot experiment. Many mice also possessed low numbers of bacteria in the brain at both 24 and 72 hr post-challenge. Because all the mice that exhibited early meningitis also had bacteraemia, it is likely that pneumococci disseminated to the brain via the bloodstream. Alternatively, bacteria may have translocated from the nasopharynx to the brain by axonal

Figure 4.1: D39 pathogenesis during IN challenge experiments D39.1 and D39.2

D39 was cultured in THY until mid-exponential phase and CD-1 mice were challenged IN. Mice were euthanased at 24 and 72 hr post-challenge, and nasopharyngeal washings (N), lung homogenates (L) and blood samples (B) were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. The time-point is indicated in subscript. **A)** and **B)** represent data obtained from the two independent experiments D39.1 and D39.2 (challenge doses 1.15×10^7 and 2.8×10^7 CFU respectively). The median for each time/niche is indicated by a bar, and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. <40 CFU.



transport through olfactory nerves (van Ginkel *et al.*, 2003). The CFU recovered from each of the niches for WCH16.1 and WCH16.2 are shown in Figures 4.2A and 4.2B respectively.

4.2.2 Extraction of *in-vivo* derived RNA

After harvesting the pneumococci from CD-1 mice as described in Section 4.2.1, RNA was extracted from the bacterial pellets using the acid-phenol method (Section 2.5.1), and treated with DNase to remove DNA. Four to five samples per niche at each time point were pooled, based on the absence of contaminating bacteria, the number of CFU harvested from the mouse, the amount of RNA present in the sample (determined by $A_{260/280}$ readings), and the absence of contaminating DNA (based on visual appearance on an agarose gel following PCR with and without reverse transcriptase using 16S rRNA primers). Subsequently, RNA samples obtained from the lungs and the brain were enriched for prokaryotic RNA using MicrobEnrich (Ambion), and linearly amplified twice for prokaryotic mRNA as described in Sections 2.5.2 and 2.5.3. Recombinant RNAsin RNAse inhibitor was added, and RNA was stored at -80 °C until use.

4.2.3 Preparation of RNA from *in vitro*-cultured pneumococci

For comparative purposes, an *in vitro* pneumococcal RNA sample was prepared against which all *in vivo*-recovered samples were compared. D39 or WCH16 pneumococci were cultured in THY to $A_{600} = 0.25$, a medium and optical density identical to the pneumococcal challenge cultures, and RNA samples were prepared as described in Section 2.5.1. The CFU/ml in the culture was determined retrospectively by plating dilutions on BA, and was found to be extremely similar to the CFU/ml of the challenge cultures. In this way, a comparison could be made between pneumococcal gene expression prior to mouse

Figure 4.2: WCH16 pathogenesis during IN challenge experiments WCH16.1 and WCH16.2

WCH16 was cultured in THY until mid-exponential phase and CD-1 mice were challenged IN. Mice were euthanased at 24 and 72 hr post-challenge, and nasopharyngeal washings (N), lung homogenates (L), blood samples (B) and brain homogenates (M) were diluted and plated on BA for enumeration. Data are CFU per niche for nasal wash and lungs, and CFU/ml for blood. The time-point is indicated in subscript. **A**) and **B**) represent data obtained from the two independent experiments WCH16.1 and WCH16.2 (challenge doses 1.7×10^7 and 3×10^7 CFU respectively). The median for each time/niche is indicated by a bar, and the limit of detection (40 CFU) is shown by a dotted line. ND denotes no growth, i.e. < 40 CFU.



challenge, and several days post-challenge.

4.2.4 Quantification of mRNAs using real-time RT-PCR

In this study, the differential expression of well characterised virulence genes *cbpA*, pspA, ply, psaA, cps2A, piaA, nanA and spxB was examined. These genes were chosen based on their putative roles in virulence and, in some instances, potential of their proteins as vaccine candidates. The abundance of each transcript present in amplified RNA recovered from nasopharyngeal, lung and blood-borne pneumococci was quantitated by real-time RT-PCR (Section 2.6.2.1). Gene-specific LUX fluorogenic primer sets labelled with JOE were designed using the Invitrogen primer designer software, employing primers specific for 16S rRNA as an internal control (Table 2.2). The amplicons for the LUX primer pairs used in this study were between 81 and 166 bp (with most amplicon sizes being 90-110 bp) and the T_m for all LUX primers were similar. 16S rRNA was chosen for use as an internal control in these experiments from a group of several other housekeeping genes, as it could not be determined if other candidates would maintain the same expression level in the niches examined in this chapter and during the different conditions presented by disease. 16S rRNA is commonly used as a bacterial internal control in quantitative RNA assays, as it is a reliable estimate of defined amounts of RNA under different environmental conditions (Lion, 2001; Tasara and Stephan, 2007). For 16S rRNA, ply, psaA, piaA, nanA and spxB, the same LUX fluorogenic primers were used for analysis of D39 and WCH16 mRNAs, due to high nucleotide sequence homology between the strains for the target genes. Serotype- or strain-specific primers were designed for cbpA, pspA and cpsA (Table 2.2). Quantitative fold differences between niches for each transcript were determined using the $2^{-\Delta\Delta Ct}$ method described by Livak and Schmittgen (2001). Amplification data for each gene were then standardised using values obtained for the 16S rRNA control, and presented as a relative fold increase/decrease between niches (described in Section 2.6.2.2). Data for each independent experiment were analysed relative to the level of the respective transcript in *S. pneumoniae* D39 grown in THY broth (Figures 4.3 and 4.4). The fold-differences in relative mRNA between niches for each experiment are presented in Table 4.1 and 4.2. Likewise, data obtained from the two WCH16 experiments are presented as transcript levels in each niche relative to *in vitro* (Figures 4.5 and 4.6), and the fold-differences in relative mRNA levels between niches are presented in Tables 4.3 and 4.4.

The trends in gene expression for *cbpA*, *pspA*, *cps2A*, *piaA*, *ply*, *nanA* and *spxB* were reproducible at 72 h post-challenge between the two replicate experiments for D39. *cbpA* mRNA was detected at the highest levels in bacteria recovered from the nasopharynx and lungs, and at lower levels in bacteria from the blood. Similarly, the level of *pspA* mRNA was highest in bacteria harvested from the nasopharynx and blood, with lower levels detected from pneumococci recovered from the lungs. In contrast, cps2A was expressed at a relatively constant level in the nasopharynx, lungs and blood. The levels of both *nanA* and *spxB* transcripts were higher in bacteria recovered from the nasopharynx than the lungs or blood. However, *psaA* did not show a consistent difference in expression between niches. The combined data from both *in vivo* experiments and an *in vitro* sample are also depicted as adjusted mRNA concentration, by comparing the Ct values for each gene relative to that of 16S rRNA in the same sample (Figure 4.7). The results indicate that the mRNA of *spxB* was the most abundant *in vitro*, followed by that of *ply* and *cps2A*, with nanA mRNA being the least abundant. In the nasopharynx, the mRNA of ply, followed by that of cps2A, pspA and psaA, was highly abundant normalised against 16s rRNA, whereas nanA mRNA was the least abundant. In the lungs, however, the mRNA of cps2A was the



Figure 4.3: Expression of D39 genes in vivo by real-time RT-PCR in D39.1

Following IN challenge with D39, mice were sacrificed at 72 hr post-challenge and RNA preparations were obtained from pneumococci in the nasopharynx, lungs and blood. Samples recovered from the same niche were pooled, as described in the text. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Data indicate the fold differences between mRNA present in different niches relative to pneumococci cultured *in vitro*. Error bars represent the SEM.

Gene	Fold difference					
	Nose/Lung	Nose/Blood	Lung/Blood			
cbpA	1.2	2.2*	1.8*			
pspA	6.5**	2.0*	0.3*			
ply	1.3	1.8**	1.4			
psaA	2.5	2.4*	1.0			
cps2A	0.67	1.0	1.4			
piaA	3.1**	2.3*	0.71			
nanA	22.8**	56.6***	2.5*			
spxB	4.2**	5.5**	1.3			

Table 4.1: Relative amounts of mRNA present in different niches for D39.1

This table indicates the relative fold differences between niches in experiment D39.1 (Figure 4.3). 16S rRNA was used as an internal control, and the quantitative fold increase or decrease for each transcript between niches was determined using the $2^{-\Delta\Delta Ct}$ method. The value represents the relative amount of mRNA in the first niche compared to the second. Data were analysed using Student's *t*-test; *, P < 0.05; **, P < 0.01; ***, P < 0.005.



Figure 4.4: Expression of D39 genes in vivo by real-time RT-PCR in D39.2

Following IN challenge with D39, mice were sacrificed at 72 hr post-challenge and RNA preparations were obtained from pneumococci in the nasopharynx, lungs and blood. Samples recovered from the same niche were pooled, as described in the text. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Data indicate the fold differences between mRNA present in different niches relative to pneumococci cultured *in vitro*. Error bars represent the SEM.

Gene	Fold difference					
	Nose/Lung	Nose/Blood	Lung/Blood			
cbpA	3.7**	6.3***	1.7			
pspA	3.8*	1.5	0.4*			
ply	2.8***	1.5*	0.67			
psaA	1.0	0.91	1.0			
cps2A	0.71	1.2	1.6			
piaA	3.1**	1.9*	0.63*			
nanA	15.3**	11.1***	0.71			
spxB	5.6*	4.1*	0.71			

Table 4.2: Relative amounts of mRNA present in different niches for D39.2

This table indicates the relative fold differences between niches in experiment D39.2 (Figure 4.4). 16S rRNA was used as an internal control, and the quantitative fold increase or decrease for each transcript between niches was determined using the $2^{-\Delta\Delta Ct}$ method. The value represents the relative amount of mRNA in the first niche compared to the second. Data were analysed using Student's *t*-test; *, *P* < 0.05; **, *P* < 0.01; ***, *P* < 0.005.



Figure 4.5: Expression of WCH16 genes in vivo by real-time RT-PCR in WCH16.1

Following IN challenge with WCH16, mice were sacrificed at 72 hr post-challenge and RNA preparations were obtained from pneumococci in the nasopharynx, lungs, blood and brain. Samples recovered from the same niche were pooled, as described in the text. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Data indicate the fold differences between mRNA present in different niches relative to pneumococci cultured *in vitro*. Error bars represent the SEM. An asterisk indicates that data could not be obtained.

Gene	Fold difference					
	Nose/Lung	Nose/Blood	Nose/ Brain	Lung/Blood	Lungs/Brain	Blood/Brain
cbpA	0.5*	27.7**	1.0	56.8**	1.2	0.021***
pspA	2.6**	1.8*	6.0**	0.67*	2.6**	3.7***
ply	2.7*	3.7**	0.62**	1.4*	0.23***	0.17***
psaA	0.67	0.56	27.0***	0.83	41.0***	47.3**
cps2A	2.0**	69.7***	1.1	142.7***	2.2**	0.015***
piaA	3.0**	0.91	0.62	0.29**	0.21**	0.70
nanA	1.9	2.0	NA	1.0	NA	NA
spxB	8.5**	25.7***	5.6**	3.0	0.66	0.22**

Table 4.3: Relative amounts of mRNA present in different niches for WCH16.1

This table indicates the relative fold differences between niches in experiment WCH16.1 (Figure 4.5). 16S rRNA was used as an internal control, and the quantitative fold increase or decrease for each transcript between niches was determined using the $2^{-\Delta\Delta Ct}$ method. The value represents the relative amount of mRNA in the first niche compared to the second. Data were analysed using Student's *t*-test; *, *P* < 0.05; **, *P* < 0.01; ***, *P* < 0.005. NA denotes data not available.



Figure 4.6: Expression of specific WCH16 genes in vivo by real-time RT-PCR in WCH16.2

Following IN challenge with WCH16, mice were sacrificed at 72 hr post-challenge and RNA preparations were obtained from pneumococci in the nasopharynx, lungs, blood and brain. Samples recovered from the same niche were pooled, as described in the text. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Data represent the fold differences between mRNA present in different niches relative to pneumococci cultured *in vitro*. Error bars represent the SEM. An asterisk indicates that data could not be obtained.

Gene	Fold difference					
	Nose/Lung	Nose/ Blood	Nose/Brain	Lung/ Blood	Lungs/Brain	Blood/Brain
cbpA	1.7	NA	NA	NA	NA	NA
pspA	0.53*	0.91	0.31*	1.7*	0.59	0.34*
ply	1.8	2.0	1.0	1.0	0.53*	0.5*
psaA	0.33*	0.27**	705.2**	0.83	0.53**	2598***
cps2A	0.5	0.48**	0.24**	1.0	210*	0.5*
piaA	3.5*	16.3**	0.77	4.7**	0.21**	0.045**
nanA	1.0	1.0	NA	1.0	NA	NA
spxB	8.2*	37.5*	0.0045*	4.6*	.0007*	0.00015*

Table 4.4: Relative amounts of mRNA present in different niches for WCH16.2

This table indicates the relative fold differences between niches in experiment WCH16.2 (Figure 4.6). 16S rRNA was used as an internal control, and the quantitative fold increase or decrease for each transcript between niches was determined using the $2^{-\Delta\Delta Ct}$ method. The value represents the relative amount of mRNA in the first niche compared to the second. Data were analysed using Student's *t*-test; *, *P* < 0.05; **, *P* < 0.01; ***, *P* < 0.005. NA denotes data not available.



Figure 4.7: Expression of D39 genes in vivo by real-time RT-PCR

Following IN challenge with D39, mice were euthanased at 72 hr post-challenge and RNA preparations were obtained from mice containing pneumococci in the nasopharynx, lungs and blood. Samples recovered from the same niche were pooled. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Ct data for each mRNA was normalised against 16S rRNA Ct values and adjusted as described in the text. Data from the two independent experiments (D39.1 and D39.2) were pooled, and are presented in the graph. The error bars represent the SEM.

most abundant, followed by that of *ply* and *psaA*. Again, *nanA* mRNA was the least abundant. In the blood, the mRNAs of *cps2A*, *ply*, *pspA*, *psaA* and *piaA* were present in high quantities, *cbpA* mRNA was present at a lower level, while *nanA* mRNA was the least abundant.

The WCH16 data from these two experiments indicate largely consistent trends in gene expression for ply, psaA cbpA, nanA and spxB at 72 hr post-challenge. SpxB mRNA levels were highest in bacteria recovered from the nasopharynx and the brains of mice, with substantially lower levels observed from bacteria recovered from the other niches. In contrast, the levels of *nanA* mRNA were expressed at a similar level in the nasopharynx, lungs, bloodstream and brain, and also in bacteria cultured *in vitro*. Levels of *cbpA* mRNA present in pneumococci from the blood and the brain in WCH16.2 and nanA mRNA harvested from the brain in both experiments were below the limit of detection for realtime RT-PCR, as indicated by a small cycle number difference observed between the sample and the no template control (NTC) for these transcripts. Despite the lack of data for these niches, it was still apparent that *cbpA* mRNA levels were detected at highest levels in bacteria recovered from the nasopharynx and lungs, compared to bacteria from the blood. In addition to Figures 4.5 and 4.6, which present data from each individual experiment, the combined data from both *in vivo* experiments and two *in vitro* samples are also presented as adjusted mRNA concentrations (Figure 4.8). Overall, the trends in gene expression observed for WCH16 were less consistent between experiments than for D39. This discrepancy could be due to the inconsistent level of bacterial infection observed for WCH16.1 and WCH16.2. A further WCH16 infection experiment needs to be conducted to clarify the results.



Figure 4.8: Expression of specific WCH16 genes in vivo by real-time RT-PCR

Following IN challenge with WCH16, mice were euthanased at 72 hr post-challenge and RNA preparations were obtained from mice containing pneumococci in the nasopharynx, lungs, blood and brain. Samples recovered from the same niche were pooled. Following RNA purifications and amplifications, mRNA was assayed by real time RT-PCR as described in the text. Ct data for each mRNA was normalised against 16S rRNA Ct values and adjusted as described in the text. Data from the two independent experiments (WCH16.2 and WCH16.2) were pooled, and are presented in the graph. The error bars represent the SEM. An asterisk indicates that data could not be obtained for one or both experiments.

Niche-specific gene expression was also examined for bacteria recovered 24 hr post-challenge using the IN infection model described in Chapter 3 (data not shown). However, although data for some genes showed similar expression trends at this time point to those observed at 72 hr, the differential expression between niches was lower. The standard deviation between experiments was far greater for mice euthanased at the 24 hr time-point than at 72 hr. These issues can probably be attributed to the lower numbers of bacteria harvested from some niches at 24 hr, which may have led to inaccuracies in realtime RT-PCR quantitation. Small quantitation errors may be exacerbated for very low RNA samples, giving an exaggerated measure of up- or down-regulation. Additionally, it is likely that bacteria had been present in each niche for less time than pneumococci harvested at 72 hr and, therefore, may not have completely adjusted their transcriptional profiles, nor have elicited the same degree of host response as for the later time-point.

4.3 Discussion

S. pneumoniae can effectively colonise the nasopharynx and is also able to proliferate and cause disease in numerous other niches. The putative *in vivo* roles of several pneumococcal proteins involved in nasopharyngeal carriage or disease have been inferred using *in vitro* studies, as described in Chapter 1. The differential expression of several virulence genes between bacteria grown *in vitro* and bacteria harvested from a sepsis model has been previously reported (Ogunniyi *et al.*, 2002). However, at the commencement of this work, studies had not examined quantitative differences in specific virulence gene mRNA levels between niches of infection. The experiments presented in this chapter use the single, continuous infection model established in Chapter 3 to identify differential expression of key pneumococcal virulence genes during asymptomatic carriage

and disease. The genes chosen encode the virulence factors CbpA, PspA, Ply, PsaA, CpsA, NanA, PiaA and SpxB, all of which appear to play important roles in pneumococcal pathogenesis.

4.3.1 Gene expression during nasopharyngeal colonisation

Colonisation of the nasopharynx is considered a prerequisite for disease and provides a base niche from where pneumococci can progress to the lungs, bloodstream, middle ear or brain and cause disease. Adherence in the upper respiratory tract is considered to be facilitated by interactions between CbpA and hpIgR, an interaction that also promotes invasion of nasopharyngeal epithelial cells (Rosenow et al., 1997; Zhang et al., 2000). A second choline binding protein, PspA, interacts with the iron chelator apolactoferrin, inhibiting apolactoferrin-mediated killing of pneumococci in the nasopharynx and allowing carriage to be established (Shaper et al., 2004). Both CbpA and PspA have been proposed as vaccine candidates due to their immunogenicity, contributions to multiple stages of pathogenesis, and exposure outside the pneumococcal capsule (Brooks-Walter et al., 1999; Hammerschmidt et al., 1997; McCool et al., 2002; Ogunniyi et al., 2001; Rosenow et al., 1997; Shaper et al., 2004; Talkington et al., 1991; Tu et al., 1999; Yother and Briles, 1992; Zhang et al., 2000). In the experiments presented in this chapter, *cbpA* was expressed more highly in the nasopharynx and the lungs than in the bloodstream when in vivo-derived RNA extracts were subjected to real-time RT-PCR analysis. The *cbpA* mRNA levels were also dramatically higher in these niches compared to pneumococci cultured in THY broth, suggesting an increased requirement for CbpA in the nasopharynx and lungs than for *in vitro* growth. This was apparent for both D39 and WCH16, and is consistent with a role for CbpA in nasopharyngeal colonisation. The finding that *cbpA* mRNA is higher in colonising pneumococci than bacteria recovered from the blood is also consistent with a recent study demonstrating that CbpA is not required for survival in the bloodstream (Orihuela *et al.*, 2004a). *pspA* mRNA levels were also elevated in colonising pneumococci compared to bacteria recovered from the lungs or cultured *in vitro* for D39 and one WCH16 experiment in the current study. These results agree with *in vitro* microarray data obtained by Orihuela *et al.* (2004b) that indicated *pspA* transcription is up-regulated in pneumococci co-incubated with Detroit-562 nasopharyngeal cells compared to when bacteria grown in tissue culture medium alone.

The pneumococcus possesses multiple proteins that are capable of reducing competing microflora in the nasopharynx, allowing it to effectively colonise this niche. The neuraminidase NanA and pyruvate oxidase SpxB are two such proteins. NanA reduces the carriage of other colonising bacteria by desialylating glycoconjugates on their surface (Shakhnovich et al., 2002). It also facilitates pneumococcal adherence by modifying glycoconjugates on host epithelium, and has been shown to promote long-term carriage (King et al., 2004; Tong et al., 2000). SpxB catalyses the formation H₂O₂, which inhibits the growth of competing microflora such as Haemophilus influenzae and Neisseria meningitidis, and may also enhance pneumococcal adherence to the nasopharyngeal epithelium and progression to the lungs (Orihuela et al., 2004a; Pericone et al., 2000; Spellerberg et al., 1996). In the D39 experiments conducted in this chapter, both nanA and spxB transcripts were present in highest amounts for pneumococci recovered from the nasopharynx rather than from the lungs or blood, which is consistent with their putative roles in carriage and transition to the lungs. These data are also consistent with a microarray study conducted by King et al. (2004) that associated increased nanA levels with transparent pneumococci (the colony opacity phenotype commonly isolated from the nasopharynx). Similarly, Pericone *et al.* (2000) reported that spxB is expressed at higher
levels in oxygen- and CO₂-rich environments, conditions which are prevalent in the nasopharynx. WCH16 data for *spxB* mRNA were similar to D39, with higher amounts present for colonising pneumococci than bacteria harvested from other niches. However, the levels of WCH16 *nanA* transcript remained relatively constant in all the niches examined and the *in vitro* sample, perhaps indicating that the role of *nanA* in establishing nasopharyngeal colonisation is less critical for this strain.

Carriage is a dynamic process, requiring a balance to be maintained between proliferation and evasion of the host's innate defences. Ply enhances disease and inflammation through its cytotoxic activity, activation of the complement pathway and a TLR-4-dependent inflammatory response, and ability to elicit the production of numerous pro-inflammatory cytokines (Boulnois et al., 1991; Braun et al., 1999; Cockeran et al., 2002; Houldsworth et al., 1994; Malley et al., 2003; Paton et al., 1984). Several studies have indicated that Ply is important in the development of sepsis, with mutants lacking Ply showing reduced replication and survival in the bloodstream of infected animals (Benton et al., 1995; Berry et al., 1989b; Berry et al., 1999). Considering the roles of Ply in inflammation and disease, one might expect it to be expressed more highly in the lungs and the bloodstream. In fact, the data presented in this chapter indicated that *ply* mRNA level was higher in the nasopharynx than the other niches examined for both D39 and WCH16. This is an indication that even though Ply is important in invasive disease, it may also be required to promote carriage. The contribution of Ply to nasopharyngeal colonisation is unclear, with some groups reporting that carriage in a mouse model can occur in the absence of Ply with little attenuation (Rayner et al., 1995; Rubins et al., 1998), while findings from another study show that a Ply-negative mutant is cleared from the nasopharynx within 4 hr post-challenge (Kadioglu et al., 2002). A recent study implicates

Ply in promoting the level of colonisation rather than the initial establishment of carriage, with a Ply insertion-duplication mutant of D39 colonising a similar number of mice compared to the WT D39, but at a lower level (Ogunniyi *et al.*, 2007). The *ply* expression data presented in this chapter support a role for Ply in the nasopharynx. As Ply lacks an N-terminal signal sequence and its release from the pneumococcal cell most often occurs following LytA-mediated autolysis (Berry *et al.*, 1989a), it is also possible that levels of free Ply in the blood may be indirectly modulated by the extent of autolysis in that niche. Alternatively, Ply production may be regulated at a translational level.

In addition to *cbpA*, *pspA*, *nanA*, *spxB* and *ply*, the results in this chapter indicate that *piaA* mRNA is also present in higher abundance in the nasopharynx than in the other niches at 72 h following challenge for D39. A higher level of *piaA* mRNA was also observed for bacteria harvested from the nasopharynx of WCH16-infected mice rather than the lungs, although the results from bacteraemic pneumococci were inconclusive. This was surprising as PiaA has been shown to be important in sepsis and lung disease, with a *piaA*⁻ strain exhibiting attenuated virulence in both models of infection (Brown *et al.*, 2001a). The results presented here may indicate an important role for of PiaA in the nasopharynx, where free iron levels are low. Mutagenesis studies investigating the effect of PiaA on nasopharyngeal colonisation would help to elucidate the requirement for PiaA in this niche.

4.3.2 Gene expression during lung disease

For D39 in particular, establishment of lung infection often precedes invasion of the bloodstream. *cbpA* mRNA levels for pneumococci recovered from the lungs were similar to that observed for colonising pneumococci, and higher than in bacteria from the bloodstream for both strains examined. This was not surprising, as CbpA has been previously shown to bind cytokine-activated human lung epithelial cells and endothelial cells, implicating the protein in lung disease (Rosenow *et al.*, 1997). Interestingly, this study found levels of both *psaA* and *cpsA* mRNAs to be higher in the lungs than the nasopharynx for WCH16 but not for D39. However, infection of this niche was a rare occurrence during WCH16 experiments and only low numbers of bacteria were recovered from harvested lungs, despite the presence of pneumococci in both the bloodstream and brains of infected mice. It is possible that lung disease is not a typical symptom of WCH16 infection. Therefore, the real-time RT-PCR data obtained for WCH16 must be looked at with some reservations. Indeed, it is possible that pneumococci harvested from the nasopharynx and may not have resulted in pneumonia, making bacteria harvested from this niche more characteristic of colonising bacteria than bacteria involved in lung disease. Additionally, the very low bacterial yield from the lungs may have resulted in inaccuracies in subsequent real-time RT-PCR analysis.

4.3.3 Gene expression during sepsis

Pneumococci can enter the circulatory system and cause sepsis, either following the development of lung infection or by invasion from the nasopharyngeal epithelium into underlying tissues. The pneumococcus has several ways of preventing clearance by the host's innate immune response in the bloodstream. One such mechanism is the ability of PspA to reduce complement-mediated opsonisation, either by blocking the formation, or accelerating the dissociation, of the C3 convertase (Tu *et al.*, 1999). Both a PspA⁻ mutant and pneumococci incubated with anti-PspA exhibit increased binding of C3 *in vitro*, and PspA⁻ pneumococci are rapidly cleared from the blood following intravenous challenge,

compared to mice infected with the wild-type (Ren *et al.*, 2003; Ren *et al.*, 2004; Tu *et al.*, 1999). Although the highest amounts of *pspA* mRNA in the current study were recovered from bacteria in the nasopharyngeal washes for D39 and WCH16, *pspA* mRNA was also noticeably higher in the bloodstream than in the lungs for both D39 and WCH16.1, which supports the role of PspA in reducing complement-mediated clearance in the blood. This observation is also consistent with a study by Orihuela *et al.* (2004), which demonstrated that *pspA* transcription is up-regulated in the blood of infected mice compared to *in vitro* culture. Interestingly, the level of *pspA* mRNA in the blood compared to the *in vitro* culture was noticeably lower for WCH16 than for D39. Considering the role of PspA in resisting complement-mediated opsonisation (Tu *et al.*, 1999), it is possible that the lower level of *pspA* mRNA observed for WCH16 could result in greater bacterial clearance in the blood. This may contribute to the initial predisposition of WCH16 to cause only moderate-level bacteraemia in CD-1 mice after infiltration of the vascular system, whereas D39-infected mice develop sepsis (Chapter 3).

The CPS possesses strong antiphagocytic properties and has been shown to be essential for virulence (Austrian, 1981a; Briles *et al.*, 1998). Furthermore, immunoelectron microscopy has shown that opaque phase variants, which are preferentially recovered from the blood, possess more CPS than transparent colonies, underscoring the importance of CPS in this niche (Kim *et al.*, 1999). Despite this, no significant difference in *cps2A* mRNA was observed between niches for D39-infected mice at 72 hr post-challenge in the current study. Orihuela *et al.* (2004) observed similar results when they compared *S. pneumoniae* TIGR4 harvested from *in vivo* and *in vitro* by microarray analysis, with *cps4A* mRNA levels being comparable between pneumococci harvested from mouse blood and grown *in vitro*. In fact, they found *cps4A* expression was slightly up-regulated in bacteria

isolated from pneumococci co-cultured with Detroit-562 nasopharyngeal cells compared to the tissue culture media control. This was interesting, considering the importance of higher CPS levels in the bloodstream preventing phagocytosis, and lower CPS levels in the nasopharynx possibly facilitating the exposure of pneumococcal surface adhesins (Briles et al., 1998). It is possible that the pneumococcal capsule is regulated at a posttranscriptional level, allowing rapid assembly when the pathogen is exposed to a new environment such as the bloodstream, in which the anti-phagocytic properties of the capsule are highly advantageous. These results are also consistent with previous work published from our laboratory that suggest that the level of encapsulation is modulated by autophosphorylation of CpsD (Morona et al., 2000) (Morona et al., 2003). However, a study by Ogunniyi et al. (2002) indicated that cps2A mRNA was present in higher levels in D39 recovered from the blood of IP-infected BALB/c mice than S. pneumoniae grown in THY broth. The reasons for the disparity between the results from Ogunniyi et al. (2002) and the current study are unclear, although the use of different mouse strains, infection models and RNA quantitation method (Ogunniyi *et al.* used relative quantitative RT-PCR) may be contributing factors.

4.3.4 Gene expression during early meningitis

In Section 3.4.1, WCH16 was shown to progress to the brain of some IN-infected CD-1 mice as early as 24 hr post-challenge, with low levels of pneumococci recovered from the majority of infected mice by 72 hr. This provided an opportunity to look at the relative expression of certain pneumococcal genes during early meningitis. Previously, CbpA has been implicated in transcytosis across brain microvascular endothelial cells, and has also been shown to be present in higher amounts in transparent pneumococci, the opacity phenotype associated with invasion of brain microvascular endothelial cells (King

et al., 2004; Ring et al., 1998). Likewise, cbpA mRNA was present in higher amounts in pneumococci recovered from brain than from the bloodstream in this study. The level of *cbpA* mRNA was similar to that seen for the nasopharynx, where CbpA has been shown to play an important role as an adhesin. This may indicate that CbpA also has an important role in the brain, or perhaps that an increased level of CbpA is associated with the transparent phenotype, which is consistent with the putative role of CbpA in crossing the BBB (Ring *et al.*, 1998). Ply and H_2O_2 (the production of which is catalysed by SpxB) both contribute to the symptoms of meningitis by causing neuronal damage. Specifically, they damage brain cell mitochondria, which results in the liberation of apoptosis-inducing factors. They then effect the release of apoptosis-inducing factors from the cell by causing a Ca^{2+} influx or (in the case of Ply) by catalysing the formation of pores in the lipid bilayer (Bermpohl et al., 2005; Braun et al., 2002). The roles of these proteins in promoting early meningitis were supported by the mRNA data in this chapter, with elevated levels of *plv* and *spxB* transcripts observed for pneumococci harvested from the brains of WCH16infected mice, compared to levels from the lungs and blood. Although Orihuela et al. (2004) found the transcription of *spxB* to be suppressed in the brain by microarray analysis, the fold-difference observed was relatively small. Furthermore, the meningitis model they employed involved direct intracisternal injection of TIGR4 with recovery of bacteria 4 hr post-challenge, which may be responsible for the disparity between their data and the results from this study.

4.3.5 Conclusion

The spread of the pneumococcus within a population occurs most readily by the aerosol route. Hence, there is minimal selective pressure for pneumococcal virulence factors that promote IPD alone. As the roles of various virulence factors have been elucidated, it has become increasingly clear that many of those involved in promoting disease also are important in nasopharyngeal colonisation. In this chapter, levels of virulence gene expression in pneumococci extracted from different mouse niches were compared by real-time RT-PCR. This was the first time pneumococci harvested from a single animal model had been used for direct comparison of niche-specific mRNA levels. Moreover, the work in this chapter elucidated the expression patterns of several important pneumococcal virulence factors during carriage. The examination of transcriptional behaviour in the nasopharynx has been problematic prior to this study, due to the low number of bacteria present in this niche, and this chapter presents the first in vivo gene expression data for pneumococci during nasopharyngeal colonisation. The results from the real-time RT-PCR analysis of D39 and WCH16 mRNA indicated that several of the genes examined in this chapter appear important in multiple niches. For the most part, they appeared up-regulated in niches in which they had previously been shown to contribute to growth or infection in the literature. Furthermore, examining the gene expression of WCH16 allowed the unique opportunity to assay mRNA from the nasopharynx and the brain of infected mice simultaneously. Transparent pneumococci are better able to establish and maintain nasopharyngeal colonisation, and possess a far greater ability to invade and migrate across BMEC, than their opaque counterparts (Ring et al., 1998; Weiser *et al.*, 1994). It was interesting to note that genes up-regulated in the nasopharynx of CD-1 mice (compared to the lungs and/or the blood) were also up-regulated in pneumococci harvested from the brain. This suggests that, despite several *in vitro* studies suggesting otherwise, there may be a link between opacity phenotype and the expression of S. pneumoniae virulence factors in vivo. To further examine pneumococcal behaviour during colonisation and disease, a global analysis of gene expression patterns in these various niches was performed in Chapter 5.

CHAPTER 5 – Global comparison of pneumococcal gene expression *in vivo* by microarray analysis

5.1 Introduction

While real-time RT-PCR is a sensitive technique to quantify relative amounts of specific transcripts *in vivo* (Chapter 4), a more global approach needs to be employed in order to identify novel or previously unconsidered genes that may play niche-specific roles in pathogenesis, and to gain a better understanding of pneumococcal behaviour *in vivo*. Several large-scale techniques have been used in the past to identify genes that are essential for virulence or that expressed more highly in certain niches and environmental conditions, such as differential fluorescence induction and signature-tagged mutagenesis (Hava and Camilli, 2002; Lau *et al.*, 2001; Marra *et al.*, 2002a; Polissi *et al.*, 1998). More recently, microarray analysis has been employed to examine pneumococcal gene expression both *in vitro* and *in vivo* (King *et al.*, 2004; Orihuela *et al.*, 2004b). However, microarray studies using *in vivo*-derived pneumococcal RNA have met with technical restrictions, resulting in data that are not completely comprehensive and must be interpreted with a degree of caution.

In this chapter, pneumococcal gene expression between different niches was assayed by RNA microarray analysis. Bacteria were recovered from all sites of interest using the IN model of infection, which simulated the natural route of pneumococcal acquisition and disease progression. These are the first microarray data generated using a single pneumococcal strain in a continuous infection model.

5.2 Comparison of gene expression in pneumococci involved in colonisation, lung infection and IPD by microarray analysis

5.2.1 Microarray analysis of D39 gene expression between niches

RNA samples harvested in Section 4.2.2 from D39.1 and D39.2 were amplified twice using SenseAmp (Genisphere), as described in Section 2.5.3. 3 μg of amplified RNA (as determined by A_{260/280} readings) from two distinct niches were labelled with either Alexa 546 or Alexa 647 dyes (Genisphere). Microarray analyses reported in this chapter were performed using the Genisphere Array 900 MPX Kit, a commercial kit that is optimised for low quantities of RNA. This made it possible to use lower amounts of *in vivo*-derived RNA without a significant labelling bias. Labelled RNA samples were hybridised to microarray slides spotted with PCR products for each of the 2240 defined open reading frames in the *S. pneumoniae* TIGR4 genome (Bacterial Microarray Group). Additional spots were present on each array, consisting of 117 PCR products for open reading frames (ORFs) that were present in the R6 genome, but absent from the TIGR4 genome, or that exhibited low nucleotide sequence homology. Each ORF was represented twice per array slide. Arrays were read using an Axon 4000B Dual-wavelength Scanner

(Axon Instruments) and then normalised by adjusting the PMT 532 nm and 635 nm intensities until 16S rRNA ratios were approximately 1: 1. This method is described in Section 2.6.3.

5.2.2 Analysis of D39 microarray results

Two cDNA microarrays for each niche comparison were performed using biological replicates labelled with either Alexa 546 or Alexa 647 dyes. The raw data recovered from the microarrays were subjected to the following analytical criteria: Arrays were firstly analysed by GenePix Pro 4.0 using filtration algorithms and signal-to-noise ratios (based on the difference in signal intensity between the spots and the background). Genes for which the corresponding spots incurred errors, or were flagged "bad" or "not found" more than 25% of the time using GenePix Pro 4.0, were removed from further analysis. A cut-off level of log₂ ratio 2-2.5 (4-fold to 6-fold difference in mRNA quantity between niches) was imposed for each array, and genes that exhibited an Alexa 546:Alexa 647 ratio that failed to reach this level were not analysed further. Two-tailed Student's ttest analysis (Microsoft Excel, 2003) was used to calculate the statistical significance of the differences in gene expression for each gene within arrays and between replicate biological experiments. Genes with high levels of significance (P < 0.01) and a minimum difference of 4-fold were considered up- or down-regulated. The results from this analysis are presented as fold-difference between niches in Table 5.1, classed according to their proposed functions, as listed on the TIGR Comprehensive Microbial Resource website. Using these criteria, 127 genes were identified as being differentially expressed between niches in a CD-1 mouse model. This is a conservative number and reflects both the stringency of analysis and the significant natural variation between biological replicates during in vivo experimentation.

		0									
Gene (Gene name, if available)	TIGR annotation				Fold incre	ase or deci	ease				
		Nasal wash vs	Lungs	Р	Nasal wash vs	Blood	Ρ	Lungs	vs Blo	poq	Ρ
Virulence factors											
bacteriocin (blpU)	Sp0041		2.5*	0.0123		5.3*	4.61E-04		5.5		9.00E-03
choline binding protein G (<i>cbpG</i>)	Sp0390		2.5*	4.61E-05		10.3	1.47E-04		3.2	+ *	0.0563
protective antigen A (<i>prtA</i>)	Sp0641		1.3^{*}	0.0456		3.6^{*}	1.56E-03		4.4	*	7.50E-05
adherence and virulence protein A	Sp0966		NA			6.6	1.08E-04		4.5	+ *	3.21E-03
neuraminidase, putative	Sp1326	24.2		8.84E-06	23.8		0.010	5.6^{+}			0.0312
choline binding protein A (<i>cbpA</i>)	Sp2190	14.5		1.05E-06	22.0		2.41E-05	7.1*			3.70E-05
Ion-acquisition and hinding											
Na/Pi cotransporter II-related protein	Sp0496		3.3*	2.08E-03		4.9^{*+}	8.00E-05		8.2		1.21E-04
(phoU)	1										
v-type sodium ATP synthase, subunit D	Sp1315	9.5		4.41E-07	8.1^{*+}		0.0115	4.2^{+}			6.81E-04
v-type sodium ATP synthase, subunit B	Sp1316	42.5		2.80E-08	16.0		1.70E-04	21.8+			0.0134
(ntpB)											
v-type sodium ATP synthase, subunit A	Sp1317	66.3		8.40E-08	6.4		1.58E-03	11.2^{+}			7.86E-03
v-type sodium ATP synthase, subunit E	Sp1320	14.8		3.16E-04	11.8 +		1.57E-03	2.4^{*+}			0.150
v-type sodium ATP synthase, subunit K	Sp1321	28.4		6.00E-04	10.7		4.38E-05	2.8^{+}			0.192
(ntpK)											
v-type sodium ATP synthase, subunit I	Sp1322	62.6		2.90E-05	53.1		3.34E-04	20.7+			7.74E-07
(ntpl)											
sodium:solute symporter family protein	Sp1328	34.3		3.20E-05	12.9		1.09E-05	20.5^{+}			0.0131
non-heme iron-containing ferritin	Sp1572		2.6^{*}	1.02E-03		1.3^{*}	0.581		5.6		4.40E-05
cation efflux system protein	Sp1857		1.6^{*}	1.58E-03		12.0	2.32E-07		5.5	*	1.24E-05
iron-compound ABC transporter, ATP-	Sp1871		3.6*	1.79E-03		12.4	1.66E-05		6.3	*	8.19E-03
binding protein											
iron-compound ABC transporter, iron-	Sp1872		1.7^{*}	1.06E-03		13.0	5.90E-05		3.6	*	4.64E-03
compound-binding protein											

Table 5.1: Differential in vivo expression of S. pneumoniae D39 genes

Table 5.1 continued										
Gene (Gene name, if available)	TIGR annotation				Fold incre	ease or deci	rease			
		Nasal wash vs	Lungs	P	Nasal wash vs	Blood	Ρ	Lungs vs	Blood	Р
hypoxanthine-guanine	Sp0012		4.0^{*}	1.51E-03		8.8	1.08E-07		12.7	9.72E-05
phosphoribosyltransferase (hpt)										
transcriptional regulator (mutR)	Sp0141		4.7	2.47E-06		4.3^{*+}	6.07E-08		4.9+	0.0127
transcriptional regulator, MerR family	Sp0501		2.2*	1.51E-03		4.7	6.35E-06		6.6*	9.94E-03
ATP-binding protein (amiE)	Sp1888		3.2*	2.17E-08		9.3	5.73E-05		3.1^{*+}	4.46E-05
permease protein (amiD)	Sp1889		2.3*	6.16E-07		8.5	6.15E-07		4.0*	1.24E-05
Competence										
competence factor transport protein	Sp0043		2.9*	5.12E-03		12.9	6.68E-07		4.5*	1.51E-04
(comB)										
response regulator (<i>comE</i>)	Sp2235		2.6*	7.67E-08		10.2	2.59E-05		4.8*	4.13E-05
Metabolism/cellular growth										
putative acyl carrier protein	Sp0038		2.1^{*}	1.56E-03		9.2	2.39E-06		2.8*	2.84E-03
phosphoglycerate mutase family protein	Sp0240		3.1^{*}	4.08E-03		5.1^{*}	9.04E-05		6.6	4.00E-03
phosphatidate cytidylyltransferase	Sp0262		3.2*	3.21E-03		9.5	1.73E-04		5.9	2.01E-06
lactate oxidase, truncation (lctO)	Sp0712		4.8	1.50E-06		4.3^{*+}	6.07E-08		2.6^{*+}	0.0390
cell division ABC transporter, permease	Sp0757		2.6^{*}	1.22E-06		4.1^{*}	9.12E-06		7.1	7.16E-04
protein (<i>ftsX</i>)										
phosphoenolpyruvate carboxylase (ppc)	Sp1068		2.1^{*}	4.11E-03		11.4	4.15E-06		4.7*	5.74E-04
macrolide-efflux protein	Sp1110		3.4*	3.91E-05		7.3	1.17E-05		4.8*	2.57E-06
6-phospho-B-galactosidase (lacG)	Sp1184	15.2		9.42E-06	9.6		5.90E-04	7.8+		6.05E-04
putative platelet activating factor	Sp1450	4.3^{*+}		0.384		4.7^{*+}	0.0275		6.0	4.05E-06
transketolase ($recP$)	Sp1615	180.7		1.50E-04	37.6		2.03E-03	6.1 +		4.17E-06
L-ribulose 5-phosphate 4-epimerase	Sp2033	1.3^{*+}		0.0683	156.0		1.67E-04		NA	
(araD)								÷		
hexulose-6-phosphate 1somerase	Sp2034		NA		63.9		1.21E-05	1.9^{*+}		0.109

Table 5.1 continued										
Gene (Gene name, if available)	TIGR annotation				Fold increa	se or decr	ease			
		Nasal wash vs	Lungs	P	Nasal wash vs	Blood	Р	Lungs vs	Blood	Ρ
hexulose-6-phosphate synthase	Sp2035	20.4+		1.51E-03	114.9		1.35E-03		NA	
glycerol kinase (<i>gplK</i>)	Sp2186	28.6+		2.66E-05	17.7		3.48E-03	1.9^{*+}		0.142
replicative DNA helicase (<i>dnaC</i>)	Sp2203		1.5^{*}	1.63E-03		7.1	2.26E-05		4.3*	0.0106
spspoJ protein	Sp2240		NA			2.0^{*+}	0.454		7.0	2.32E-05
Sugar transport										
putative PTS system IIA component	Sp0645	6.1		4.24E-07		2.5^{*+}	0.0890		1.8^{++}	0.0538
PTS system, lactose-specific IIBC	Sp1185	6.8		1.59E-06	4.5*+		6.82E-04	2.1^{++}		6.23E-03
components										
PTS system, nitrogen regulatory	Sp1620	11.6		1.19E-05	4.6^{*+}		6.82E-04	1.9^{*+}		0.347
component IIA, putative										
PTS system, IIA component	Sp2036	9.0		1.63E-06	90.5		6.85E-04	2.1^{*+}		0.395
PTS system, IIB component	Sp2037	2.3^{+}		4.34E-04	41.2		3.66E-05		NA	
PTS system, membrane component,	Sp2038	1.8^{+}		0.178	28.9		6.44E-05	2.5^{+} +		0.124
putative										
Transcription and translation										
tRNA(Ile)-lysidine synthetase (tilS)	Sp0011		2.8*	4.13E-06		13.5	3.00E-06		8.6	6.19E-05
ribosomal protein S8 (rpsH)	Sp0224		1.7^{*}	4.38E-04		9.0	1.36E-05		3.8*	9.34E-06
prolyl-tRNA synthetase (proS)	Sp0264	2.1^{*+}		0.437		8.6	1.95E-05		3.2*	8.15E-05
N utilization substance protein A (nusA)	Sp0553		1.6^{*}	2.16E-04		6.5	6.38E-04		6.3	9.17E-07
ATP-dependent RNA helicase,	Sp0761		3.7*	1.11E-03		8.9	1.29E-05		5.2*	2.66E-04
DEAD/DEAH box family										
putative transcriptional regulator	Sp1131	8.4		6.19E-08	4.9*+		5.47E-03		1.6^{+}	0.695
transcription antiterminator (<i>lacT</i>)	Sp1187	11.2		1.92E-04	4.2*		1.03E-03	4.6^{+}		1.08E-03
putative ATP-dependent RNA helicase	Sp1586		4.8	2.40E-07		2.1^{*}	0.119		4.7*	3.58E-05
transcriptional regulator	Sp1809		2.6*	2.40E-03		7.3	3.60E-04		7.2	8.59E-06

Table 5.1 continued										
Gene (Gene name, if available)	TIGR annotation				Fold incr	ease or deci	rease			
		Nasal wash vs	Lungs	Ρ	Nasal wash v	s Blood	Ρ	Lungs vs	Blood	Ρ
transcriptional regulator, MerR family	Sp1856		2.1^{*}	2.22E-04		8.0	3.18E-06		5.0	6.26E-07
transcription antitermination protein	Sp2007		2.9*	9.96E-04		7.2	2.90E-04		6.0	4.88E-06
transcriptional regulator, BglG family	Sp2032	12.1		3.42E-05	34.4		7.41E-06	6.1		2.21E-03
transcriptional regulator, MarR family	Sp2062		2.2*	6.45E-04		3.1^{+}	0.175		5.7	2.21E-05
transcriptional regulator, authentic	Sp2123	12.4		4.31E-03	3.1^{+}		0.209	1.7^{+} +		0.358
frameshift										
ribosomal protein L9 (rpll)	Sp2204		2.3*	5.52E-03		7.1	2.26E-05		3.7*	8.65E-05
Protein fate										
preprotein translocase, SecY subunit	Sp0230		3.4*	9.45E-04		8.6	2.60E-06		3.6*	0.0302
putative preprotein translocase, SecE subunit	Sp2008		2.0*	1.39E-04		4.9*	9.92E-05		8.4	6.01E-05
SOS repair MATE efflux family protein (<i>dinF</i>),	Sp1939		4.1	1.57E-05		5.9*	1.75E-03		7.9	9.85E-04
putative SOS repair										
Cell wall and membrane										
glycosyl transferase	Sp0102		2.6^{*}	1.48E-03		7.4	5.42E-07		4.0*	1.06E-04
beta-lactam resistance factor (fibA)	Sp0615		1.5*	0.250		7.9	8.19E-05		3.7*	4.56E-05
putative membrane protein	Sp0858		2.3*	6.64E-03		8.8	8.01E-07		9.9	7.6E-05
endo-B-N-acetylglucosaminldase (lytB)	Sp0965		2.9*	0.104		9.4	2.90E-05		4.8^{*+}	8.49E-04
peptidoglycan N-acetylglucosamine	Sp1479		NA			7.4	5.35E-05		3.0^{*+}	3.30E-04
deacetylase A $(pgdA)$										
glycosyl transferase, authentic	Sp1769	62.9		1.34E-05	20.7^{++}		1.08E-03	4.5^{*+}		1.39E-05
frameshift										
cell wall surface anchor family protein	Sp1772	13.4		2.22E-05	9.4		8.14E-05	4.2^{*+}		0.257

Gene (Gene name, if available)	TIGR annotation				Fold incre	ase or decr	ease			
		Nasal wash w	s Lungs	P	Nasal wash vs	Blood	Ρ	Lungs vs	Blood	Ρ
glycosyl transferase, putative	Sp1838		3.9	1.47E-07		1.7^{*+}	0.148		11.4^{*+}	8.54E-03
E										
Transposon										
IS630-Sp1, transposase Orf2	Sp0016		3.7*	2.90E-05		9.1	2.15E-05		4.8*	2.58E-04
IS1381, transposase ($orfB$)	Sp1086		1.4^{*}	5.52E-08		6.4	1.10E-11		6.0^{*}	5.84E-04
IS1167, transposase, degenerate	Sp1101		4.4	9.09E-06		9.4	9.88E-05		7.5*	2.30E-03
IS66 family element, Orf3, degenerate	Sp1311		1.8^{*}	9.06E-03		8.9	4.09E-07		3.1*	7.49E-03
IS1380-Sp1, transposase	Sp1337	27.6		4.81E-08	57.2		2.89E-04	9.0^{++}		0.0876
IS1380-Sp1, transposase	Sp1352	87.0		8.74E-09	36.4		5.41E-03	27.8		0.0106
IS1380-Sp1, transposase	Sp1418	68.7		2.10E-04	12.2		7.33E-06	9.0*+		0.171
IS1380-Spn1, transposase	Sp1439	17.7		2.46E-06	23.7+		6.52E-05	11.0		0.0140
transposase, IS630-Spn1 related, Orf2	Sp1496	4.7		4.78E-09	1.9^{*+}		0.0301	1.3^{+}		0.759
IS1380-Sp1, transposase	Sp1503	17.6		1.34E-06	26.7		0.0105	8.5*+		2.45E-04
transposase, IS200 family	Sp1622	19.8		1.73E-05	15.0^{++}		6.41E-03	5.6^{+}		1.20E-03
transposase, IS1380-Spn1 related,	Sp2089	12.4		1.77E-04	1.5^{*+}		0.655		NA	
truncation										
IS66 family element, Orf3, degenerate	Sp2211		1.4*	4.75E-03		6.9	4.22E-06		4.4*	1.34E-05
Unknown genes/ unknown function										
GTP-binding protein	Sp0004		NA			11.8	5.31E-06		9.2	1.77E-04
hypothetical protein	Sp0008		NA			4.5^{*+}	0.0284		12.8	5.89E-04
conserved hypothetical protein	Sp0095		1.9*	4.43E-03	1.1^{*+}		0.724		5.4	3.59E-07
hypothetical protein	Sp0144		3.6*	3.87E-03		11.2	2.44E-03		18.9	2.87E-05
acetyltransferase, GNAT family	Sp0204		NA			23.5	4.11E-05		5.5^{+}	0.0255
conserved hypothetical protein	Sp0239		5.0	2.08E-04		2.9^{*+}	0.0122		1.3^{+}	0.228
hypothetical protein	Sp0455		1.5^{*}	9.78E-08		10.4	7.67E-04		4.5^{*+}	2.59E-04
conserved domain protein	Sp0500		2.6^{*}	2.74E-03		9.6	1.99E-06		6.0^{*}	5.21E-05

Table 5.1 continued

	TICD currentiation				Pold incurre		00000			
Gene (Gene name, n avanaure)	I I I I I I I I I I I I I I I I I I I				roiu micrea	ise of nect	ease			
		Nasal wash vs	Lungs	Ρ	Nasal wash vs	Blood	Ρ	Lungs vs	Blood	Ρ
acetyltransferase, GNAT family	Sp0590		2.8*	0.0383		10.1	9.67E-05		6.7*	2.31E-06
chorismate binding enzyme	Sp0655		2.9*	4.58E-05		7.1	5.61E-05		6.3	5.03E-08
hypothetical protein	Sp0777		2.8*	4.31E-04		8.2	1.40E-06		3.6*	7.84E-03
hypothetical protein	Sp0781		2.5*	3.79E-03		4.6^{*+}	2.00E-04		7.4	5.19E-05
hypothetical protein	Sp0830	11.6		3.02E-07		3.1*	7.66E-03		3.5*	1.60E-05
pilin gene inverting-related protein	Sp0917		3.0*	0.0243		15.3	2.86E-05		7.2*	2.18E-03
acetyltransferase, GNAT family	Sp1023		1.4^{*}	1.45E-03		7.7	1.16E-05		3.5*	1.88E-04
hypothetical protein	Sp1059	19.2		1.78E-05	4.1^{*+}		0.144		1.1^{*+}	0.786
acetyltransferase, GNAT family	Sp1082		1.5*	0.350		13.0	3.41E-05		5.1*	6.98E-06
aminotransferase, class-V	Sp1094		3.3*	3.65E-03		9.3	9.28E-05		3.5*	1.55E-03
hypothetical protein	Sp1139	15.2		9.42E-06	10.2		6.15E-03	2.9^{*+}		0.212
conserved hypothetical protein	Sp1143	10.5		1.09E-05	3.4^{++}		0.0116	1.6^{+}		0.463
hypothetical protein	Sp1189	32.2		1.92E-04	6.3^{+}		8.79E-03	1.8^{+}		0.095
SAP domain protein	Sp1292	15.9		9.23E-05	10.1		9.98E-04	2.6^{+}		2.29E-03
conserved domain protein	Sp1363	3.6*+		0.256		9.9	1.71E-05		5.3*	1.27E-04
hypothetical protein	Sp1558		2.5*	4.35E-04		5.9	1.42E-10		5.0*	2.23E-05
conserved domain protein	Sp1612	35.8		1.20E-03	11.5		1.03E-06	6.2^{+}		5.5E-04
hypothetical protein	Sp1643	37.0		1.05E-03	1.9^{+}		0.0889		NA	
ylmH protein	Sp1662		1.6^{*}	0.0107		6.6	4.15E-08		3.6*	2.94E-03
ROK family protein	Sp1675	30.5		9.93E-07	4.8^{+}		0.173	1.8^{+}		0.366
ABC transporter, permease protein	Sp1688	19.5		5.08E-07	5.6		0.0168	9.8		1.19E-03
conserved hypothetical protein	Sp1747		3.4*	3.66E-07		3.7*	1.98E-05		<i>T.T</i>	7.26E-05
hypothetical protein	Sp1862		3.0^{*+}	0.485		6.1^{*}	3.06E-04		8.7	2.76E-05
acetyltransferase, GNAT family	Sp1943		2.8*	2.23E-04		8.7	1.50E-08		5.7	1.05E-03
CBS domain protein	Sp1963	2.8*+		0.456		6.2*	6.51E-05		7.6	3.17E-04
conserved hypothetical protein	Sp2031	19.2^{++}		3.57E-03	131		9.73E-05		2.3^{+}	0.126
conserved hypothetical protein	Sp2059	1.7^{*}		0.488		6.8	7.05E-09		5.2	1.17E-05

Table 5.1 continued

Table 5.1 continued							
Gene (Gene name, if available)	TIGR annotation		Fold increas	e or decrease			
		Nasal wash vs Lungs P	Nasal wash vs	Blood P	Lungs vs	Blood	P
glycosyl hydrolase-related protein	Sp2141	8.9 1.59E-	-03 11.9	1.47E-0	4 4.6*+		0.0281
ROK family protein	Sp2142	7.0 1.91E-	-04 19.8	5.58E-0	5.2*+		3.73E-03
conserved hypothetical protein	Sp2160	16.2 1.33E-	-06 4.6*	0.0876	2.2^{+}		0.445
hypothetical protein	Sp2185	9.3 2.11E-	-03 18.3	1.93E-0	5 12.0		1.92E-05
conserved hypothetical protein	Sp2202	1.7* 6.93E-	-04	6.5 5.47E-0	8	5.6*	2.50E-03

RNA was extracted from in vivo-derived pneumococci, enriched for prokaryotic RNA and amplified, as described in the text. Microarray experiments were then performed comparing pneumococci harvested from different niches using Genisphere MPX900, and analysed using Genepix Pro 4.0. Data were analysed as described in Section 5.2.2 and Student's t test analysis was used to calculate the statistical significance of the differences in gene expression for each gene within arrays and between replicate biological experiments. Figures shown in grey did not pass analysis criteria, due to data for that gene being below the cut-off threshold value for one, or both, experiments (*) or due to error or flag tags, as determined by Genepix Pro 4.0 (+). NA indicates that data could not be obtained, as the majority of spots were tagged as errors (hence, did not return a value). A dye reversal experiment was performed to determine if there was a significant dye labelling bias using the Genisphere Array 900 MPX Kit. This consisted of performing microarray labelling and hybridisation in duplicate, labelling an RNA sample with Alexa 546 dye in one array experiment, and Alexa 647 in the other. The dye reversal was performed using nasopharyngeal and blood samples from D39.1. Analysis of one microarray slide identified a total of 245 genes that exhibited a significant fold-difference in gene expression between the niches. 205 (84%) of the identified genes were similarly up- or down- regulated on the dye reversal array. An opposing fold increase or decrease was observed for 12 (5%) genes, while 28 (11%) could not be compared due to error or flag tags by Genepix Pro 4.0. These results indicated that labelling bias using the labelling method provided by the Genisphere Array 900 MPX Kit with Alexa 546 and Alexa 647 was minimal.

Analysis of the microarray data identified *in vivo* differential expression of a large number of genes involved in fatty acid, energy or sugar metabolism. Genes involved in sugar metabolism were largely up-regulated in the nasopharynx compared to other niches, whereas genes with a role in fatty acid metabolism were expressed more highly in the blood. *lacG* and genes belonging to sucrose phosphotransferase systems (PTS) appeared up-regulated in the nasopharynx compared to either the blood or lungs. Likewise, several transposases exhibited higher expression in the nasopharynx than other niches. Although the gene encoding the major pneumococcal neuraminidase *nanA* was not identified in these experiments due to the corresponding array spots falling below the cutt-off criteria, a putative neuraminidase encoded by Sp1326 was up-regulated in lungs, and even more so in the nasopharynx, compared to the blood. Conversely, levels of competence factors *comB*

and *comE* mRNA were found to be higher in the blood than the nasopharynx, as were levels of *dinF* transcript which encodes a protein thought to be involved in SOS DNA repair. *amiD* and *amiE* transcripts were also present in higher amounts in pneumococci recovered from the blood than the nasopharynx. These two proteins encode transmembrane components of the Ami oligopeptide ABC transporter, which has previously been hypothesised to be involved in sensing environmental changes, indirectly resulting in the regulation of certain pneumococcal genes (Claverys *et al.*, 2000).

The microarray results for the virulence factors examined by real-time RT-PCR analysis in Chapter 4 are shown in Table 5.2. For many of these genes, the corresponding spot intensities on the microarrays performed in this chapter were unable to be accurately determined, either falling below the cut-off points, incurring an error when analysed by GenePix Pro 4.0, or possessing a *P*-value > 0.01. However, spots corresponding to *cbpA* did pass the analysis criteria, indicating the amount of *cbpA* transcript present in nasopharyngeal pneumococci was 22-fold higher than for bacteria recovered from the blood and 14.5- fold higher than for bacteria recovered from the lungs, which was consistent with the higher level of *cbpA* mRNA observed in the nasopharynx by real-time RT-PCR. Interestingly, although a difference in *nanA* quantity was not identified by microarray analysis, a putative neuraminidase gene (Sp1326) was up-regulated 24-fold higher in bacteria harvested from the nasopharynx compared to other niches, suggesting a potential role for this neuraminidase in carriage as well.

5.2.3 Preliminary results from WCH16 microarray analysis

Microarray analysis was also used to examine the differential expression of WCH16 genes in different niches in an IN mouse infection model. Nucleic acid samples

Gene	TIGR annotation				Fold in	crease oi	· decrease			
		Nasal	Lungs	Р	Nasal	Blood	Р	Lungs	Blood	Р
		wash			wash					
cbpA	Sp2190	14.5		1.05E-06	22.0		2.41E-05	7.1*		3.70E-05
pspA	Spr0121		NA		1.1*		0.922	2.7*		0.222
ply	Sp1923		2.2*	2.62E-04		1.2*	0.871		3.6*	4.56E-05
psaA	Sp1650	1.5*		1.79E-06		4.6*	2.45E-03		2.5*	4.18E-03
cps4A	Sp0346		2.0*	0.0289	1.1^{*+}		0.948		NA	
piaA	Sp1032	1.9*+		0.0292		2.4*+	2.80E-03		2.3*+	1.31E-03
nanA	Sp1963	1.9*		2.07E-04	1.9*		0.353		4.5*+	3.19E-04
spxB	Sp0730		NA		1.2*		0.847		NA	

Table 5.2: Differential expression of *S. pneumoniae* D39 virulence genes, as determined by microarray analysis

RNA was extracted from *in vivo*-derived pneumococci, enriched for prokaryotic RNA and amplified using Genisphere SenseAmp. Microarrays were then performed comparing pneumococci harvested from different niches using Genisphere MPX900, and analysed as described in the text for the combined data from two independent biological experiments. Figures shown in grey did not pass analysis criteria, due to data for that gene being below the cut-off threshold value for one, or both, experiments (*) or due to more than 25% of spots being tagged "error" or "flag", as determined by Genepix Pro 4.0 (+). NA indicates that data could not be obtained, as the majority of spots were tagged as errors (hence, did not return a value).

recovered from in vivo-derived pneumococci in experiments WCH16.1 and WCH16.2 (Chapter 4) were amplified, labelled and used in microarray hybridisation experiments as described for D39 in Sections 5.2.1 and 5.2.2. Only the nasopharyngeal and blood samples could be compared by microarray due to poor RNA integrity for samples recovered from the lungs, and time constraints prohibited the recovery of fresh RNA from WCH16infected mice to conduct further array analyses. Genes that exhibited different levels of expression between bacteria recovered from the nasopharynx and blood are listed in Table 5.3. In contrast to D39, there were only a few genes that were identified in both WCH16.1 and WCH16.2 arrays as being differentially expressed between niches. The different magnitude of disease observed in the replicate animal experiments may have contributed to this disparity (Figure 4.5). Furthermore, the use of a PCR-product array based on the genomic sequence of TIGR4 and R6 (a derivative of D39) to analyse the unrelated serotype 6A strain WCH16, may have resulted in reduced affinity for WCH16 genes with Nevertheless, many of the genes that exhibited niche-dependent variant sequences. expression for WCH16 were also identified in the D39 microarray experiments (Section These are marked with an asterisk in Table 5.3. Several genes involved in 5.2.2). competence were markedly up-regulated in the blood compared to the nasopharynx, including *comB* and *comE* which were found to be regulated similarly in this niche as for D39. *comX1* and *cgiA* were also identified in these WCH16 experiments as exhibiting increased expression in the blood, but were not differentially expressed in D39. However, although *comX1* showed similar expression in the nasopharynx and blood for D39.2 and hence didn't pass the analysis criteria, it was markedly up-regulated in the blood for D39.1 demonstrating a 9.9-fold increase in this niche.

Once again, the specific virulence genes examined in Chapter 4 did not pass the

Table 5.3: Differential expression of S. pneumoniae WCH16 genes as determined by microarray analysis

RNA was extracted from *in vivo*-derived WCH16 pneumococci, enriched for prokaryotic RNA and amplified as described in the text. Microarray analyses were performed comparing pneumococci harvested from the nasopharynx and blood niches using Genisphere MPX900, and analysed using Genepix Pro 4.0. Genes for which the corresponding spots incurred errors or were flagged more than 25% of the time using GenePix Pro 4.0 were removed from further analysis. A cut-off level of log₂ ratio 2-2.5 (4- fold to 6- fold difference in mRNA quantity between niches) was imposed for each array and Student's *t* test analysis (Microsoft Excel) was used to calculate the statistical significance of the differences in gene expression for each gene within arrays and between replicate biological experiments. An asterisk indicates that these genes were similarly identified in D39 arrays.

Gene (Gene name, if available)	TIGR annotation	Fold	increase or d	lecrease
		Nasal wash	Blood	Р
Virulence factors				
adhesion lipoprotein (lmb)	Sp1002	6.0		7.37E-06
Ion-acquisition and binding				
iron-compound ABC transporter, ATP- binding protein*	Sp1871		11.8	4.44E-03
iron-compound ABC transporter, iron- compound-binding protein ^D	Sp1872		17.0	8.58E-04
Competence				
transcriptional regulator (comX1)	Sp0014		7.3	3.60E-05
competence factor transport protein (comB)*	Sp0043		11.1	1.23E-05
competence protein (cglA)	Sp2053		9.7	6.17E-04
response regulator (comE)*	Sp2235		11.0	8.03E-03
Metabolism/cellular growth				
alcohol dehydrogenase, iron-containing	Sp2026		13.3	2.42E-05
hexulose-6-phosphate isomerase, putative*	Sp2034	12.4		4.47E-04
hexulose-6-phosphate synthase, putative*	Sp2035	13.1		5.32E-04
Sugar transport				
L-ribulose 5-phosphate 4-epimerase, putative (<i>araD</i>)*	Sp2036	7.2		9.56E-04
PTS system, membrane component, putative*	Sp2038	6.5		1.59E-05
Cell wall and membrane				
glycosyl hydrolase, family 1	Sp2021		5.8	1.54E-04
Transposon				
IS3-Spn1, transposase, authentic point mutation	Sp0850		6.1	1.01E-04
Unknown genes/ unknown function				
hypothetical protein*	Sp0144	7.4		7.40E-04
conserved hypothetical protein	Sp0146	5.5		1.39E-04
conserved hypothetical protein	Sp0627		6.3	5.45E-03
oxidoreductase, Gfo/Idh/MocA family	Sp1686		6.3	1.29E-03
conserved hypothetical protein*	Sp2031	11.6		1.09E-04

analysis criteria that were set (Table 5.4). The large *P*-values obtained are indicative of the variation between biological experiments also observed by real-time RT-PCR in Chapter 4. In contrast to the D39 array data described earlier in this chapter, and also the differential expression for *cbpA* between the nasopharynx and blood observed by real-time RT-PCR (Chapter 4), WCH16 did not appear to up-regulate the expression of *cbpA* in the nasopharynx by microarray analysis in this study. Serogroup 6 pneumococci encode *cbpA* belonging to a different clade than D39 *cbpA* (Iannelli *et al.*, 2002). Therefore, variations in nucleotide sequence between WCH16 *cbpA* and the TIGR4 *cbpA*, with which the microarray slides were spotted, may account for the discrepant results. Additionally, *cpsA* (which is highly conserved between many pneumococcal serotypes (Morona *et al.*, 1999)) demonstrated a 3.8-fold increase in expression in the bloodstream compared to the nasopharynx of infected mice. Further biological experiments for WCH16 would need to be conducted in order to clarify the preliminary results reported in this section.

5.3 Discussion

The transcriptional changes that allow the pneumococcus to adapt to various host environments are poorly understood. This chapter describes the use of microarray analysis to directly compare gene expression at different sites in CD-1 mice following IN infection (as established in Chapter 3). This has allowed comparison of gene expression profiles possessed by pneumococci asymptomatically colonising the nasopharynx and bacteria that had progressed to cause lung disease and IPD. This is the first report of such *in vivo* studies using a single pneumococcal strain and a continuous infection model.

Gene	TIGR	Fol	d increase or	decrease
	annotation			
		Nasal wash	Blood	Р
cbpA	Sp2190		2.2*	0.292
pspA	Spr0121		1.1*	0.759
ply	Sp1923	2.0*		9.27E-03
psaA	Sp1650		3.5*	0.313E-03
cps4A	Sp0346		3.8*+	0.148
piaA	Sp1032		2.0*+	0.0829
nanA	Sp1963	1.5*		0.429
spxB	Sp0730	1.1^{*+}		0.938

Table 5.4: Differential expression of *S. pneumoniae* WCH16 virulence genes as determined by microarray analysis

RNA was extracted from *in vivo*-derived WCH16 pneumococci, enriched for prokaryotic RNA and amplified using Genisphere SenseAmp. Microarray were then performed comparing pneumococci harvested from different niches using Genisphere MPX900, and analysed as described in the text. Figures shown in grey did not pass analysis criteria, due to data for that gene being below the cut-off threshold value for one, or both, experiments (*) or due to more than 25% of spots being tagged "error" or "flag", as determined by Genepix Pro 4.0 (+). NA indicates that data could not be obtained, as the majority of spots were tagged as errors (hence, did not return a value).

5.3.1 Global *in vivo* gene expression analysis of *S. pneumoniae* using a single mouse model and challenge strain

Due to the difficulty in recovering sufficient bacteria from various in vivo niches, studies examining global pneumococcal gene expression have been largely performed in vitro. However, in vitro studies cannot entirely simulate an in vivo environment. Perhaps because of this, DFI failed to detect promoter inductions for most of the intensively studied pneumococcal virulence genes under different in vitro conditions (Marra et al., 2002a). Furthermore, only minor changes in gene expression profiles between opaque and transparent phenotypes were reported by King et al. (2004), despite multiple groups observing a strong association between opacity phenotype and *in vivo* niche (Cundell *et al.*, 1995b; Ring et al., 1998; Weiser et al., 1994; Weiser and Kapoor, 1999). Although nanA mRNA was present in higher amounts in transparent pneumococci, albeit at the modest level of 2.1-fold for 6B strains, other virulence factors such as CPS and CbpA were expressed similarly in both phenotype variants. The small number of genes that exhibited opacity phenotype-specific gene expression, and the lack of promoter induction observed by DFI for many key virulence factors, indicates that there may not be a 'switch' that alters the expression of many genes simultaneously and that other in vivo factors, such as unspecified environmental conditions and host immune factors, may be crucial in inducing the transcriptional changes that allow pneumococci to efficiently adapt to different niches during pathogenesis.

The D39 microarray data in this chapter identified differential expression of genes involved in sugar uptake and metabolism, fatty acid and energy metabolism, cellular growth, and ion acquisition. In general, gene expression paralleled their putative roles in pneumococcal virulence and pathogenesis. Genes that contribute to sugar metabolism were largely up-regulated in the nasopharynx compared to other niches. King et al. (2004) also identified several sugar metabolism genes that were expressed more highly in transparent pneumococci (the opacity phenotype associated with pneumococci in the nasopharynx) than opaque. Enzymes involved in sugar metabolism may promote carriage by cleaving both bacterial and host glycoproteins, thus reducing clearance or enhancing adherence in the nasopharynx. Several pneumococcal surface proteins were up-regulated in the nasopharynx for D39, such as a cell wall surface anchor family protein (Sp1772) and the well-characterised CBP, CbpA (Sp2190), which is a putative adhesin in this niche. CbpA was also identified by STM as being important for lung infection (Hava and Camilli, 2002) and was observed by Orihuela et al. (2004) to be up-regulated following intimate contact with nasopharyngeal epithelial cells, supporting the results of this work. Two other CBPs were identified by microarray analysis in this chapter. *lytB* encodes a CBP that has been shown to play a role in cell division (Garcia *et al.*, 2000), and displayed almost a 10fold up-regulation in the blood compared to the nasopharynx, despite previous studies indicating that a *lvtB* mutant was attenuated in carriage but not sepsis (Gosink *et al.*, 2000). cbpG (a serine protease) was also expressed more highly in the bloodstream than the nasopharynx, which is consistent with microarray results from Orihuela et al. (2004) and previous mutagenesis studies by Gosink et al. (2000) implicating CbpG in sepsis.

Although microarray studies performed by Orihuela *et al.* (2004) found no regulation of competence genes in the blood or pneumococci co-cultured with nasopharyngeal epithelial cells compared to baseline controls (bacteria grown in C+Y media or tissue culture media, respectively), data from the current study indicate that both *comB* and *comE* transcripts are up-regulated for D39 and WCH16 in the blood relative to the nasopharynx *in vivo*. This supports previous reports indicating that competence-related

genes undergo increased transcription during exponential growth such as during the development of sepsis (Claverys and Havarstein, 2002). Preliminary microarray data from the current study indicated that competence genes *comX1* and *cgiA* are also induced in the blood for WCH16. D39 results also suggested that pneumococci harvested from the blood expressed a higher amount of mRNA for the putative SOS repair gene, *dinF*. *dinF* is the third gene in the competence-induced *cin-recA* operon (Mortier-Barriere *et al.*, 1998; Pearce *et al.*, 1995), and the microarray data indicating an up-regulation of competence genes *comB* and *comE* in the blood are consistent with the increase in *dinF* mRNA observed in this niche for D39.

In Chapter 4, D39 *cps2A* was found to be expressed similarly in all niches when examined by real-time RT-PCR. Likewise, *cps2A* was not identified as being differentially expressed by the D39 microarray hybridisations performed in this chapter. This was surprising considering both the essential requirement for CPS in virulence and the difference in CPS quantities between opacity phase variants (Austrian, 1981a; King *et al.*, 2004), and may suggest that CPS is not regulated at the transcriptional level (Morona *et al.*, 2000). However, several other genes involved in cell wall modification were found to be differentially expressed *in vivo*. Peptidoglycan N-acetylglucosamine deacetylase A (*pgdA*) was up-regulated 7.4-fold in the blood compared to the nasopharynx. PgdA deacetylases peptidoglycan, which provides resistance against degradation by lysozyme in mucosal secretions (Vollmer and Tomasz, 2000). Mutagenesis studies have also shown PgdA⁻ pneumococci to possess reduced virulence in an IP infection model, suggesting that it may be a virulence determinant (Vollmer and Tomasz, 2002), and results from the current study indeed support a role for it in the blood. Genes with roles in fatty acid metabolism were also up-regulated in the blood. Changes in the lipid composition of the cell membrane

occur in response to environmental conditions such as oxidative stress and temperature in *Lactobacillus helveticus* (Guerzoni *et al.*, 2001). Also, changes in membrane fluidity and fatty acid composition have been associated with pneumococcal opacity phenotype (Aricha *et al.*, 2004; Saluja and Weiser, 1995). The study presented in this chapter raises the possibility that niche-dependent alterations in membrane lipid composition also occur *in vivo*.

In contrast to D39, WCH16 cpsA was up-regulated 3.8-fold in the bloodstream compared to the nasopharynx of infected mice. The fold regulation was below the predetermined cut-off criteria, and the variation between spot intensity resulted in this value not being significant when analysed by Student's *t*-test (two-tailed). This was consistent with expression trends observed for cpsA by real-time RT-PCR, which also indicated substantial variation in expression level between WCH16.1 and WCH16.2. WCH16 and D39 exhibit different kinetics of pathogenesis. WCH16 establishes bacteraemia in the absence of significant lung infection following IN challenge, and maintains a sub-septic level of bacteraemia for at least 72 hr. Many WCH16-infected mice also develop meningitis within 72 hr. In contrast, mice IN-infected with D39 develop bacteraemia a few days after challenge which rapidly progresses to sepsis and death of the animal, usually within 24 hr of pneumococci first infiltrating the blood (Chapter 3). CPS has been shown to be essential to pneumococcal virulence (Austrian, 1981a), so it is surprising that WCH16 appeared to up-regulate *cps2A* expression in the bloodstream compared to in the nasopharynx, whereas D39 did not. However, due to the disparities between biological experiments, further experiments are needed to confirm the niche-specific differential expression of WCH16 cpsA.

5.3.2 Comparison of D39 microarray data with other published studies

Several studies have been published that identify genes essential for virulence or expressed more highly in certain niches and environmental conditions. Some of the major studies are presented in Table 5.5.

Genes that exhibited differential expression between niches in the current study were compared against other global virulence gene studies in the literature (Table 5.5). Of the various STM, DFI and microarray studies reported, an STM study by Hava and Camilli (2002), a DFI study by Marra et al. (2002) and a microarray study by Orihuela et al. (2004) were chosen to compare with the microarrays performed in this chapter. These studies represented a cross-section of currently published methods for examination of virulence genes on a global level. While all three methodologies have contributed to elucidating the relevance and expression of S. pneumoniae virulence genes on a global scale, each has met with limitations. STM uses transposons tagged with a unique DNA sequence marker to construct a bank of insertion mutants. The tag sequences from a pool of these mutants are amplified prior to co-infection of the relevant animal, and amplified again following the recovery of bacteria from the host. Hybridisation of tags to arrays comprising DNA from the pool of mutants indicates the presence or absence of each STM mutant following infection and, therefore, whether that particular gene is essential for virulence. In the study published by Hava and Camilli (2002), 6149 mutants were screened for abrogated virulence in an IN challenge lung infection model. However, as STM is based on the generation of random mutants, screening is not comprehensive. This may have been partly responsible for relatively few of the known virulence factors previously reported in the literature, such as genes of the CPS locus, PspA and CbpA, being identified by the three S. pneumoniae STM studies (Table 5.5). Also, while STM studies identify genes whose

Year	Author	Screening method	Strain (Serotype)
1998	Polissi <i>et al</i> .	STM	G54 (19F)
2001	Lau <i>et al</i> .	STM	0160993 (3)
2002	Marra <i>et al</i> .	DFI	D39 (2)
2002	Hava and Camilli	STM	TIGR4 (4)
2004	King <i>et al</i> .	cDNA microarray	6A-O (6A), 6A-T (6A), 6B-O (6A), 6B-T (6B)
2004	Orihuela <i>et al</i> .	cDNA microarray	T4X (4), T4R (4, un- encapsulated), D39 (2)
2006	Oggioni et al.	Real-time RT-PCR	TIGR4 (4)

Table 5.5: Published studies examining genes required for colonisation or invasive disease

This table lists the major studies in the literature that identify genes essential to virulence or expressed more highly in certain niches and environmental conditions, the type of experiment conducted, and the *S. pneumoniae* strain and serotype used. Shaded cells indicate the studies that were cross-referenced with results from this chapter.

products are important for virulence, they do not give an indication of their regulation. Specific promoter activity can be monitored by using a fluorescent or luminescent reporter The DFI studies performed by Marra et al. (2002) examined D39 strains protein. expressing gfp under the control of various pneumococcal promoters. They examined promoter activity in a mouse lung infection model, a gerbil otitis media model, and a mouse IP chamber implant model (requiring the surgical insertion of a chamber containing bacteria into the peritoneal cavity). Additionally, promoter activity under a variety of *in vitro* conditions that simulated different aspects of niches in the host were examined; low iron, high osmolarity, carbon dioxide, growth on blood agar and growth at different temperatures. The half-life of wild-type GFP is 26 hr (although derivatives exist with halflives of approximately 6 hr), so DFI is limited to examining promoter induction *in vitro* or following direct challenge of a niche and is unable to report transcriptional fluctuations in real time (Corish and Tyler-Smith, 1999). More recently, DNA microarray analysis has been used to complement existing DFI gene expression studies. Orihuela et al. (2004) performed microarray studies examining the regulation of S. pneumoniae genes in several models compared to a baseline control. The study identified genes that were up- or downregulated in the bloodstream of BALB/cJ mice following IN challenge with a D39 derivative, or in the brains of New Zealand white rabbit 4 hr after intracisternal infection with a TIGR4 derivative, compared to growth in C+Y media. To simulate nasopharyngeal colonisation, an un-encapsulated derivative of TIGR4 was co-cultured with Detroit-562 epithelial cells. RNA was extracted from the recovered bacteria and compared with that extracted from bacteria grown in tissue culture media. The authors raised several considerations in their discussion regarding their use of a different S. pneumoniae strain and disease model to examine each aspect of pathogenesis; carriage, meningitis and bacteraemia. Specifically, there is likely to be strain-related gene regulation *in vivo*, and also host factors and responses that are specific to each disease model. Therefore, the comparisons that can be made regarding differential gene expression between niches are limited.

33% of the genes reported in this chapter as exhibiting significant differences in expression between niches were also identified by at least one other study. However, only 8% of identified genes exhibited changes that could be directly compared to other studies by cross-referencing (e.g. pneumococci were examined in the same niche in both studies) and were found similarly regulated. These genes are shown in Table 5.6. Direct comparisons in gene expression between studies were not always possible due to the different criteria used, and different niches or conditions examined, in the various publications. Moreover, none of the studies in the literature (or the current study reported in this chapter) were completely comprehensive. Genes that appeared to be important to carriage or disease in this study and others, but were unable to be directly compared because of the aforementioned reasons, are listed in Table 5.7. Orihuela et al. (2004) performed a similar comparison between their work and the literature, and likewise found their data contained few genes that had previously been identified in other studies. Perhaps a reason for this disparity is the use of different S. pneumoniae strains in the studies, particularly considering that there may be strain-dependent differences in gene expression in vivo. Additionally, many of the studies examined different aspects of disease, and thus used different infection models. Another consideration is that the work presented in this chapter directly compares gene expression between niches. Thus, genes that are similarly up-regulated in all niches examined compared to growth in un-inducing conditions such as in vitro, were not identified. Such genes are likely to be important to multiple aspects of virulence, but not *in vitro* growth.

Gene (gene name, if available)	TIGR	Regulation in	Paper identified in ^b
	annotation	this work ^a	
competence factor transport protein (<i>comB</i>)	Sp0043	0.078 (N/B)	Marra (+3.7, blood) Orihuela (-48.8, CSF)
hypothetical protein	Sp0144	0.090 (N/B) 0.053 (L/B)	Marra (lung infection) Orihuela (+13.0, blood)
conserved hypothetical protein	Sp0239	0.20 (N/L)	Orihuela (+3.1, ECC)
choline binding protein G	Sp0390	0.097 (N/B)	Orihuela (+1.7, blood)
protective antigen A (prtA)	Sp0641	0.28 (N/B) 0.23 (L/B)	Marra (+2.6, blood)
aminotransferase, class-V	Sp1094	0.11 (N/B)	Marra (+1.9, blood agar; +1.9 otitis media; +4.8, CO ₂)
6-phospho-beta-galactosidase (<i>lacG</i>)	Sp1184	15.2 (N/L) 9.6 (N/B)	Orihuela (+6.5, ECC)
hypothetical protein	Sp1558	0.17 (N/B)	Marra (+, blood)
putative ATP-dependent RNA	Sp1586	0.21 (N/L)	Orihuela (+3.1, ECC)
ROK family protein	Sp1675	30.5 (N/L)	King (+4.64, transparent)
ATP-binding protein (<i>amiE</i>)	Sp1888	0.11 (N/B)	Orihuela (+2.4, blood; +2.9, CSF)
MATE efflux family protein DinF	Sp1939	0.24 (N/L) 0.13 (L/B)	Hava (essential in lung)
choline binding protein A (<i>cbpA</i>)	Sp2190	22.0 (N/B) 14.5 (N/L) 7.1 (L/B)	Hava (essential in lung) Orihuela (+2.3, ECC)

Table 5.6: Comparison of differentially expressed and essential virulence genes reported in the literature with genes identified in this study: genes exhibiting similarity

^a Indicates the difference in mRNA between niches as determined by microarray analysis in this study (niches that were compared are in parentheses).

^b Results from the following experiments: Hava and Camilli, 2002 (niche in which gene is essential); King *et al.*, 2004 (fold-difference in gene transcript between opacity phenotypes); Marra *et al.*, 2002 (promoter induction ratio under specified *in vitro* condition or *in vivo* niche); Orihuela *et al.*, 2004 (fold-difference in gene transcript in a specific niche compared to a baseline control). These studies are described Section 5.3.2. CO_2 refers to carbon dioxide shift, ECC refers to an epithelial cell contact model, and CSF refers to cerebrospinal fluid. + and – indicate up- and down-regulation of gene expression under the specified condition, respectively.

Table 5.7: Comparison of differentially expressed and essential virulence genes identified in the literature and this study: genes that were identified in multiple studies, but in different conditions or niches

Gene (gene name, if available)	TIGR annotation	Regulation in this work ^a	Paper identified in ^b
GTP-binding protein	Sp0004	11.8 (N/B) 9.2 (L/B)	Marra (+1.1, blood)
bacteriocin (blpU)	Sp0041	0.18 (L/B)	Orihuela (-3.1, CSF)
conserved hypothetical protein	Sp0095	0.19 (L/B)	Hava (essential in lung)
glycosyl transferase	Sp0102	0.14 (N/B)	Hava (essential in lung)
transcriptional regulator (mutR)	Sp0141	0.21 (N/L)	Orihuela (+3.0, blood)
phosphoglycerate mutase family protein	Sp0240	0.15 (L/B)	Hava (essential in lung)
transcriptional regulator, MerR family	Sp0501	0.21 (N/B)	Orihuela (-5.7, CSF)
beta-lactam resistance factor	Sp0615	0.13 (N/B)	Orihuela (+4.0, ECC)
putative PTS system IIA component	Sp0645	6.1 (N/L)	Hava (essential in lung)
chorismate binding enzyme	Sp0655	0.14 (N/B) 0.16 (L/B)	Hava (essential in lung)
endo-ß- <i>N</i> -acetylglucosaminldase (<i>lytB</i>)	Sp0965	0.11 (L/B)	Marra (+7.96, static temperature shift) Orihuela (+3.5, ECC)
acetyltransferase, GNAT family macrolide-efflux protein	Sp1023 Sp1110	0.13 (N/B) 0.14 (N/B)	Hava (essential in lung) Orihuela (-4.9, CSF)
SAP domain protein	Sp1292	15.9 (N/L) 10.1 (N/B)	Hava (essential in lung)
IS66 family element, Orf3, degenerate	Sp1311	0.11 (N/B)	Orihuela (-27.4, CSF)

TIGR	Regulation in	Paper identified in ^b
annotation	this work ^a	
Sp1321	28.4 (N/L)	Hava (essential in lung)
	10.7 (N/B)	
Sp1328	34.3 (N/L)	Hava (essential in lung)
	12.9 (N/B)	
Sp1439	17.7 (N/L)	Marra (lung infection)
Sp1479	0.14 (N/B)	Marra (+2.30, otitis media)
		Orihuela (-3.2, CSF)
Sp1572	0.18 (I /B)	Oribuela (84 blood)
Sp1372	0.18 (L/D)	Offituera (-8.4, 01000)
Sp1143	10.5 (N/L)	Hava (essential in lung)
Sp1710	0.20 (L/B)	Orihuela (+2.2, CSF)
Sp1772	13.4 (N/L)	Hava (essential in lung)
Sp1856	0.13 (N/B)	Hava (essential in lung)
	0.20 (L/B)	
Sp2225	0.009 (N/D)	Oribuele (15.0, CSE)
SP2233	0.070 (IN/D)	Officia (-13.0, CSF)
Sp2240	0.14 (L/B)	Marra (lung infection)
	TIGR annotation Sp1321 Sp1328 Sp1439 Sp1479 Sp1572 Sp1143 Sp1710 Sp1856 Sp2235 Sp2240	TIGR Regulation in this work ^a Sp1321 28.4 (N/L) 10.7 (N/B) Sp1328 34.3 (N/L) 12.9 (N/B) Sp1439 17.7 (N/L) Sp1572 0.14 (N/B) Sp1143 10.5 (N/L) Sp1710 0.20 (L/B) Sp1856 0.13 (N/B) 0.20 (L/B) Sp2235 0.098 (N/B)

^a Indicates the difference in mRNA between niches as determined by microarray analysis in this study (niches that were compared are in parentheses).

^b Results from the following experiments: Hava and Camilli, 2002 (niche in which gene is essential); King *et al.*, 2004 (fold-difference in gene transcript between opacity phenotypes); Marra *et al.*, 2002 (promoter induction ratio under specified *in vitro* condition or *in vivo* niche); Orihuela *et al.*, 2004 (fold-difference in gene transcript in a specific niche compared to a baseline control). These studies are described in Section 5.3.2. CO_2 refers to carbon dioxide shift, ECC refers to an epithelial cell contact model, and CSF refers to cerebrospinal fluid. + and – indicate up- and down-regulation of gene expression under the specified condition, respectively.
5.3.3 Considerations and conclusion

In this chapter, some of the limitations faced by other microarray studies were addressed. Primarily, a single model of infection was used to examine pneumococci involved in colonisation, lung disease and IPD. This satisfies the considerations raised by Orihuela et al. (2004) pertaining to the use of different animal models and pneumococcal strains to recover sufficient bacteria for microarray analysis. In their study, bacteraemia was examined for a D39 derivative following intratracheal infection of BABLB/cJ mice, while a TIGR4 derivative was used to induce meningitis in rabbits by intracisternal injection and then harvested 4 hr post-infection to examine pneumococcal behaviour in the brain. The limited overlap between genes identified by the three STM studies in the literature, each of which examined a different strain, is likely to be partly due to STM screenings not being completely comprehensive, as not all genes in each genome would have been tested. It may also suggest strain-specific roles and requirements for many pneumococcal genes. Furthermore, transcription regulation by two component signal transduction systems HK/RR04 and HK/RR09 response regulators RR04 and RR09 have both been shown to exhibit serotype-specific differences in regulatory activities (Hendriksen et al., 2007; McCluskey et al., 2004). Therefore, studies using different pneumococcal serotypes to compare multiple facets of disease must be viewed with these compounding considerations in mind. In the present work, bacterial expression in the nasopharynx, lungs and blood were examined following IN challenge under anaesthesia, and subsequent infection of blood and brain occurred by normal pathogenic routes. Bacteria were harvested 72 hr post-challenge, a time-point at which pneumococci were considered to be adapted to each specific niche and had presumably been present for a sufficient duration to allow the generation of a host response. As pneumococci had entered the bloodstream and brain by natural progression of disease rather than following direct challenge of these niches, the possibility that high inocula created unnatural levels of infection or host responses was reduced.

However, the use of the continuous infection model was not without its own drawbacks. The main limitation was inconsistent bacterial levels in both the lungs and blood at 72 hr following IN infection of mice, which is unavoidable in a continuous IN infection model due to animal-to-animal variation. Additionally, the genes that failed to meet the analysis criteria described in Section 5.2.2 were not examined. Conducting further array experiments might generate reliable data for these genes and provide a more comprehensive indication of comparative *in vivo* gene expression for the two serotypes examined. Such experiments could not be conducted during this study because of time limitations. It is possible that, due to the sensitivity limitations of microarray and the many algorithms and restrictions that are necessary to analyse microarray data, transcripts present in low abundance or that exhibit only minor fluctuations in expression between niches, may not be identifiable using this technique. As indicated by the real-time RT-PCR results reported in Chapter 4, genes such as *pspA* may be up-regulated significantly in both colonising and bacteraemic pneumococci compared to in vitro, indicating an importance in colonisation and systemic disease. However, the difference in expression between niches may not be large enough to detect by microarray analysis. Therefore, the use of more sensitive RNA quantitation techniques such as real-time RT-PCR is essential to corroborate microarray data. Despite these considerations, the study described in this chapter is the first global transcriptional analysis of pneumococci involved in colonisation versus invasive disease. Moreover, bacteria were recovered from a continuous mouse model. The microarray experiments identified niches in which metabolism, transcription and translation, cell wall and membrane synthesis and competence genes are up-regulated. This chapter also identifies some genes that have also been reported by other studies as having niche-specific importance and, therefore, are likely to be relevant to several pneumococcal serotypes.

CHAPTER 6 – Final discussion

6.1 Summary of this work

S. pneumoniae is a major human pathogen, and is responsible for significant morbidity and mortality in both developed and developing countries. It is adept at establishing and maintaining nasopharyngeal colonisation, with 5-95% of healthy children under five years of age carrying at least one serotype at any time depending on age, geographical area, genetic background, and socio-economic conditions (Bogaert et al., 2004a; Crook et al., 2004). Occasionally, perhaps due to the host possessing a compromised or immature immune system, the pneumococcus translocates from this niche to other sites within the host, where it can cause diseases such as otitis media, meningitis, sepsis and pneumonia. Worldwide, pneumococcal diseases are estimated to be responsible for almost one million deaths of children under five each year, most of which occur in developing countries (World Health Organization, 2007). S. pneumoniae also causes a significant disease burden in developed countries, with approximately seven million causes of otitis media being attributed to pneumococci annually in the USA, and one in 500 children suffering IPD before two years of age in Australia (Dowell et al., 1999; Mackenzie et al., 2005). The mechanism of regulation of virulence factors that enable the pneumococcus to be both an asymptomatic commensal and highly invasive pathogen is

Although the differential expression of several virulence genes poorly understood. between bacteria grown *in vitro* and harvested from a sepsis model has been previously described (Ogunniyi et al., 2002), at the commencement of this work, no studies had examined quantitative differences in S. pneumoniae virulence gene mRNA levels between different host niches during disease pathogenesis. Furthermore, pneumococcal virulence gene expression during nasopharyngeal carriage had not been investigated. This thesis aimed to examine the transcriptional patterns of S. pneumoniae virulence genes in vivo, and hence gain a better understanding of pneumococcal behaviour during carriage and disease. Specifically, this required the establishment of an animal infection model from which reasonable numbers of bacteria could be harvested from multiple sites free from contamination with other bacteria, and the optimisation of RNA extraction and amplification procedures. The animal model needed to be appropriate for several different S. pneumoniae strains to allow the examination of strain-specific expression patterns. Realtime RT-PCR and microarray analyses were used to examine the transcriptional profiles of S. pneumoniae recovered from different niches in this model, for specific virulence factors and on a genomic scale, respectively.

6.1.1 Recovery of pneumococci from an IN infection model and RNA extraction and modifications

Previously, the inability to recover reasonable and consistent numbers of bacteria from different *in vivo* sites within the one animal species had necessitated the use of multiple invasive infection models, none of which had permitted the examination of pneumococcal gene expression during carriage.

In Chapter 3, an IN infection model for CD-1 mice was described (adapted from

Wu et al., 1997) whereby pneumococci could be harvested from distinct anatomical sites, namely the nasopharynx, lungs, blood and brain of a single mouse with negligible crosscontamination from other host niches, other microbial flora, or host cells. Mice exhibiting sepsis possess high numbers of bacteria in the bloodstream. However, potential crosscontamination of tissue samples with pneumococci circulating in the blood has often been overlooked in other studies that have enumerated bacteria in host tissue, such as the lungs. In the current model, this problem was circumvented by removing the blood by PBS perfusion prior to organ harvesting. This thesis aimed to examine pneumococcal gene expression during colonisation and disease progression using real-time RT-PCR and microarray analysis. Previous studies had encountered difficulties due to the limited amount of mRNA that could be harvested from in vivo sites. This was overcome by tailoring the procedure by which pneumococci are harvested from mice to optimise yield of bacteria, to minimise RNA degradation, and by linearly amplifying the extracted RNA prior to use in assays (Chapter 3). Furthermore, host tissue contamination from lung and brain samples was reduced by differential centrifugation prior to RNA extraction, followed by the enrichment of RNA samples for prokaryotic RNA species. This enabled downstream linear amplification procedures to be conducted with greater efficiency.

The infection model described here overcomes many of the complications raised by Orihuela *et al.* (2004) regarding the mRNA studies they performed. Their study utilised several different animal, cell culture and challenge models to simulate infection *in vivo* in order to recover sufficient bacteria for transcriptional analysis. A D39 intratracheal challenge model was used to examine bacteraemia, whereas meningitis was modelled by intracisternal injection of a TIGR4 derivative. Nasopharyngeal carriage was simulated *in vitro* by co-culturing an un-encapsulated TIGR4 derivative with Detroit-562 nasopharyngeal cells. The main advantage of the IN infection model and RNA processing described in Chapter 3 is that gene expression in the nasopharynx, lungs, blood and brain could be examined using a single pneumococcal strain and a continuous infection model. This overcomes the potential interpretational complications imposed by strain- and animal model-specific transcriptional regulation. Furthermore, an IN infection model simulates the normal route of acquisition of *S. pneumoniae*, and progression to the blood and brain. This also avoids introducing unnaturally high levels of bacteria directly into these niches, which is a consideration when using challenge models such as IP and intracisternal injection, and allows for natural host responses to be generated. As such, this model appears well suited to the examination of immune responses to *S. pneumoniae* infection *in vivo*.

6.1.2 D39 and WCH16 possess different pathogenesis profiles

In the CD-1 IN infection model established in Chapter 3, both D39 (a well studied serotype 2 laboratory strain originally isolated by Avery in 1916 [Avery *et al.*, 1944]) and WCH16 (a more recent serotype 6A clinical isolate) were shown to persist in the nasopharynx for 72 hr and progress to other niches. By enumerating levels of pneumococci in specific sites using this model, performing tail bleeds and monitoring disease progression in individual mice using luminescent D39 and WCH16 derivatives, two distinct pathogenesis patterns emerged (Figure 6.1). D39 pathogenesis is characterised by the development of lung disease followed by infiltration of the blood by pneumococci (Chapter 3 and 4). Mice exhibiting bacteraemia quickly develop sepsis and often die within 24 hr after the establishment of bacteraemia. WCH16 is also virulent and able to cause invasive disease, usually progressing to the bloodstream and brain within 48 hr post-challenge. However, this often occurs in the absence of detectable lung disease, which



Figure 6.1: Distinct pathogenicity profiles of D39 and WCH16

The pathogenesis of D39 and WCH16 infection in CD-1 mice after IN challenge, as determined in Chapter 3. Bold arrows indicate the preferred progression of infection for each strain. Grey arrows represent secondary routes of pathogenesis.

perhaps suggests that the preferred route for WCH16 to infiltrate the bloodstream is by translocating across the nasopharyngeal epithelium. A further disparity between the two strains is that mice challenged IP with approximately 2×10^7 WCH16 bacteria often maintain a constant, sub-septic, level of bacteraemia for several days and have double the median survival time of mice similarly infected with D39 (Chapter 3). The reasons for the different outcomes of bacteraemia were not elucidated in this work.

6.1.3 D39 and WCH16 genes that exhibit contrasting *in vivo* regulation

The different pathogenicity profiles observed for D39 and WCH16 in Chapter 3 raises the possibility that pneumococci might exhibit strain- or serotype- specific virulence gene regulation patterns *in vivo*, which may influence the course of disease. This appeared to be the case for several virulence factors examined by real-time RT-PCR in Chapter 4, as depicted in Figure 6.2. This figure summarises the in vivo niches in which specific pneumococcal virulence factors are up-regulated, and so perhaps are more likely to be important for virulence in that niche. An interesting discrepancy between the strains was the marked up-regulation of D39 nanA in the nasopharynx compared to other niches, but in WCH16 there was little *in vivo* transcriptional regulation relative to *in vitro* levels. NanA is thought to contribute to carriage by both desialylating the cell surfaces of competing microflora and modifying host cell glycoconjugate receptors, providing a competitive advantage for the pneumococcus and promoting adherence to the nasopharyngeal epithelium (King et al., 2004; Shakhnovich et al., 2002). Hence, it might be expected that a lack of transcriptional regulation for WCH16 nanA during pathogenesis would be reflected by attenuated colonisation compared to D39, which exhibits a marked upregulation of nanA in the nasopharynx. However, in this study both D39 and WCH16 were able to establish similar levels of colonisation in CD-1 mice for at least 72 hr after IN



 Table 6.2: Importance of specific S. pneumoniae virulence factors in vivo

Genes that were found to be up-regulated in specific *in vivo* niches by real-time RT-PCR in this study, relative to other niches and *in vitro*. Arrows indicate the preferred progression of disease for each strain.

challenge, based on bacterial numbers present in nasopharyngeal washings. Of course, it is possible that increased levels of NanA promote tighter pneumococcal adherence to the nasopharyngeal epithelium and that these bacteria may not be eluted by washing with a buffer (Briles *et al.*, 2005). This possibility was not examined in the current work, but could be determined in the future by enumerating bacteria in the nasopharyngeal tissue. PsaA also appears to exhibit different expression patterns during D39 and WCH16 infections. PsaA-negative mutants have been shown in other studies to exhibit greatly attenuated colonisation, lung disease and bacteraemia in mice, and also possess defective oxidative stress responses (Marra *et al.*, 2002b; McAllister *et al.*, 2004; Tseng *et al.*, 2002). This work suggests that PsaA plays a larger role during WCH16 sepsis than carriage, as it is more highly expressed in the blood than in other niches. *In vivo* expression was constitutive during D39 infection, which may indicate that PsaA contributes to both colonisation and invasive disease for D39, either in its primary role as a component of a Mn^{2+} transporter or in other undefined roles.

The work in thesis clearly indicates that *in vivo* transcriptional regulation of some virulence factors is strain-specific. It is also possible that strain-specific expression levels of certain genes such as *ply* may be partly responsible for corresponding differences in pathogenicity between strains. This highlights the importance of examining several *S. pneumoniae* strains and serotypes before drawing generalised conclusions about the contributions of specific proteins to virulence. This may be of particular importance when considering candidates for inclusion in protein-based pneumococcal vaccines, which need to provide broad protection against carriage and invasive disease caused by a wide spectrum of strains/serotypes.

This study found that several important virulence factors exhibited similar *in vivo* expression patterns for both D39 and WCH16, which may indicate the essential requirement of these factors for colonisation and virulence in multiple serotypes (Chapters 4 and 5). Expression of the putative nasopharyngeal adhesin CbpA is noticeably upregulated during carriage for both D39 and WCH16, which supports its ascribed roles in carriage (Rosenow *et al.*, 1997; Zhang *et al.*, 2000). Likewise, D39 and WCH16 bacteria recovered from the nasopharynx express higher amounts of *spxB* transcript than bacteria recovered from other niches. The increased expression of *cbpA* and *spxB* in the nasopharynx is likely to help colonisation of this site by promoting adherence to the epithelium and suppressing the growth of competing microflora (Pericone *et al.*, 2000; Rosenow *et al.*, 1997; Spellerberg *et al.*, 1996).

Interestingly, both D39 and WCH16 express higher levels of *ply* in the nasopharynx than the lungs or bloodstream - niches where the cytotoxin is considered to be crucial to disease progression and development (Berry *et al.*, 1992; Cockeran *et al.*, 2002; Feldman *et al.*, 1990; Houldsworth *et al.*, 1994; Paton *et al.*, 1984; Steinfort *et al.*, 1989). This finding reinforces the notion that IPD is incidental to carriage, with pneumococci experiencing little selective pressure in the blood. Moreover, bacteraemia is counterproductive to the spread of *S. pneumoniae* in the community, and it is likely that the low incidence of IPD relative to the numbers of *S. pneumoniae* carriers has allowed strains that are capable of causing severe morbidity and mortality to persist within populations rather than being eliminated due to negative selection. With this in mind, it is unsurprising that many pneumococcal virulence factors involved in disease have also been implicated as being important for carriage. PspA is one such virulence factor, capable of promoting both

carriage and bacteraemia by inhibiting apolactoferrin-mediated killing of pneumococci in the nasopharynx, and also reducing complement-mediated opsonophagocytosis in the blood (Shaper *et al.*, 2004; Tu *et al.*, 1999). The real-time RT-PCR results in this study support roles for PspA in the nasopharynx and the blood for D39, with increased expression observed in both these niches, relative to the lungs and *in vitro*. Likewise, microarray data for D39 and WCH16 indicated no significant difference in *pspA* expression between bacteria recovered from the nasopharynx and blood. In addition, several virulence factors appear to be important in both carriage and meningitis. This is discussed in Section 6.1.5.

6.1.5 Bacteria recovered from the nasopharynx and brain exhibit similar expression of some key virulence genes

Pneumococci can be visually separated into two distinct opacity phenotypes, based on colony morphology when viewed on transparent agar (Weiser *et al.*, 1994). Despite there being few *in vitro* transcriptional differences between opaque and transparent variants reported, increased protein levels of several virulence factors such as CbpA and PspA have been associated with a specific opacity phenotype (Kim and Weiser, 1998; King *et al.*, 2004; Rosenow *et al.*, 1997). Furthermore, the transparent phenotype has been associated with carriage and bacteria crossing the BBB, whereas bacteria in the blood are predominantly of the opaque phenotype (Cundell *et al.*, 1995b; Ring *et al.*, 1998; Weiser *et al.*, 1994; Weiser and Kapoor, 1999). The current study compared pneumococci recovered from the nasopharynx and the brain, and clearly demonstrated that WCH16 pneumococci involved in colonisation and early meningitis exhibit similar expression profiles for key virulence genes, as summarised in Figure 6.2. Specifically, *cbpA*, *ply*, *piaA* and *spxB* were found to be up-regulated in both these niches compared to bacteria recovered from the bloodstream and/or lungs. Moreover, microarray analysis indicated that genes with roles in fatty acid metabolism are up-regulated in the blood, which supports previous studies by other groups associating membrane fluidity and fatty acid composition with pneumococcal opacity phenotype (Aricha *et al.*, 2004; Saluja and Weiser, 1995). While these observations are not direct evidence supporting the existence of an 'opacity phenotype switch', they do suggest there may be a link between transcriptional changes, opacity phenotype and *in vivo* niche.

6.1.6 Global microarray analysis identifies differentially expressed genes during carriage or disease

Microarray analysis of D39 *in vivo*-derived pneumococci identified genes that exhibit niche-specific expression that were not specifically examined by real-time RT-PCR (Chapter 5). An Array 900 MPX Kit (Genisphere) was used for microarray experiments performed in this study, which is optimised for smaller quantities of RNA. Of particular interest, the microarray analysis found that mRNAs for competence genes *comB* and *comE* are clearly up-regulated in the blood compared to the nasopharynx for both D39 and WCH16. *comX1* and *cglA* are also up-regulated in this niche for WCH16. These results are consistent with the findings from other groups indicating increased transcription of competence genes occurs during exponential growth (Claverys and Havarstein, 2002). The expression of *dinF* (a gene in the competence-induced *cin-recA* operon (Mortier-Barriere *et al.*, 1998; Pearce *et al.*, 1995)) parallels that of competence genes, indicating a role for both competence and DNA repair during sepsis. These findings disagree with a recent study by Oggioni *et al.* (2006), which reported TIGR4 competence genes to be repressed in the blood. This may indicate competence has serotype-specific importance. The CBPs *lytB*, encoding a cell wall hydrolase essential for daughter cell separation (Garcia *et al.*, *et al*

1998), and *cbpG*, encoding a putative serine protease (Gosink *et al.*, 2000), are also expressed more highly during sepsis than carriage for D39, with pneumococci recovered from the blood possessing approximately 10-fold more of both transcripts (Chapter 5). The finding for *cbpG* strongly supports microarray data from Orihuela *et al.* (2004) and mutagenesis studies performed by Gosink *et al.* (2000), which found *cbpG* to be induced in the blood and necessary for full virulence in this niche. Conversely, the absence of LytB has previously been shown to have no effect on sepsis in an IP infection model (Gosink *et al.*, 2000). However, those experiments used a serotype 4 *S. pneumoniae* strain, and so it is possible that LytB has a role in sepsis for D39, but not for the serotype 4 strain used by Gosink *et al.* (2000).

Microarray analysis also identified several genes with unknown functions exhibiting large fold-differences in expression between colonising bacteria and bacteria involved in bacteraemia, thus indicating that they may contribute to virulence. In particular, hypothetical proteins encoded by Sp0144 and Sp1558 exhibit significantly elevated expression in the blood compared to the nasopharynx, and have been similarly identified by either DFI or other microarray studies (Marra *et al.*, 2002a; Orihuela *et al.*, 2004b). Additionally, a putative neuraminidase encoded by Sp1326 was up-regulated in the respiratory tract compared to the blood. The differential expression of these genes could be confirmed using real-time RT-PCR, and it may also be interesting to examine their role in further by mutagenesis studies.

6.1.7 Conclusion

An understanding of how the pneumococcus responds transcriptionally to different environments in the host has been sought for many years, but it has hitherto remained elusive due to technical constraints. Although other global *S. pneumoniae* virulence gene discovery and expression studies have been reported in the literature (Hava and Camilli, 2002; Lau *et al.*, 2001; Marra *et al.*, 2002a; Polissi *et al.*, 1998), knowledge of differential niche-specific *in vivo* gene expression is limited. The work in this thesis has examined pneumococcal gene expression in a mouse model on both a specific and genomic level, and has determined *in vivo* transcriptional patterns. This is the first work that has examined multiple facets of *S. pneumoniae* infection using bacteria recovered from a single mouse model, and therefore is novel and complements existing studies in the literature. It reflects pneumococcal pathogenesis better than other studies which have required use of direct niche-specific challenge models and multiple *S. pneumoniae* strains to simulate human infection. Moreover, it suggests the existence of strain-specific gene expression for several key virulence factors, which may have implications for the development of protein-based pneumococcal vaccines. Similar gene expression profiles for pneumococci during carriage and early meningitis also allude to a link between transcriptional regulation, opacity phenotype, and host niche.

6.2 Future directions

Due to time constraints and technical complications, replicate microarrays using samples recovered from distinct biological experiments could not be completed for all niche comparisons during this study. In addition, the WCH16 data presented in Chapter 5 were inconclusive due to considerable variation between experiments. Replicate biological experiments and corresponding micorarray experiments comparing WCH16 expression *in vivo* need to be completed to clarify the results. Furthermore, selected genes that were found to be differentially expressed by microarray analysis should be validated by realtime RT-PCR using specific oligonucleotides. This is particularly so for the competence genes *comB* and *comE*, which were not found to be up-regulated in the blood compared to Detroit-562 epithelial cells by Orihuela *et al.* (2004), but were found to be induced during sepsis in the current study for two strains of different serotype. Further analysis is also needed for the genes encoding CBPs *cbpG* and *lytB*. LytB demonstrated increased transcription in the bloodstream of infected mice for D39, despite studies by other groups using a serotype 4 strain *lytB*⁻ mutant finding no attenuation in sepsis compared to the WT (Gosink *et al.*, 2000). Therefore, mutagenesis studies examining the role of *lytB* in D39 virulence would help elucidate the role of this CBP in sepsis and indicate whether it is a strain-dependent virulence determinant. A better understanding of the disparities in transcriptional profiles between strains is important.

It is important to note that none of the studies examining global pneumococcal gene expression *in vivo* using techniques such as STM and microarray analysis have been entirely comprehensive, including the microarray data presented in Chapter 5 of this thesis (Hava and Camilli, 2002; King *et al.*, 2004; Lau *et al.*, 2001; Orihuela *et al.*, 2004b; Polissi *et al.*, 1998). Nevertheless, the existing studies can be viewed in parallel with each other, suggesting that a combination of approaches needs to be employed to gain a complete picture of pneumococcal regulatory behaviour *in vivo*. The use of bioluminescence could be extended to examine *in vivo* promoter activity, by constructing a plasmid possessing a *lux* cassette under the control of specific promoters. Lux would be the preferred reporter for such a system, as the Lux proteins require less time to produce a signal than GFP (which undergoes significant post-translational modifications, requiring over 4 hr to form the fluorophore) (Cody *et al.*, 1993; Heim *et al.*, 1994). Lux proteins also have a significantly shorter half-life than GFP, which would allow promoter kinetics to be

monitored in real time (Cody *et al.*, 1993; Tombolini and Jansson, 1998). From a host perspective, many symptoms of IPD are thought to be a result of cyclic interactions between *S. pneumoniae* and the host, leading to an uncontrolled inflammatory response. Using real-time RT-PCR to examine the host cytokine levels and pneumococcal gene expression simultaneously may give a clearer picture of both pneumococcal behaviour and the host's response. This may also elucidate host-pathogen responses that lead to the pneumococcus either being cleared from a niche, or progressing to invasive disease.

Bibliography

Alloing, G., Trombe, M.C. and Claverys, J.P. (1990) The ami locus of the gram-positive bacterium *Streptococcus pneumoniae* is similar to binding protein-dependent transport operons of gram-negative bacteria. *Mol. Microbiol.* **4**: 633-644.

Andersson, B., Eriksson, B., Falsen, E., Fogh, A., Hanson, L.A., Nylen, O., Peterson, H. and Svanborg Eden, C. (1981) Adhesion of *Streptococcus pneumoniae* to human pharyngeal epithelial cells in vitro: differences in adhesive capacity among strains isolated from subjects with otitis media, septicemia, or meningitis or from healthy carriers. *Infect. Immun.* **32**: 311-317.

Andersson, B., Leffler, H., Magnusson, G. and Svanborg Eden, C. (1983) Molecular mechanisms of adhesion of *Streptococcus pneumoniae* to human oropharyngeal epithelial cells. *Scand. J. Infect. Dis. Suppl.* **39**: 45-47.

Appelbaum, P.C. (1995) New prospects for antibacterial agents against multidrugresistant pneumococci. *Microb. Drug Resist.* **1**: 43-48.

Aricha, B., Fishov, I., Cohen, Z., Sikron, N., Pesakhov, S., Khozin-Goldberg, I., Dagan, R. and Porat, N. (2004) Differences in membrane fluidity and fatty acid composition between phenotypic variants of *Streptococcus pneumoniae*. *J. Bacteriol.* **186**: 4638-4644.

Armstrong, R.R. (1931) A swift and simple method for deciding pneumococcal "type". *Br. Med. J.* 1: 214-215.

Austrian, R. (1981a) Some observations on the pneumococcus and on the current status of pneumococcal disease and its prevention. *Rev. Infect. Dis.* **3**(suppl): S1-17.

Austrian, R. (1981b) Pneumococcus: the first one hundred years. *Rev. Infect. Dis.* **3**: 183-189.

Avery, O.T., MacLeod, C.M. and McCarty, M. (1944) Studies on the chemical nature of the substance inducing transformation of pneumococcal types. Induction of transformation by a deoxyribonucleic acid fraction isolated from pneumococcus type III. *J. Exp. Med.* **79**: 137-158.

Balachandran, P., Brooks-Walter, A., Virolainen-Julkunen, A., Hollingshead, S.K. and Briles, D.E. (2002) Role of pneumococcal surface protein C in nasopharyngeal carriage and pneumonia and its ability to elicit protection against carriage of *Streptococcus pneumoniae*. *Infect. Immun.* **70**: 2526-2534.

Balachandran, P., Hollingshead, S.K., Paton, J.C. and Briles, D.E. (2001) The autolytic enzyme LytA of *Streptococcus pneumoniae* is not responsible for releasing pneumolysin. *J. Bacteriol.* **183**: 3108-3116.

Barry, B., Muffat-Joly, M., Bauchet, J., Faurisson, F., Gehanno, P., Pocidalo, J.J. and Carbon, C. (1996) Efficacy of single-dose ceftriaxone in experimental otitis media induced by penicillin- and cephalosporin-resistant *Streptococcus pneumoniae*. *Antimicrob*. *Agents Chemother.* **40**: 1977-1982.

Beard, S.J., Salisbury, V., Lewis, R.J., Sharpe, J.A. and MacGowan, A.P. (2002) Expression of *lux* genes in a clinical isolate of *Streptococcus pneumoniae*: Using bioluminscence to monitor gemifloxacin activity. *Antimicrob. Agents Chemother.* **46**: 538-542.

Benton, K.A., Everson, M.P. and Briles, D.E. (1995) A pneumolysin-negative mutant of *Streptococcus pneumoniae* causes chronic bacteremia rather than acute sepsis in mice. *Infect. Immun.* **63**: 448-455.

Bergeron, Y., Ouellet, N., Deslauriers, A.-M., Simard, M., Olivier, M. and Bergeron, M.G. (1998) Cytokine kinetics and other host factors in response to pneumococcal pulmonary infection in mice. *Infect. Immun.* **66**: 820-822.

Bermpohl, D., Halle, A., Freyer, D., Dagand, E., Braun, J.S., Bechmann, I., Schroder, N.W. and Weber, J.R. (2005) Bacterial programmed cell death of cerebral endothelial

cells involves dual death pathways. J Clin. Invest. 115: 1607-1615.

Berry, A.M., Lock, R.A., Hansman, D. and Paton, J.C. (1989a) Contribution of autolysin to virulence of *Streptococcus pneumoniae*. *Infect. Immun.* **57**: 2324-2330.

Berry, A.M., Ogunniyi, A.D., Miller, D.C. and Paton, J.C. (1999) Comparative virulence of *Streptococcus pneumoniae* strains with insertion-duplication, point, and deletion mutations in the pneumolysin gene. *Infect. Immun.* **67**: 981-985.

Berry, A.M. and Paton, J.C. (1996) Sequence heterogeneity of PsaA, a 37-kilodalton putative adhesin essential for virulence of *Streptococcus pneumoniae*. *Infect. Immun.* **64**: 5255-5262.

Berry, A.M. and Paton, J.C. (2000) Additive attenuation of virulence of *Streptococcus pneumoniae* by mutation of the genes encoding pneumolysin and other putative pneumococcal virulence proteins. *Infect. Immun.* **68**: 133-140.

Berry, A.M., Paton, J.C. and Hansman, D. (1992) Effect of insertional inactivation of the genes encoding pneumolysin and autolysin on the virulence of *Streptococcus pneumoniae* type 3. *Microb. Pathog.* **12**: 87-93.

Berry, A.M., Yother, J., Briles, D.E., Hansman, D. and Paton, J.C. (1989b) Reduced virulence of a defined pneumolysin negative mutant of *Streptococcus pneumoniae*. *Infect. Immun.* **57**: 2037-2042.

Blunden, A.S., Hannant, D., Livesay, G. and Mumford, J.A. (1994) Susceptibility of ponies to infection with *Streptococcus pneumoniae* (capsular type 3). *Equine Vet. J.* 26: 22-28.

Bogaert, D., de Groot, R. and Hermans, P.W.M. (2004a) *Streptococcus pneumoniae* colonisation: the key to pneumococcal disease. *Lancet* **4**: 144-154.

Bogaert, D., Hermans, P.W., Adrian, P.V., Rumke, H.C. and de Groot, R. (2004b) Pneumococcal vaccines: an update on current strategies. *Vaccine* **22**: 2209-2220. **Boulnois, G.J.** (1992) Pneumococcal proteins and the pathogenesis of disease caused by *Streptococcus pneumoniae*. *J. Gen. Microbiol.* **138**: 249-259.

Boulnois, G.J., Paton, J.C., Mitchell, T.J. and Andrew, P.W. (1991) Structure and function of pneumolysin, the multifunctional, thiol-activated toxin of *Streptococcus pneumoniae*. *Mol. Microbiol.* **5**: 2611-2616.

Braun, J., Novak, R., Gao, G., Murray, P. and Shenep, J. (1999) Pneumolysin, a protein toxin of *Streptococcus pneumoniae*, induces nitric oxide production from macrophages. *Infect. Immun.* **67**: 3750-3756.

Braun, J.S., Sublett, J.E., Freyer, D., Mitchell, T.J., Cleveland, J.L., Tuomanen, E.I. and Weber, J.R. (2002) Pneumococcal pneumolysin and H(2)O(2) mediate brain cell apoptosis during meningitis. *J Clin. Invest.* **190**: 19-27.

Briles, D.E., Ades, E., Paton, J.C., Sampson, J.S., Carlone, G.M., Huebner, R.C., Virolainen, A., Swiatlo, E. and Hollingshead, S.K. (2000) Intranasal immunization of mice with a mixture of the pneumococcal proteins PsaA and PspA is highly protective against nasopharyngeal carriage of *Streptococcus pneumoniae*. *Infect. Immun.* **68**: 796-800.

Briles, D.E., King, J.D., Gray, M.A., McDaniel, L.S., Swiatlo, E. and Benton, K.A. (1996) PspA, a protection-eliciting pneumococcal protein: immunogenicity of isolated native PspA in mice. *Vaccine* **14**: 858-867.

Briles, D.E., Novak, L., Hotomi, M., van Ginkel, F.W. and King, J. (2005) Nasal colonization with Streptococcus pneumoniae includes subpopulations of surface and invasive pneumococci. *Infect. Immun.* **73**: 6945-6951.

Briles, D.E., Tart, R.C., Swiatlo, E., Dillard, J.P., Smith, P., Benton, K.A., Ralph,
B.A., Brooks-Walter, A., Crain, M.J., Hollingshead, S.K. and McDaniel, L.S. (1998)
Pneumococcal diversity: considerations for new vaccine strategies with emphasis on
pneumococcal surface protein A (PspA). *Clin. Microbiol. Rev.* 11: 645-657.

Briles, D.E., Yother, J. and McDaniel, L.S. (1988) Role of pneumococcal surface protein A in the virulence of *Streptococcus pneumoniae*. *Rev. Infect. Dis.* **10**(Suppl 2): S372-S374.

Brock, S.C., McGraw, P.A., Wright, P.F. and Crowe, J.E.J. (2002) The human polymeric immunoglobulin receptor facilitates invasion of epithelial cells by *Streptococcus pneumoniae* in a strain-specific and cell type-specific manner. *Infect. Immun.* **70**: 5091-5095.

Brooks-Walter, A., Briles, D.E. and Hollingshead, S.K. (1999) The *pspC* gene of *Streptococcus pneumoniae* encodes a polymorphic protein, PspC, which elicits cross-reactive antibodies to PspA and provides immunity to pneumococcal bacteremia. *Infect. Immun.* **67**: 6533-6542.

Brown, J.S., Gilliland, S.M. and Holden, D.W. (2001a) A *Streptococcus pneumoniae* pathogenicity island encoding an ABC transporter involved in iron uptake and virulence. *Mol. Microbiol.* **40**: 572-585.

Brown, J.S., Ogunniyi, A.D., Woodrow, M.C., Holden, D.W. and Paton, J.C. (2001b) Immunization with components of two iron uptake ABC transporters protects mice against systemic *Streptococcus pneumoniae* infection. *Infect. Immun.* **69**: 6702-6706.

Burrell, M., Mackintosh, M.E. and Taylor, C.E. (1986) Isolation of *Streptococcus pneumoniae* from the respiratory tract of horses. *Equine Vet. J.* **18**: 183-186.

Butler, J.C. and Cetron, M. (1999) Pneumococcal drug resistance: the new 'special enemy of old age'. *Clin. Infect. Dis.* 28: 730-735.

Camara, M., Boulnois, G.J., Andrew, P.W. and Mitchell, T.J. (1994) A neuraminidase from *Streptococcus pneumoniae* has the features of a surface protein. *Infect. Immun.* **62**: 3688-3695.

Campbell, N.A. (1996) *Biology*. Menlo Park, California, The Benjamin/Cummings Publishing Company.

Carlsen, B.D., Kawana, M., Kawana, C., Tomasz, A. and Giebink, G.S. (1992) Role of the bacterial cell wall in middle ear inflammation caused by *Streptococcus pneumoniae*. *Infect. Immun.* **60**: 2850-2854.

Catterall, J.R. (1999) Streptococcus pneumoniae. Thorax 54: 929-937.

Centers for Disease Control and Prevention (1997) Prevention of Pneumococcal Disease: Recommendations of the Advisory Committee on Immunization Practices (ACIP) [online]. *MMWR* **46**: 1-24.

Centers for Disease Control and Prevention (2005) *Streptococcus pneumoniae* disease [online]. http://www.cdc.gov/ncidod/dbmd/diseaseinfo/streppneum_t.htm (March, 2007).

Chapman, P.S., Green, C., Main, J.P., Taylor, P.M., Cunningham, F.M., Cook, A.J. and Marr, C.M. (2000) Retrospective study of the relationships between age, inflammation and the isolation of bacteria from the lower respiratory tract of thoroughbred horses. *Vet. Rec.* **146**: 91-95.

Choo, S., Seymour, L., Morris, R., Quataert, S., Lockhart, S., Cartwrigh, K. and Finn, A. (2000) Immunogenicity and reactogenicity of a pneumococcal conjugate vaccine administered combined with a *Haemophilus influenzae* type b conjugate vaccine in United Kingdom infants. *Pediatr. Infect. Dis. J.* **19**: 854-862.

Claverys, J.P., Grossiord, B. and Alloing, G. (2000) Is the Ami-AliA/B oligopeptide permease of *Streptococcus pneumoniae* involved in sensing environmental conditions? *Res. Microbiol.* **151**: 457-463.

Claverys, J.P. and Havarstein, L.S. (2002) Extracellular-peptide control of competence for genetic transformation in *Streptococcus pneumoniae*. *Front. Biosci.* **1**: d1798-1814.

Cockeran, R., Durandt, C., Feldman, C., Mitchell, T.J. and R, A. (2002) Pneumolysin activates the synthesis and release of interleukin-8 by human neutrophils *in vitro*. *J. Infect. Dis.* **186**: 562-565.

Cody, C.W., Prasher, D.C., Westler, W.M., Prendergast, F.G. and Ward, W.W. (1993) Chemical structure of the hexapeptide chromophore of the Aequorea green-fluorescent protein. *Biochemistry* **32**: 1212-1218.

Converse, G.M. III and Dillon, H.C. Jr (1977) Epidemiological studies of Streptococcus pneumoniae in infants: methods of isolating pneumococci. *J. Clin. Microbiol.* **5**: 293-296.

Corish, P. and Tyler-Smith, C. (1999) Attenuation of green fluorescent protein half-life in mammalian cells. *Protein Eng.* **12**: 1035-1040.

Crain, M.J., Waltman, W.D. 2nd, Turner, J.S., Yother, J., Talkington, D.F., McDaniel, L.S., Gray, B.M. and Briles, D.E. (1990) Pneumococcal surface protein A (PspA) is serologically highly variable and is expressed by all clinically important capsular serotypes of *Streptococcus pneumoniae*. *Infect. Immun.* **58**: 3293-3299.

Crook, D., Brueggemann, A.B., Sleeman, K. and Peto, T.E.A. (2004) *Pneumococcal carriage*. Washington DC, ASM Press.

Cundell, D.R., Gerald, N.P., Gerald, C., Idanpaan-Heikkila, I. and Tuomanen, E.I. (1995a) *Streptococcus pneumoniae* anchor to activated human cells by the receptor for platelet-activating factor. *Nature* **377**: 435-438.

Cundell, D.R. and Tuomanen, E.I. (1994) Receptor specificity of adherence of *Streptococcus pneumoniae* to human type-II pneumocytes and vascular enothelial cells in vitro. *Microb. Pathog.* **17**: 361-374.

Cundell, D.R., Weiser, J.N., Shen, J., Young, A. and Tuomanen, E.I. (1995b) Relationship between colonial morphology and adherence of *Streptococcus pneumoniae*. *Infect. Immun.* **63**: 757-761.

Cunha, B.A., Qadri, S.M., Ueno, Y., Walters, E.A. and Domenico, P. (1997) Antibacterial activity of trovafloxacin against nosocomial Gram-positive and Gramnegative isolates. *J. Antimicrob. Chemother.* **39**(Suppl. B): 29-34. Dave, S., Brooks-Walter, A., Pangburn, M.K. and McDaniel, L.S. (2001) PspC, a pneumococcal surface protein, binds human factor H. *Infect. Immun.* **69**: 3435-3437.

Davidson, M., Parkinson, A.J., Bulkow, L.R., Fitzgerald, M.A., Peters, H.V. and Parks, D.J. (1994) The epidemiology of invasive pneumococcal disease in Alaska, 1986-1990--ethnic differences and opportunities for prevention. *J. Infect. Dis.* **170**: 368-376.

Dintilhac, A., Alloing, G., Granadel, C. and Claverys, J.P. (1997) Competence and virulence of *Streptococcus pneumoniae:* Adc and PsaA mutants exhibit a requirement for Zn and Mn resulting from inactivation of putative ABC metal permeases. *Mol. Microbiol.* 25: 727-739.

Doern, G.V., Heilmann, K.P., Huynh, H.K., Rhomberg, P.R., Coffman, S.L. and Brueggemann, A.B. (2001) Antimicrobial resistance among clinical isolates of *Streptococcus pneumoniae* in the United States during 1999-2000, including a comparison of resistance rates since 1994-1995. *Antimicrob. Agents Chemother.* **45**: 1721-1729.

Dowell, S.F., Butler, J.C., Giebink, G.S., Jacobs, M.R., Jernigan, D., Musher, D.M., Rakowsky, A. and Schwartz, B. (1999) Acute otitis media: management and surveillance in an era of pneumococcal resistance--a report from the Drug-resistant *Streptococcus pneumoniae* Therapeutic Working Group. *Pediatr. Infect. Dis. J.* **18**: 1-9.

Feldman, C., Mitchell, T.J., Andrew, P.W., Boulnois, G.J., Read, R.C., Todd, H.C., Cole, P.J. and Wilson, R. (1990) The effect of *Streptococcus pneumoniae* pneumolysin on human respiratory epithelium in vitro. *Microb. Pathog.* **9**: 275-284.

Fraenkel, A. (1886) Weitere Beitrage zur Lehre von den Mikrococcen der genuinen fibrinosen Pneumoniae. *Zeitschrift für Klinische Medicin* **11**: 437-458.

Francis, K.P., Joh, D., Bellinger-Kawahara, C., Hawkinson, M.J., Purchio, T.F. and Contag, P.R. (2000) Monitoring bioluminescent *Staphylococcus aureus* infections in living mice using a novel luxABCDE construct. *Infect. Immun.* **68**: 3594-3600.

Garcia, E., Llull, D., Munoz, R., Mollerach, M. and Lopez, R. (2000) Current trends in

capsular polysaccharide biosynthesis of *Streptococcus pneumoniae*. *Res. Microbiol.* **151**: 429-435.

Garcia, J.L., Sanchez-Beato, A.R., Medrano, F.J. and Lopez, R. (1998) Versatility of choline-binding domain. *Microb. Drug Resist.* **4**: 25-36.

Garcia, P., Garcia, J.L., Garcia, E. and Lopez, R. (1986) Nucleotide sequence and expression of the pneumococcal autolysin gene from its own promoter in *Escherichia coli*. *Gene* **43**: 265-272.

Garcia, P., Gonzalez, M.P., Garcia, E., Lopez, R. and Garcia, J. (1999a) LytB, a novel pneumococcal murein hydrolase essential for cell separation. *Mol. Microbiol.* **31**: 1275-1281.

Garcia, P., Paz Gonzalez, M., Garcia, E., Garcia, J.L. and Lopez, R. (1999b) The molecular characterization of the first autolytic lysozyme of *Streptococcus pneumoniae* reveals evolutionary mobile domains. *Mol. Microbiol.* **33**: 128-138.

Giebink, G.S. (1999) Otitis media: the chinchilla model. Microb. Drug Resist. 5: 57-72.

Gosink, K.K., Mann, E.R., Guglielmo, C., Tuomanen, E.I. and Masure, H.R. (2000) Role of novel choline binding proteins in virulence of *Streptococcus pneumoniae*. *Infect. Immun.* **68**: 5690-5695.

Gram, C. (1884) Ueber die isolirte Farbung der Schizomyceten in SchnittÄund Trockenpraparaten. *Fortschr. Med.* 2: 185-189.

Gray, B.M., Converse, G.M. III and Dillon, H.C.J. (1980) Epidemiologic studies of *Streptococcus pneumoniae* in infants: acquisition, carriage, and infection during the first 24 months of life. *J. Infect. Dis.* **142**: 923-933.

Griffith, F. (1928) The significance of pneumococcal types. J. Hyg. 27: 113–159.

Guerzoni, M.E., Lanciotti, R. and Cocconcelli, S. (2001) Alteration in cellular fatty acid

composition as a response to salt, acid, oxidative and thermal stresses in *Lactobacillus helveticus*. *Microbiology* **147**: 2255-2264.

Hakansson, A., Roche, H., Mirza, S., McDaniel, L.S., Brooks-Walter, A. and Briles, D.E. (2001) Characterization of binding of human lactoferrin to pneumococcal surface protein A. *Infect. Immun.* **69**: 3372-3381.

Hammerschmidt, S., Bethe, G., Remane, P.H. and Chhatwal, G.S. (1999) Identification of Pneumococcal Surface Protein A as a lactoferrin-binding protein of *Streptococcus pneumoniae*. *Infect. Immun.* **67**: 1683-1687.

Hammerschmidt, S., Talay, S.R., Brandtzaeg, P. and Chhatwal, G.S. (1997) *SpsA*, a novel pneumococcal surface protein with specific binding to secretory immunoglobulin A and secretory component. *Mol. Microbiol.* **25**: 1113-1124.

Hava, D.L. and Camilli, A. (2002) Large-scale identification of serotype 4 *Streptococcus pneumoniae* virulence factors. *Mol. Microbiol.* **45**: 1389-1405.

Havarstein, L.S., Coomaraswamy, G. and Morrison, D.A. (1995) An unmodified heptadecapeptide pheromone induces competence for genetic transformation in *Streptococcus pneumoniae*. *Proc. Natl. Acad. Sci. USA*. **92**: 11140-11144.

Heidelberger, M. (1927) Immunologically specific polysaccharides. *Chemical Reviews* **3**: 403-423.

Heidelberger, M. and Avery, O.T. (1923) The soluble specific substance of pneumococcus. J. Exp. Med. 38: 73-79.

Heim, R., Prasher, D.C. and Tsien, R.Y. (1994) Wavelength mutations and posttranslational autoxidation of green fluorescent protein. *Proc. Natl. Acad. Sci. USA* **91**: 12501-12504.

Hendriksen, W.T., Silva, N., Bootsma, H.J., Blue, C.E., Paterson, G.K., Kerr, A.R., de Jong, A., Kuipers, O.P., Hermans, P.W. and Mitchell, T.J. (2007) Regulation of gene

expression in *Streptococcus pneumoniae* by response regulator 09 is strain dependent. *J. Bacteriol.* **189**: 1382-1389.

Hill, P.J., Rees, C.E., Winson, M.K. and Stewart, G.S. (1993) The application of lux genes. *Biotechnol. Appl. Biochem.* 17: 3-14.

Holmes, A.R., McNab, R., Millsap, K.W., Rohde, M., Hammerschmidt, S., Mawdsley, J.L. and Jenkinson, H.F. (2001) The *pavA* gene of *Streptococcus pneumoniae* encodes a fibronectin-binding protein that is essential for virulence. *Mol. Microbiol.* **41**: 1395-1408.

Houldsworth, S., Andrew, P.W. and Mitchell, T.J. (1994) Pneumolysin stimulates production of tumor necrosis factor alpha and interleukin-1 beta by human mononuclear phagocytes. *Infect. Immun.* **62**: 1501-1503.

Hussain, M., Melegaro, A., Pebody, R.G., George, R., Edmunds, W.J., Talukdar, R., Martin, S.A., Efstratiou, A. and Miller, E. (2005) A longitudinal household study of *Streptococcus pneumoniae* nasopharyngeal carriage in a UK setting. *Epidemiol. Infect.* **133**: 891-898.

Iannelli, F., Oggioni, M.R. and Pozzi, G. (2002) Allelic variation in the highly polymorphic locus pspC of *Streptococcus pneumoniae*. *Gene* **284**: 63-71.

Ibrahim, Y.M., Kerr, A.R., McCluskey, J. and Mitchell, T.J. (2004) Role of HtrA in the virulence and competence of *Streptococcus pneumoniae*. *Infect. Immun.* **72**: 3584-3591.

Ibrahim, Y.M., Kerr, A.R., Silva, N.A. and Mitchell, T.J. (2005) Contribution of the ATP-dependent protease ClpCP to the autolysis and virulence of *Streptococcus pneumoniae*. *Infect. Immun.* **73**: 730-740.

Ishizuka, S., Yamaya, M., Suzuki, T., Nakayama, K., Kamanaka, M., Ida, S., Sekizawa, K. and Sasaki, H. (2001) Acid exposure stimulates the adherence of *Streptococcus pneumoniae* to cultured human airway epithelial cells: effects on platelet-activating factor receptor expression. *Am. J. Respir. Cell. Mol. Biol.* **24**: 459-468.

Jakubovics, N.S. and Jenkinson, H.F. (2001) Out of the iron age: new insights into the critical role of manganese homeostasis in bacteria. *Microbiology* **147**: 1709-1718.

Janulczyk, R., Iannelli, F., Sjoholm, A.G., Pozzi, G. and Bjorck, L. (2000) Hic, a novel surface protein of *streptococcus pneumoniae* that interferes with complement function. *J. Biol. Chem.* **275**: 37257-37263.

Jarva, H., Jokiranta, T.S., Wurzner, R. and Meri, S. (2003) Complement resistance mechanisms of streptococci. *Mol. Immuno.* **40**: 95-107.

Johnson, S.E., Dykes, J.K., Jue, D.L., Sampson, J.S., Carlone, G.M. and Ades, E.W. (2002) Inhibition of pneumococcal carriage in mice by subcutaneous immunization with peptides from the common surface protein pneumococcal surface adhesin a. *J. Infect. Dis.* **185**: 489-496.

Johnston, J.W., Myers, L.E., Ochs, M.M., Benjamin, W.H.J., Briles, D.E. and Hollingshead, S.K. (2004) Lipoprotein PsaA in virulence of *Streptococcus pneumoniae*: surface accessibility and role in protection from superoxide. *Infect. Immun.* **72**: 5858-5867.

Jomaa, M., Terry, S., Hale, C., Jones, C., Dougan, G. and Brown, J. (2006) Immunization with the iron uptake ABC transporter proteins PiaA and PiuA prevents respiratory infection with *Streptococcus pneumoniae*. *Vaccine* **24**: 5133-5139.

Jomaa, M., Yuste, J., Paton, J.C., Jones, C., Dougan, G. and Brown, J.S. (2005) Antibodies to the iron uptake ABC transporter lipoproteins PiaA and PiuA promote opsonophagocytosis of *Streptococcus pneumoniae*. *Infect. Immun.* **73**: 6852-6859.

Kadioglu, A., Sharpe, J.A., Lazou, I., Svanborg, C., Ockleford, C., Mitchell, T.J. and Andrew, P.W. (2001) Use of green fluorescent protein in visualisation of pneumococcal invasion of broncho-epithelial cells *in vivo*. *FEMS Microbiol*. *Lett.* **194**: 105-110.

Kadioglu, A., Taylor, S., Iannelli, F., Pozzi, G., Mitchell, T.J. and Andrew, P.W. (2002) Upper and lower respiratory tract infection by *Streptococcus pneumoniae* is

affected by pneumolysin deficiency and differences in capsule type. *Infect. Immun.* **70**: 2886-2890.

Kalin, M., Ortqvist, A., Almela, M., Aufwerber, E., Dwyer, R., Henriques, B., Jorup, C., Julander, I., Marrie, T.J., Mufson, M.A., Riquelme, R., Thalme, A., Torres, A. and Woodhead, M.A. (2000) Prospective study of prognostic factors in community-acquired bacteremic pneumococcal disease in 5 countries. *J. Infect. Dis.* **182**: 840-847.

Katzenellenbogen, E. and Jennings, H.J. (1983) Structural determination of the capsular polysaccharide of *Streptococcus pneumoniae* type 19A (57). *Carbohydr. Res.* **124**: 235-245.

Kausmally, L., Johnsborg, O., Lunde, M., Knutsen, E. and Havarstein, L.S. (2005) Choline-binding protein D (CbpD) in *Streptococcus pneumoniae* is essential for competence-induced cell lysis. *J. Bacteriol.* **187**: 4338-4345.

Kawamuru, Y., Hou, X.-G., Sultana, F., Miura, H. and Ezaki, T. (1995) Determination of 16S rRNA sequences of *Streptococcus mitis* and *Streptococcus gordonii* and phylogenetic relationships among members of the genus *Streptococcus*. *Int. J. Syst. Bacteriol.* **45**: 406-408.

Kayaoglu, G. and Orstavik, D. (2004) Virulence factors of *Enterococcus faecalis*: relationship to endodontic disease. *Crit. Rev. Oral Biol. Med.* **15**: 308-320.

Kim, J.O., Romero-Steiner, S., Sorensen, U.B., Blom, J., Carvalho, M., Barnard, S.,
Carlone, G. and Weiser, J.N. (1999) Relationship between cell surface carbohydrates and intrastrain variation on opsonophagocytosis of *Streptococcus pneumoniae*. *Infect. Immun.*67: 2327-2333.

Kim, J.O. and Weiser, J.N. (1998) Association of intrastrain phase variation in quantity of capsular polysaccharide and teichoic acid with the virulence of *Streptococcus pneumoniae*. *J. Infect. Dis.* **177**: 368-377.

King, S.J., Hippe, K.R., Gould, J.M., Bae, D., Peterson, S., Cline, R.T., Fasching, C.,

Janoff, E.N. and Weiser, J.N. (2004) Phase variable desialylation of host proteins that bind to *Streptococcus pneumoniae in vivo* and protect the airway. *Mol. Microbiol.* **54**: 159-171.

Kirkham, L.-A.S., Jefferies, J.M.C., Kerr, A.R., Jing, Y., Clarke, S.C., Smith, A. and Mitchell, T.J. (2006) Identification of invasive serotype 1 pneumococcal isolates that express nonhemolytic pneumolysin *J. Clin. Microbiol.* **44**: 151-159.

Koedel, U., Scheld, W.M. and Pfister, H.-W. (2002) Pathogenesis and pathophysiology of pneumococcal meningitis. *Lancet* **2**: 721-736.

Krontz, D.P. and Strausbaugh, L.J. (1980) Effect of meningitis and probenecid on the penetration of vancomycin into cerebrospinal fluid in rabbits. *Antimicrob. Agents Chemother.* **18**: 882-886.

Laemmli, U.K. (1970) Cleavage of structural proteins during the assembly of the head of bacteriophage T4. *Nature* **227**: 680-685

Lanie, J.A., Ng, W.L., Kazmierczak, K.M., Andrzejewski, T.M., Davidsen, T.M., Wayne, K.J., Tettelin, H., Glass, J.I. and Winkler, M.E. (2007) Genome sequence of Avery's virulent serotype 2 strain D39 of *Streptococcus pneumoniae* and comparison with that of unencapsulated laboratory strain R6. *J Bacteriol.* **189**: 38-51.

Lau, G.W., Haataja, S., Lonetto, M., Kensit, S.E., Marra, A., Bryant, A.P., McDevitt,
D., Morrison, D.A. and Holden, D.W. (2001) A functional genomic analysis of type 3
Streptococcus pneumoniae virulence. Mol. Microbiol. 40: 555-571.

Le Gouill, C., Parent, J.L., Rola-Pleszczynski, M. and Stankova, J. (1997) Structural and functional requirements for agonist-induced internalization of the human platelet-activating factor receptor. *J. Biol. Chem.* **272**: 21289-21295.

Lee, S.H. and Camilli, A. (2000) Novel approaches to monitor bacterial gene expression in infected tissue and host. *Curr. Opin. Microbiol.* **3**: 97-101.

LeMessurier, K.S. (2002) Characterisation of a nasopharyngeal carriage model of *Streptococcus pneumoniae* D39 in outbred CD1 mice. In *Department of Molecular Biosciences*, Adelaide, University of Adelaide.

Levine, O.S., Farley, M., Harrison, L.H., Lefkowitz, L., McGeer, A. and Schwartz, B. (1999) Risk factors for invasive pneumococcal disease in children: a population-based case-control study in North America. *Pediatrics* **103**: E28.

Lexau, C.A., Lynfield, R., Danila, R., Pilishvili, T., Farley, M.M., Schaffner, W., Reingold, A., Bennett, N.M., Hadler, J., Cieslak, P.R. and Whitney, C.G. (2005) Changing epidemiology of invasive pneumococcal disease among older adults in the era of pediatric pneumococcal conjugate vaccine. *JAMA* **294**: 2043-2051.

Lion, T. (2001) Current recommendations for positive controls in RT-PCR assays. *Leukemia* **15**: 1033-1037.

Livak, K.J. and Schmittgen, T.D. (2001) Analysis of relative gene expression data using real-time quantitative PCR and the $2^{-\Delta\Delta C}_{T}$ method. *Methods* **25**: 402-408.

Llull, D., Lopez, R. and Garcia, E. (2001) Genetic bases and medical relevance of capsular polysaccharide biosynthesis in pathogenic streptococci. *Curr. Mol. Med.* 1: 475-791.

Lock, R.A., Zhang, Q.Y., Berry, A.M. and Paton, J.C. (1996) Sequence variation in the *Streptococcus pneumoniae* pneumolysin gene affecting haemolytic activity and electrophoretic mobility of the toxin. *Microb. Pathog.* **21**: 71-83.

Mackenzie, G., Carapetis, J., Morris, P. and Leach, A. (2005) Current issues regarding the use of pneumococcal conjugate and polysaccharide vaccines in Australian children. *J. Paediatr. Child Health* **41**: 201-208.

MacLeod, C.M. and Kraus, M.R. (1950) Relation of virulence of pneumococcal strains for mice to the quantity of capsular polysaccharide formed in vitro. *J. Exp. Med.* **92**: 1-9.

Malfroot, A., Verhaegen, J., Dubru, J.M., Van Kerschaver, E. and Leyman, S. (2004) A cross-sectional survey of the prevalence of *Streptococcus pneumoniae* nasopharyngeal carriage in Belgian infants attending day care centres. *Clin. Microbiol. Infect.* **10**: 797-803.

Malley, R., Philipp, H., Morse, S.C., Cieslewicz, M.J., Lipsitch, M., Thompson, C.M., Kurt-Jones, E., Paton, J.C., Wessels, M.R. and Golenbock, D.T. (2003) Recognition of pneumolysin by Toll-like receptor 4 confers resistance to pneumococcal infection. *Proc. Natl. Acad. Sci. USA* **100**: 1966-1971.

Marra, A., Asundi, J., Bartilson, M., Lawson, S., Fang, F., Christine, J., Wiesner, C., Brigham, D., Schneider, W.P. and Hromockyj, A.E. (2002a) Differential fluorescence induction analysis of *Streptococcus pneumoniae* identifies genes involved in pathogenesis. *Infect. Immun.* **70**: 1422-1433.

Marra, A., Lawson, S., Asundi, J.S., Brigham, D. and Hromockyj, A.E. (2002b) In vivo characterization of the *psa* genes from *Streptococcus pneumoniae* in multiple models of infection. *Microbiology* **148**: 1483-1491.

Mbelle, N., Huebner, R.E., Wasas, A.D., Kimura, A., Chang, I. and Klugman, K.P. (1999) Immunogenicity and impact on nasopharyngeal carriage of a nonavalent pneumococcal conjugate vaccine. *J. Infect. Dis.* **180**: 1171-1176.

McAllister, L.J., Tseng, H.J., Ogunniyi, A.D., Jennings, M.P., McEwan, A.G. and Paton, J.C. (2004) Molecular analysis of the psa permease complex of *Streptococcus pneumoniae*. *Mol. Microbiol.* **53**: 889-901.

McCarty, M. and Avery, O.T. (1946) Studies on the chemical nature of the substance inducing transformation of pneumococcal tyles: II. Effect of desoxyribonuclease on the biological activity of the transforming substance. *J. Exp. Med.* **83**: 89-96.

McCluskey, J., Hinds, J., Husain, S., Witney, S. and Mitchell, T.J. (2004) A twocomponent system that controls the expression of pneumococcal surface antigen A (PsaA) and regulates virulence and resistance to oxidative stress in *Streptococcus pneumoniae*. *Mol. Microbiol.* **51**: 1661-1675. McCool, T.L., Cate, T.R., Moy, G. and Weiser, J.N. (2002) The immune response to pneumococcal proteins during experimental human carriage. *J. Exp. Med.* **195**: 359-365.

McCullers, J.A. and Tuomanen, E.I. (2001) Molecular pathogenesis of pneumococcal pneumonia. *Front. Biosci.* 1: D877-889.

McDaniel, L.S., Yother, J., Vijayakumar, M., McGarry, L., Guild, W.R. and Briles, D.E. (1987) Use of insertional inactivation to facilitate the studies of biological properties of pneumococcal surface protein A (PspA). *J. Exp. Med.* **165**: 381-394.

McKesser, S. (2003) The characterisation of phase variation and a novel fimbrial protein in *S. pneumoniae*. In *School of Pharmaceutical, Molecular and Biomedical Sciences*, Adelaide, University of South Australia.

Meighen, E.A. (1991) Molecular biology of bacterial bioluminescence. *Microbiol. Rev.* 55: 123-142.

Meyer, J.C., Koterba, A., Lester, G. and Purich, B.L. (1992) Bacteraemia and pneumonia in a neonatal foal caused by *Streptococcus pneumoniae* type 3. *Equine Vet. J.* 24: 407-410.

Mitchell, T.J., Andrew, P.W., Saunders, F.K., Smith, A.N. and Boulnois, G.J. (1991) Complement activation and antibody binding by pneumolysin via a region of the toxin homologous to a human acute-phase protein. *Mol. Microbiol.* **5**: 1883-1888.

Morona, J.K., Morona, R., Miller, D.C. and Paton, J.C. (2003) Mutational analysis of the carboxy-terminal (YGX)4 repeat domain of CpsD, an autophosphorylating tyrosine kinase required for capsule biosynthesis in *Streptococcus pneumoniae*. *J. Bacteriol.* **185**: 3009-3019.

Morona, J.K., Morona, R. and Paton, J.C. (1999) Analysis of the 5' portion of the type 19A capsule locus identifies two classes of cpsC, cpsD, and cpsE genes in *Streptococcus pneumoniae*. *J. Bacteriol.* **181**: 3599-3605.

Morona, J.K., Paton, J.C., Miller, D.C. and Morona, R. (2000) Tyrosine phosphorylation of CpsD negatively regulates capsular polysaccharide biosynthesis in *Streptococcus pneumoniae*. *Mol. Microbiol.* **35**: 1431-1442.

Mortier-Barriere, I., de Saizieu, A., Claverys, J.P. and Martin, B. (1998) Competencespecific induction of recA is required for full recombination proficiency during transformation in *Streptococcus pneumoniae*. *Mol. Microbiol*. **27**(1): 159-170.

Mostov, K.E. (1994) Transepithelial transport of immunoglobulins. *Annu. Rev. Immunol.*12: 63-84.

Mulholland, K. (1999) Strategies for the control of pneumococcal diseases. *Vaccine* 17: S79-S84.

Musher, D.M. (1992) Infections caused by *Streptococcus pneumoniae*: clinical spectrum, pathogenesis, immunity, and treatment. *Clin. Infect. Dis.* **14**: 801-807.

Musher, D.M., Luchi, M.J., Watson, D.A., Hamilton, R. and Baughn, R.E. (1990) Pneumococcal polysaccharide vaccine in young adults and older bronchitics: determination of IgG responses by ELISA and the effect of adsorption of serum with non-type-specific cell wall polysaccharide. *J. Infect. Dis.* **161**: 728-735.

Musher, D.M., Phan, H.M. and Baughn, R.E. (2001) Protection against bacteremic pneumococcal infection by antibody to pneumolysin. *J. Infect. Dis.* **183**: 827-230.

Nabors, G.S., Braun, P.A., Herrmann, D.J., Heise, M.L., Pyle, D.J., Gravenstein, S., Schilling, M., Ferguson, L.M., Hollingshead, S.K., Briles, D.E. and Becker, R.S. (2000) Immunization of healthy adults with a single recombinant pneumococcal surface protein A (PspA) variant stimulates broadly cross-reactive antibodies to heterologous PspA molecules. *Vaccine* **18**: 1743-1754.

Neidhardt, F.C., Curtiss III, R., Ingraham, J.L., Lin, E.C.C., Low, K.B., Magasanik, B., Reznikoff, W.S., Riley, M., Schaechter, M. and Umbarger, H.E. (1996) *Escherichia*
coli and Salmonella: cellular and molecular biology. Washington, DC, ASM Press.

Neufeld, F. and Schnitzler, R. (1928) *Pneumokokken Handbuch der Pathogenen Mikroorganismen, 3rd ed.* Berlin, Germany, Gustav Fischer.

Nuorti, J.P., Butler, J.C., Farley, M.M., Harrison, L.H., McGeer, A., Kolczak, M.S. and Breiman, R.F. (2000) Cigarette smoking and invasive pneumococcal disease. Active Bacterial Core Surveillance Team. *N. Engl. J. Med.* **342**: 681-689.

O'Donoghue, J.M., Schweid, A.I. and Beaty, H.N. (1974) Experimental pneumococcal meningitis I: a rabbit model. *Proc. Soc. Exp. Biol. Med.* **146**: 571-574.

Ogunniyi, A.D., Giammarinaro, P. and Paton, J.C. (2002) The genes encoding virulence-associated proteins and the capsule of *Streptococcus pneumoniae* are upregulated and differentially expressed *in vivo. Microbiology* **148**: 2045-2053.

Ogunniyi, A.D., Lemessurier, K.S., Graham, R.M., Watt, J.M., Briles, D.E., Stroeher, U.H. and Paton, J.C. (2007) Contribution of Pneumolysin, PspA and PspC (CbpA) to Pathogenicity of Streptococcus pneumoniae D39 in a Mouse Model. *Infect. Immun.* **75**: 1843-1851.

Ogunniyi, A.D., Woodrow, M.C., Poolman, J.T. and Paton, J.C. (2001) Protection against *Streptococcus pneumoniae* elicited by immunization with pneumolysin and CbpA. *Infect. Immun.* **69**: 5997-6003.

Orihuela, C.J., Gao, G., Francis, K.P., Yu, J. and Tuomanen, E.I. (2004a) Tissuespecific contributions of pneumococcal virulence factors to pathogenesis. *J. Infect. Dis.* **190**: 1661-1669.

Orihuela, C.J., Gao, G., McGee, M., Yu, J., Francis, K.P. and Tuomanen, E. (2003) Organ-specific models of *Streptococcus pneumoniae* disease. *Scand. J. Infect. Dis.* 35: 647-652.

Orihuela, C.J., Radin, J.N., Sublett, J.E., Gao, G., Kaushal, D. and Tuomanen, E.I.

(2004b) Microarray analysis of pneumococcal gene expression during invasive disease. *Infect. Immun.* **72**: 5582-5596.

Overweg, K., Pericone, C.D., Verhoef, G.G., Weiser, J.N., Meiring, H.D., De Jong, A.P., De Groot, R. and Hermans, P.W. (2000) Differential protein expression in phenotypic variants of *Streptococcus pneumoniae*. *Infect. Immun.* **68**: 4604-4610.

Pallares, R., Linares, J., Vadillo, M., Cabellos, C., Manresa, F., Viladrich, P.F., Martin, R. and Gudiol, F. (1995) Resistance to penicillin and cephalosporin and mortality from severe pneumococcal pneumonia in Barcelona. *Spain. N. Engl. J. Med.* **333**: 474-486.

Parsons, H.K. and Dockrell, D.H. (2002) The burden of invasive pneumococcal disease and the potential for reduction by immunisation. *Int. J. Antimicrob. Agents* **19**(2): 85-93.

Pasteur, L. and Chamberland, M.M. (1881) Note sur la maladie nouvelle provoquée par la salive d'un enfant mort de la rage. *Bull. Acad. Méd (Paris)* **10**: 94-103.

Paton, J.C. (1996) The contribution of pneumolysin to the pathogenicity of *Streptococcus pneumoniae*. *Trends Microbiol*. **4**: 103-106.

Paton, J.C., Andrew, P.W., Boulnois, G.J. and Mitchell, T.J. (1993) Molecular analysis of the pathogenicity of *Streptococcus pneumoniae*: the role of pneumococcal proteins. *Annu. Rev. Microbiol.* **47**: 89-115.

Paton, J.C., Lock, R.A. and Hansman, D.J. (1983) Effect of immunization with pneumolysin on survival time of mice challenged with *Streptococcus pneumoniae*. *Infect. Immun.* **40**: 548-552.

Paton, J.C. and Morona, J.K. (2000) *Streptococcus pneumoniae* Capsular polysaccharide. In *Gram-Positive Pathogens*. V.A. Fischetti, Novick, R.P., Ferretti, J.J., Portnoy, D.A. and Rood, J.I. (eds). Washington, D.C., ASM Press, pp. 201-213.

Paton, J.C., Rowan-Kelly, B. and Ferrante, A. (1984) Activation of human complement by the pneumococcal toxin pneumolysin. *Infect. Immun.* **43**: 1085-1087. **Pearce, B.J., Naughton, A.M., Campbell, E.A. and Masure, H.R.** (1995) The rec locus, a competence-induced operon in *Streptococcus pneumoniae*. *J. Bacteriol.* **177**: 86-93.

Pericone, C.D., Overweg, K., Hermans, P.W. and Weiser, J.N. (2000) Inhibitory and bactericidal effects of hydrogen peroxide production by *Streptococcus pneumoniae* on other inhabitants of the upper respiratory tract. *Infect. Immun.* **68**: 3990-3997.

Pericone, C.D., Park, S., Imlay, J.A. and Weiser, J.N. (2003) Factors contributing to hydrogen peroxide resistance in *Streptococcus pneumoniae* include pyruvate oxidase (SpxB) and avoidance of the toxic effects of the fenton reaction. *J. Bacteriol.* **185**: 6815-6825.

Poehling, K.A., Talbot, T.R., Griffin, M.R., Craig, A.S., Whitney, C.G., Zell, E., Lexau, C.A., Thomas, A.R., Harrison, L.H., Reingold, A.L., Hadler, J.L., Farley, M.M., Anderson, B.J. and Schaffner, W. (2006) Invasive pneumococcal disease among infants before and after introduction of pneumococcal conjugate vaccine. *JAMA* 295: 1668-1674.

Polissi, A., Pontiggia, A., Feger, G., Altieri, M., Mottl, H., Ferrari, L. and Simon, D. (1998) Large-scale identification of virulence genes from *Streptococcus pneumoniae*. *Infect. Immun.* **66**: 5620-5629.

Qazi, S.N., Counil, E., Morrissey, J., Rees, C.E., Cockayne, A., Winzer, K., Chan, W.C., Williams, P. and Hill, P.J. (2001) *agr* expression precedes escape of internalized *Staphylococcus aureus* from the host endosome. *Infect. Immun.* **69**: 7074-7082.

Rayner, C.F.J., Jackson, A.D., Rutman, A., Dewar, A., Mitchell, T.J., Andrew, P.W., Cole, P.K. and Wilson, R. (1995) Interaction of pneumolysin-sufficient and -deficient isogenic variants of *Streptococcus pneumoniae* with human respiratory musoca. *Infect. Immun.* **63**: 442-447.

Ren, B., Szalai, A.J., Hollingshead, S.K. and Briles, D.E. (2004) Effects of PspA and antibodies to PspA on activation and deposition of complement on the pneumococcal

surface. Infect. Immun. 72: 114-122.

Ren, B., Szalai, A.J., Thomas, O., Hollingshead, S.K. and Briles, D.E. (2003) Both family 1 and family 2 PspA proteins can inhibit complement deposition and confer virulence to a capsular serotype 3 strain of *Streptococcus pneumoniae*. *Infect. Immun.* **71**: 75-85.

Ring, A., Weiser, J.N. and Tuomanen, E.I. (1998) Pneumococcal trafficking across the blood-brain barrier. *J.Clin. Invest.* **102**: 347-360.

Robbins, J., Austrian, R., Lee, C., Rastogi, S., Schiffman, G., Henricksen, J., Makela, P., Broome, C., Facklam, R., Tiesjema, R., and Parke, J. Jr (1983) Consideration for formulating the second-generation pneumococcal capsular polysaccharide vaccine with emphasis on the cross-reactive types. *J. Infect. Dis.* **148**: 1136-1159.

Robbins, J., Lee, C., Rastogi, S., Schiffman, G. and Henrichsen, J. (1979) Comparative immunogenicity of group 6 pneumococcal type 6A(6) and type 6B(26) capsular polysaccharides. *Infect. Immun.* **26**: 1116-1122.

Robbins, J.B. and Schneerson, R. (1990) Polysaccharide-protein conjugates: a new generation of vaccines. *J. Infect. Dis.* **161**(5): 821-832.

Romero-Steiner, S., Pilishvili, T., Sampson, J.S., Johnson, S.E., Stinson, A.R., Carlone, G.M. and Ades, E.W. (2003) Inhibition of pneumococcal adherence to human nasopharyngeal epithelial cells by anti-PsaA antibodies. *Clin. Diag. Lab. Immuno.* **10**: 246–251.

Rosenow, C., Ryan, P., Weiser, J.N., Johnson, S., Fontan, P., Ortqvist, A. and Masure, H.R. (1997) Contribution of novel choline-binding proteins to adherence, colonization and immunogenicity of *Streptococcus pneumoniae*. *Mol. Microbiol.* **25**: 819-829.

Rossjohn, J., Gilbert, R.J., Crane, D., Morgan, P.J., Mitchell, T.J., Rowe, A.J., Andrew, P.W., Paton, J.C., Tweten, R.K. and Parker, M.W. (1998) The molecular

mechanism of pneumolysin, a virulence factor from *Streptococcus pneumoniae*. J. Mol. Biol. **284**: 449-461.

Rubins, J.B., Charboneau, D., Paton, J.C., Mitchell, T.J., Andrew, P.W. and Janoff, E.N. (1995) Dual function of pneumolysin in the early pathogenesis of murine pneumococcal pneumonia. *J. Clin. Invest.* **95**: 142-150.

Rubins, J.B., Hanson Paddock, A., Charboneau, D., Berry, A.M., Paton, J.C. and Janoof, E.N. (1998) Pneumolysin in pneumococcal adherence and colonization. *Microb. Pathog.* **25**: 337-342.

Russell, M.W., Martin, M.H., Wu, H.Y., Hollingshead, S.K., Moldoveanu, Z. and Mestecky, J. (2000) Strategies of immunization against mucosal infections. *Vaccine* **19**(Suppl. 1): S112-117.

Sahm, D.F., Karlowsk, J.A., Kelly, L.J., Critchley, I.A., Jones, M.E., Thornsberry, C., Mauriz, Y. and Kahn, J. (2001) Need for annual surveillance of antimicrobial resistance in *Streptococcus pneumoniae* in the United States: 2-year longitudinal analysis. *Antimicrob. Agents Chemother.* **45**: 1037-1042.

Saluja, S.K. and Weiser, J.N. (1995) The genetic basis of colony opacity in *Streptococcus pneumoniae*: evidence for the effect of box elements on the frequency of phenotypic variation. *Mol. Microbiol.* **16**: 215-227.

Schrag, S.J., Beall, B. and Dowell, S.F. (2000) Limiting the spread of resistant pneumococci: biological and epidemiologic evidence for the effectiveness of alternative interventions. *Clin. Microbiol. Rev.* **13**: 588-601.

Schuchat, A., Robinson, K., Wenger, J.D., Harrison, L.H., Farley, M., Reingold, A.L., Lefkowitz, L. and Perkins, B.A. (1997) Bacterial meningitis in the United States in 1995. Active Surveillance Team. N. Engl. J. Med. 337: 970-976.

Shakhnovich, E.A., King, S.J. and Weiser, J.N. (2002) Neuraminidase expressed by *Streptococcus pneumoniae* desialylates the lipopolysaccharide of neisseria meningitidis

and haemophilus influenzae: a paradigm for interbacterial competition among pathogens of the human respiratory tract. *Infect. Immun.* **70**: 7161-7164.

Shaper, M., Hollingshead, S.K., Benjamin, W.H.J. and Briles, D.E. (2004) PspA protects *Streptococcus pneumoniae* from killing by apolactoferrin, and antibody to PspA enhances killing of pneumococci by apolactoferrin. *Infect. Immun.* **72**: 5031-5040.

Simberkoff, M.S., Moldover, N.H. and Rahal Jr., J.J. (1980) Absence of detectable bactericidal and opsonic activities in normal and infected human cerebrospinal fluids. A regional host defense deficiency. *J. Lab. Clin. Med.* **95**: 362-372.

Smith, H. and Bannister, B. (1973) Cerebrospinal-fluid immunoglobulins in meningitis. *Lancet* **2**: 591–593.

Sorensen, U.B., Blom, J., A, B.-A. and Henrichsen, J. (1988) Ultrastructural localization of capsules, cell wall polysaccharide, cell wall proteins, and F antigen in pneumococci. *Infect. Immun.* **56**: 1890-1896.

Spellerberg, B., Cundell, D.R., Sandros, J., Pearce, B.J., Idanpaan-Heikkila, I., Rosenow, C. and Masure, H.R. (1996) Pyruvate oxidase, as a determinant of virulence in *Streptococcus pneumoniae*. *Mol. Microbiol.* **19**: 803-813.

Steenhoff, A.P., Shah, S.S., Ratner, A.J., Patil, S.M. and McGowan, K.L. (2006) Emergence of vaccine-related pneumococcal serotypes as a cause of bacteremia. *Clin. Infect. Dis.* **42**: 907-914.

Steinfort, C., Wilson, R. and Mitchell, T.J. (1989) Effects of *Streptococcus pneumoniae* on human respiratory epithelium in vitro. *Infect. Immun.* **57**: 2006-2013.

Sternberg, G.M. (1885) The pneumonia-coccus of Friedlander (*Micrococcus Pasteuri*, Stenberg). *Ann. J. Med. Sci.* **90**: 106-123.

Tai, S.S., Lee, C. and Winter, R.E. (1993) Hemin utilization is related to virulence of *Streptococcus pneumoniae*. *Infect. Immun.* **61**: 5401-5405.

Talbot, U.M., Paton, A.W. and Paton, J.C. (1996) Uptake of *Streptococcus pneumoniae* by respiratory epithelial cells. *Infect. Immun.* **64**: 3772-3777.

Talkington, D.F., Crimmins, D.L., Voellinger, D.C., Yother, J. and Briles, D.E. (1991) A 43-kilodalton pneumococcal surface protein, PspA: isolation, protective abilities, and structural analysis of the amino-terminal sequence. *Infect. Immun.* **59**: 1285-1289.

Tasara, T. and Stephan, R. (2007) Evaluation of housekeeping genes in *Listeria monocytogenes* as potential internal control references for normalizing mRNA expression levels in stress adaptation models using real-time PCR. *FEMS Microbiol. Lett.* **269**: 265-272.

Tettelin, H., Nelson, K.E., Paulsen, I.T., Eisen, J.A., Read, T.D., Peterson, S., Heidelberg, J., DeBoy, R.T., Haft, D.H., Dodson, R.J., Durkin, A.S., Gwinn, M., Kolonay, J.F., Nelson, W.C., Peterson, J.D., Umayam, L.A., White, O., Salzberg, S.L., Lewis, M.R., Radune, D., Holtzapple, E., Khouri, H., Wolf, A.M., Utterback, T.R., Hansen, C.L., McDonald, L.A., Feldblyum, T.V., Angiuoli, S., Dickinson, T., Hickey, E.K., Holt, I.E., Loftus, B.J., Yang, F., Smith, H.O., Venter, J.C., Dougherty, B.A., Morrison, D.A., Hollingshead, S.K. and Fraser, C.M. (2001) Complete genome sequence of a virulent isolate of *Streptococcus pneumoniae*. *Science* **293**: 498-506.

Thompson, J.F., Hayes, L.S. and Lloyd, D.B. (1991) Modulation of firefly luciferase stability and impact on studies of gene regulation. *Gene* **103**: 171-177.

Timoney, J.F. (2004) The pathogenic equine streptococci. Vet. Res. 35: 397-409.

Tipper, D.J. and Strominger, J.L. (1965) Mechanism of action of penicillins: a proposal based on their structural similarity to acyl-D-alanyl-D-alanine. *Proc. Natl. Acad. Sci. USA* 54: 1133-1141.

Tomasz, A. (1981) Surface components of *Streptococcus pneumoniae*. *Rev. Infect. Dis.* **3**: 190-211.

Tombolini, R. and Jansson, J.K. (1998) Monitoring of GFP-tagged bacterial cells. *Methods Mol. Biol.* **102**: 285-298.

Tong, H.H., Blue, L.E., James, M.A. and DeMaria, T.F. (2000) Evaluation of the virulence of a *Streptococcus pneumoniae* neuraminidase-deficient mutant in nasopharyngeal colonization and development of otitis media in the chinchilla model. *Infect. Immun.* **68**: 921-924.

Torzillo, P.J., Hanna, J.N., Morey, F., Gratten, M., Dixon, J. and Erlich, J. (1995) Invasive pneumococcal disease in central Australia. *Med. J. Aust.* **162**: 182-186.

Towbin, H., Staehelin, T. and Gordon, J. (1979) Electrophoretic transfer of proteins from polyacrylamide gels to nitrocellulose sheets: procedure and some applications. *Proc. Natl. Acad. Sci. USA* **79**: 4350-4354.

Tseng, H.J., McEwan, A.G., Paton, J.C. and Jennings, M.P. (2002) Virulence of *Streptococcus pneumoniae*: PsaA mutants are hypersensitive to oxidative stress. *Infect. Immun.* **70**: 1635-1639.

Tu, A.H., Fulgham, R.L., McCrory, M.A., Briles, D.E. and Szalai, A.J. (1999) Pneumococcal surface protein A inhibits complement activation by *Streptococcus pneumoniae*. *Infect. Immun.* **67**: 4720-4724.

Tuomanen, E., Rich, R. and Zak, O. (1987) Induction of pulmonary inflammation by components of the pneumococcal cell surface. *Am. Rev. Respir. Dis.* **135**: 867-874.

van Ginkel, F.W., McGhee, J.R., Watt, J.M., Campos-Torres, A., Parish, L.A. and Briles, D.E. (2003) Pneumococcal carriage results in ganglioside-mediated olfactory tissue infection. *Proc. Natl. Acad. Sci. USA.* **100**: 14363-14367.

van Heerbeek, N., Straetemans, M., Wiertsema, S.P., Inqels, K.J., Rijkers, G.T., Schilder, A.G., Sanders, E.A. and Zielhuis, G.A. (2006) Effect of combined pneumococcal conjugate and polysaccharide vaccination on recurrent otitis media with effusion. *Pediatrics* **117**: 603-608.

Volanakis, J.E. (1990) Participation of C3 and its ligands in complement activation. *Curr. Top Microbiol. Immunol.* **153**: 1-21.

Vollmer, W. and Tomasz, A. (2000) The pgdA gene encodes for a peptidoglycan N-acetylglucosamine deacetylase in *Streptococcus pneumoniae*. *J. Biol. Chem.* **275**: 20496-20501.

Vollmer, W. and Tomasz, A. (2002) Peptidoglycan N-Acetylglucosamine deacetylase, a putative virulence factor in *Streptococcus pneumoniae*. *ASM* **70**: 7176–7178.

Waltman, W.D., McDaniel, L.S., Gray, B.M. and Briles, D.E. (1990) Variation in the molecular weight of PspA (pneumococcal surface protein A) among *Streptococcus pneumoniae*. *Microb. Pathog.* **8**: 61-69.

Weiser, J.N., Austrian, R., Sreenivasan, P.K. and Masure, H.R. (1994) Phase variation in pneumococcal opacity: relationship between colonial morphology and nasopharyngeal colonization. *Infect. Immun.* **62**: 2582-2589.

Weiser, J.N. and Kapoor, M. (1999) Effect of intrastrain variation in the amount of capsular polysaccharide on genetic transformation of *Streptococcus pneumoniae*: implications for virulence studies of encapsulated strains. *Infect. Immun.* **67**: 3690-3692.

Weiser, J.N., Bae, D., Epino, H., Gordon, S.B., Kapoor, M., Zenewicz, L.A. and Shchepetov, M. (2001) Changes in availability of oxygen accentuate differences in capsular polysaccharide expression by phenotypic variants and clinical isolates of *Streptococcus pneumoniae*. *Infect. Immun.* **69**: 5430-5439.

Whalan, R.H., Funnell, S.G.P., Bowler, L.D., Hudson, M.J., Robinson, A. and Dowson, C.G. (2006) Distribution and genetic diversity of the ABC transporter lipoproteins PiuA and PiaA within *Streptococcus pneumoniae* and related Streptococci. *J. Bacteriol.* **188**: 1031-1038.

Whatmore, A.M., King, S.J., Doherty, N.C., Sturgeon, D., Chanter, N. and Dowson,

C.G. (1999) Molecular characterization of equine isolates of *Streptococcus pneumoniae*: natural disruption of genes encoding the virulence factors pneumolysin and autolysin. *Infect. Immun.* **67**: 2776-2782.

Winslow, C.E.A., Broadhurst, J., Buchanan, R.E., Krumwiede, C.J., Rogers, L.A. and Smith, G.H. (1920) The families and genera of the bacteria: final report of the committee of the Society of American Bacteriologists on characterization of bacterial types. *J. Bacteriol.* **5**: 191-229.

Witte, W. (1999) Antibiotic resistance in Gram-positive bacteria: epidemiological aspects. *J. Antimicro. Chem.***44**(Suppl. A): 1-9.

WorldHealthOrganization(2007)Streptococcuspneumoniae[online].http://www.who.int/vaccine_research/diseases/ari/en/index5.html(May, 2007).

Wu, H.-Y., Virolainen, A., Mathews, B., King, J., Russell, M.W. and Briles, D.E. (1997) Establishment of a *Streptococcus pneumoniae* nasopharyngeal colonization model in adult mice. *Microb. Pathog.* **23**: 127-137.

Yarbrough, L., Wu, F. and Wu, C. (1976) Molecular mechanism of the rifampicin -RNA polymerase interaction. *Biochemistry* **15**: 2669-2676.

Yother, J. and Briles, D.E. (1992) Structural properties and evolutionary relationships of PspA, a surface protein of *Streptococcus pneumoniae*, as revealed by sequence analysis. *J. Bacteriol.* **174**: 601-609.

Zhang, J.-R., Mostov, K.E., Lamm, M.E., Nanno, M., Shimida, S., Ohwaki, M. and Tuomanen, E.I. (2000) The polymeric immunoglobulin receptor translocates pneumococci across human nasopharyngeal epithelial cells. *Cell* **102**: 827-837.

Zighelboim, S. and Tomasz, A. (1980) Penicillin-binding proteins of multiply antibioticresistant South African strains of *Streptococcus pneumoniae*. *Antimicrob. Agents Chemother.* **17**: 434-442.

Appendix

Journal Publications

LeMessurier, K.S., Ogunniyi, A.D. and Paton J.C. (2006) Differential expression of key pneumococcal virulence genes *in vivo*. *Microbiology* **152**: 305-311.

Ogunniyi, A.D.*, **LeMessurier, K.S.***, Graham, R.M., Watt, J.M., Briles, D.E., Stroeher, U.H. and Paton, J.C. (2007) Contributions of pneumolysin, pneumococcal surface protein A (PspA), and PspC to pathogenicity of Streptococcus pneumoniae D39 in a mouse model. *Infect. Immun.* **75**: 1843-1851 (* Authors contributed equally).

Mahdi, L.K., Ogunniyi, A.D., **LeMessurier, K.S.** and Paton, J.C. (2007) Virulence gene expression and innate immune responses during pathogenesis of pneumococcal infection in a murine model (submitted).

Poster presentations

LeMessurier, K.S., Ogunniyi, A.D. and Paton J.C. Differential expression of key pneumococcal virulence genes *in vivo*. 7th European Meeting on the Molecular Biology of the Pneumococcus, 8-11 May 2005, Braunschweig, Germany.

LeMessurier, K.S., Ogunniyi, A.D. and Paton J.C. Differential expression of pneumococcal virulence genes *in vivo*. Abstract P05.06, p 198. 5th International Symposium on Pneumococci and Pneumococcal Diseases, 2-6 April 2006, Alice Springs, Australia.

LeMessurier, K.S., Ogunniyi, A.D. and Paton, J.C. (2006) Differential expression of key pneumococcal virulence genes *in vivo Microbiology v. 152, pp. 305-311, February 2006*

NOTE: This publication is included in the print copy of the thesis held in the University of Adelaide Library.

It is also available online to authorised users at:

http://dx.doi.org/10.1099/mic.0.28438-0